# "Ownership gets you up in the morning, but can keep you awake at night'

# An exploration of the development of work-related psychological ownership in early career professionals

**Deborah Taylor** 

A thesis submitted in partial fulfilment of the requirements of Bournemouth University for the degree of Doctor of Philosophy

May 2022

**Bournemouth University** 

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

#### Abstract

Studies of Psychological Ownership (PO) have captured the imagination of scholars in organisational research and beyond due to the many suggested positive outcomes which occur from feelings of "mine". Nevertheless, the initial conceptualisation of PO has been under investigated, leaving early questions still unanswered and warranting further research. Consequently, this research aims to contribute to our understanding by exploring how PO develops in early career professionals who are entering the workplace.

A qualitative methodological approach consisting of twenty-six interviews with Work Placement Students (WPS), Supervisors and Placement Development Advisors (PDA) captures a rich, detailed tapestry of PO development. To date, quantitative studies dominate PO research, and this study aims to provide some much-needed methodological diversity.

Findings indicate four key conclusions. Firstly, that job-related PO can develop earlier than originally suggested, often via a building block approach with participants showing examples of task ownership within the first three months of their work placement and most demonstrating job role and project PO promptly thereafter.

Secondly, conclusions were reached that for job-related PO to be quickly and successfully developed, there is a requirement for ownership permissioning via an active relational "giving and taking ownership" process. Job-related ownership was expediated through the organisational culture clearly signalling ownership expectations and the supervisor actively "giving" ownership and demonstrating trust in the individual. "Taking" ownership was demonstrated by participants in two forms; either once permissioned, the individual job crafted to continually develop job-related ownership; or if ownership was not given or permissioned, it was sometimes still taken by the incumbent.

Thirdly, new outcomes from job-related PO were observed in this group of participants, including pride in their work and an increase in confidence. Nonetheless, if ownership was felt strongly, it often weighed heavily on the shoulders of participants, and this burden of ownership should not be underestimated. Given individuals are only just understanding "how work,

ii

works", employers need to be aware of the negative implications when individuals feel they fall short of perceived personal, supervisory and organisational expectations. Consequently, there needs to be management of the potential high stress levels this weight can cause.

Finally, a possible new PO target was identified in this research as careerrelated PO. For these young workers who have been conditioned to believe there is no job for life, the attachment to developing their career seemed exceedingly strong in some participants. This has potential to cause tension between other PO targets, such as the job role and the organisation. Career PO may become their one, consistent attachment in working life.

### Contents

Abstract	ii
Contents	iv
List of Figures	viii
List of Tables	viii
List of Appendices	ix
List of Abbreviations	x
Acknowledgements	xi
Chapter 1	1
1.1 Introduction	1
1.2 Background to the researcher and the research	3
1.2.1 Work Placements	4
1.3 The relevance of the research	9
1.4 Aims, objectives and research questions relating to the research	9
1.4.1 Objectives	10
1.4.2 Research Questions	10
1.4.3 Thesis Structure	10
Chapter 2 Literature Review: Psychological Ownership	12
2.1 Introduction to the Literature Review	12
2.2 Theoretical Origins of Psychological Ownership	12
2.3 Defining Psychological Ownership	14
2.4 Conceptual differences to similar theories	18
2.5 The Development of feelings of Possession and Ownership	19
2.6 Psychological Ownership Motives	20
2.7 Psychological Ownership Antecedents/Routes	26
2.7.1 Controlling the Target	26
2.7.2 Intimate Knowledge of the Target	28
2.7.3 Investing Self in Target	29
2.7.4 Safety	31
2.7.5 Self Congruity	32
2.7.6 Possessions and the Extended Self	35
2.8 Psychological Ownership Emergence	39
2.9 Measuring PO and Theories of Possession and Ownership	43
2.10 Psychological Ownership Sampling	45
2.11 Changing nature of Possessions	47

2.11.1 Liquidity of Possessions	49
2.12 Psychological Ownership Outcomes	51
2.13 Shadow side of Psychological Ownership	52
2.14 Demise of Psychological Ownership	54
2.15 Conclusion	55
Chapter 3 Methodology	60
3.1 Introduction	60
3.2 Research Aim, Objectives and Questions	61
3.2.1 Objectives	61
3.2.2 Research Questions	62
3.3 Philosophical Assumptions	62
3.4 Theoretical Perspective	64
3.5 Methodology: Qualitative Approach/Research Methods	64
3.6 Data Collection	66
3.6.1 Data Sampling	67
3.6.2 Sample Composition	68
3.6.3 Data Collection Process	72
3.6.4 Interviews	73
3.6.5 Research Ethics	77
3.6.6 Triangulation	78
3.7 Data Analysis	78
3.8 Trustworthy Research	84
3.8.1 Reflexive Thematic Analysis: Researcher Perspective (subjectivity)	85
3.8.2 Data Integrity and Communicating Results	87
3.8.3 Limitations	88
3.9 Summary	89
Chapter 4 Psychological Ownership in Early Career Professionals; Findings and Discussion	า 91
4.1 Introduction to Chapter	91
4.1.1 Theme Overview	91
4.1.2 Participant Context	92
4.1.3 Participant's interpretation of Ownership	95
4.2 Theme One: Giving Ownership; Job-related PO	97
4.2.1 Supervisor Role in "Giving Ownership"	103
4.2.2 Organisational Culture	109
4.3 Trust	116
4.4 Theme Two: "Taking Job-related Ownership"	122
4.5 Control	122
ν	

4.6 Investing Self in Target	. 126
4.7 Self-Congruity	. 127
4.8 Intimate Knowledge of the Target	. 130
4.9 Psychological Safety	. 133
4.10 Job Crafting	. 135
4.11 Theme Three: Pleasure and Pain	. 142
4.11.1 The Pleasurable Side of Job-Related Ownership	. 142
4.11.2 The Pain of Job-Related Psychological Ownership	. 148
4.12 Theme Four: Longevity or liquidity?	. 156
4.12.1 Decoupling / Divestment	. 156
4.12.2 Career-related Psychological Ownership	. 161
4.12.3 Future Selves and Liquid Ownership	. 166
4.12.4 Organisational Psychological Ownership	. 171
4.13 Bringing the Themes Together: Discussion	. 177
Chapter 5 Conclusion	. 191
5.1 Objective one: To explore the formation of work-related psychological ownership in individuals at the start of their career.	. 191
5.2 Objective two: To uncover the positive and negative outcomes suggested from the participant's experience of job-related psychological ownership	. 194
5.3 Objective three: To identify areas where qualitative research can add value to Psychological Ownership studies	. 195
5.4 Objective four: To provide a framework for how PO may be utilised and developed by	
organisations and their key stakeholders (e.g., line managers, HR)	. 197
5.5 Recommendations	. 198
5.5.1 Educational Providers	. 198
5.5.2 Students/Early Career Professionals	. 199
5.5.3 Employers	. 200
5.5.4 Researchers	. 202
5.6 Contributions to Knowledge	
	. 203
5.7 Future Research Directions	
<ul><li>5.7 Future Research Directions</li><li>5.8 Final Researcher Reflections</li></ul>	. 204
	. 204 . 207
5.8 Final Researcher Reflections	. 204 . 207 . 209
5.8 Final Researcher Reflections References	. 204 . 207 . 209 . 223
5.8 Final Researcher Reflections References Appendices	. 204 . 207 . 209 . 223 . 223
5.8 Final Researcher Reflections References Appendices Appendix One: Ethics Checklist	. 204 . 207 . 209 . 223 . 223 . 224
<ul> <li>5.8 Final Researcher Reflections</li> <li>References</li> <li>Appendices</li> <li>Appendix One: Ethics Checklist</li> <li>Appendix Two: Participant Information Form and Participant Agreement Form</li> </ul>	. 204 . 207 . 209 . 223 . 223 . 224 . 228

Appendix Five: Employer Interview Questions	. 234
Appendix Six: Placement Development Advisor Interview Questions	. 238
Appendix Seven: Data Analysis Phase One (Data familiarisation)	. 241
Appendix Eight: Data Analysis Phase Two (Generating Codes including Phase Two Codes)	. 243
Appendix Nine: Data Analysis Phase Two Continued (Data Coding Reduction)	. 247
Appendix Ten: Data Analysis Phase Three (Generating Initial Themes)	. 250
Appendix Eleven: Data Analysis Phase Four and Five (Reviewing and Defining Themes)	. 251
Appendix Twelve: A Mind Map of the Messy Reality	. 252
Appendix Thirteen: Extracts from Reflective Journal	. 253

# List of Figures

Figure One and Five	Psychological Ownership Overview	Page 55/188
Figure Two	Data Collection and Analysis Timeline	Page 66
Figure Three	Final Themes	Page 84
Figure Four	Elements that Contribute to Giving Ownership	Page 97
Figure Six	Job-related PO development in early career professionals	Page 189
Figure Seven	Differences regarding job-related PO development in early career professionals and PO overview	Page 189

# List of Tables

Table One	Researcher's philosophical assumptions and theoretical perspective	Page 61
Table Two	Work Placement Student (WPS) Demographics	Page 71
Table Three	Employer and PDA Demographics	Page 72
Table Four	WPS Organisational Information	Page 93
Table Five	Employer and PDA roles	Page 95

# List of Appendices

Appendix	Title
Appendix One	Ethics Checklist
Appendix Two	Participant Information Form and Participant Agreement Form
Appendix Three	Pilot Interview Questions
Appendix Four	WPS Interview Questions
Appendix Five	Employer Interview Questions
Appendix Six	Placement Development Advisor Questions
Appendix Seven	Data Analysis Phase One
Appendix Eight	Data Analysis Phase Two
Appendix Nine	Data Analysis Phase Two (continued)
Appendix Ten	Data Analysis Phase Three
Appendix Eleven	Data Analysis Phase Four and Five
Appendix Twelve	A Mind Map of the Messy Reality
Appendix Thirteen	Sample Reflective Account

# List of Abbreviations

ECP	Early Career Professional
PAO	Possessions and Ownership
PO	Psychological Ownership
ТА	Thematic Analysis
WPS	Work Placement Student

### Acknowledgements

I would like to express my gratitude to my supervisors at Bournemouth University. Firstly to Dr Lois Farquharson for her support and direction throughout the whole process. Without Lois, I would not have moved into an academic role and so I am extremely grateful for the opportunity. To Professor Adele Ladkin, who joined the supervision team in my final 18 months. Thank you for your encouragement, direction and humour. I have really enjoyed working with you. A big acknowledgement of support to my original supervisor, Professor Dean Patton who gave me the confidence that this was something I could achieve. I miss our conversations!

A big thanks to all my participants and the students involved in the study. I appreciate the time you made for me.

To my friends, a big thanks for all their encouragement and guidance throughout.

To my family, thank you for supporting me throughout this very long journey. To my parents, my sister, Richard, Ian, Jack and Josh thanks for the encouragement. You always asked questions and showed an interest even after all this time. Enjoy reading it, ha-ha! To Lottie for the cuddles and finally most of all to Col, who encouraged me to take on such a big commitment even when it meant working during the evenings and weekends. I think we'll both be glad to get our weekends back!

# Chapter 1 1.1 Introduction

Why are some individuals more attached to their job role or organisation than their colleagues and what are the outcomes of this attachment for these individuals and their employers? These are questions that have been posed by scholars this century who are interested in the development of a phenomenon named psychological ownership (PO) which is suggested to have significant workplace benefits, including job satisfaction, affective commitment, in-role performance, organisational commitment and organisational citizenship behaviour ((Van Dyne and Pierce 2004; Mayhew et al. 2007; Avey et al. 2009; Wang et al. 2018; Zhang 2020).

Early PO research (Pierce et al. 2001, 2003) captured the imagination of organisational behaviour scholars, as this affective and cognitive attachment can bind individuals to workplace targets such as their job role, their team or the organisation itself. This attachment or desire to have and to control things within our possession and make them part of our extended self (Pierce et al. 2001) is similar to how we feel and invest in tangible items such as our homes and personal effects (Belk 1988; Dittmar 1992). It should therefore be no surprise that theories regarding possession and ownership are seen as the conceptual core of PO and the concept of "mine" (Pierce et al. 2001).

Nevertheless, there are clear differences between PO and ownership of possessions due to the lack of legal or social sanctions that may come with most traditional forms of ownership including employee ownership schemes or worker co-operatives (Pierce and Furo 1990; Denegri-Knott et al. 2016). Instead, PO suggests a psychological state or mental attachment in which an employee develops possessive feelings towards certain targets, many of which are intangible (Pierce et al. 2001, 2003; Pierce et al. 2004). That attachment facilitates certain feelings which can have both positive or negative personal and organisational implications, although most research has emphasised the positive aspects such as those mentioned previously.

Whilst PO now goes far beyond its original organisational behaviour roots, including areas as diverse as augmented reality (Yuan et al. 2021), there are

still many unanswered questions about PO in the workplace. Organisational researchers have considered PO's conceptualisation and roots (Pierce et al. 2001, 2003; Van Dyne and Pierce 2004), it's development, measurement, outcomes and antecedents, yet there are early areas related to its conceptualisation flagged by Pierce et al. (2001, 2003) which still require investigation.

In particular whilst PO research has concentrated on work-based targets such as the job role or organisation, comparatively little conceptualisation has been undertaken into other areas such as collective PO, work-based tools or other work-related aspects such as organisational culture. Early PO research (Pierce et al. 2004, p.454) suggested a need for a more "fine grained view of PO" and whilst the research community have relied heavily on the initial conceptualisation (and an alternate perspective from Avey et al. (2009) underpinned with regulatory focus theory (Higgins 1997)), there is more interest in antecedents than detailed conceptualisation. Dawkins et al. (2017) PO review gave a rallying cry for further refinement yet five years on little has changed. Most research continues within the original framework, using the original surveys focussing on similar generalised full time knowledge worker populations. Whilst PO research has transcended countries and cultures, research into work populations remain rigid, leaving organisations unable to transfer PO understanding of experienced knowledge workers to other employment groups. Consequently it is challenging to ascertain exactly how PO is developed at the start of a career or how it waxes and wanes during an individual's organisational lifetime. Additionally, given the working world also relies on contract workers, those in the gig economy and other forms of shortterm employment, further refining of PO to encapsulate other workers would broaden and deepen scholar's knowledge.

This study will focus on **early career professionals** which is one of the aforementioned under-researched population groups thus starting a conversation about atypical working populations. The term early career professional does not always specify a particular timespan, although the American Psychology Association suggest it refers to individuals having only worked in industry for eight - ten years of their career. In the context of this

study it applies to individuals taking their first career step into the workplace. By exploring the development of work-related PO in individuals at the start of their career, using a sample of students on a work placement, there is also opportunity to simultaneously build PO understanding in early career professionals and individuals in a contract rather than a permanent job role.

Furthermore, this study uses a **qualitative method** to accomplish this study. Organisational based PO studies have traditionally used the survey (with Likert scales) method to ascertain feelings of ownership which has been beneficial at early inception, nevertheless Dawkins et al. (2017) has questioned the conceptualisation of the most popular survey introduced by Van Dyne and Pierce (2004). Given PO can be situated within our working lives which generally involve significant social interaction, qualitative studies would provide an opportunity to hear the lived experiences of PO development and some of the potential influences.

#### **1.2 Background to the researcher and the research**

The researcher's background has taken her from a business workplace to working in a university. Her experience managing the recruitment for an investment bank resulted in her involvement in early career talent programmes. This led to a role as a Placement Co-ordinator at a South Coast university working with students looking for work placements. Whilst she has moved into an academic role, her interest in early career professionals remains hence the context of this study. Having seen the transformational effect that a successful work placement can have on an early career professional in terms of career aspirations, self-belief and self-identity, resulted in the researcher's belief that for some individuals (who undertake work placements), this is the start of their career. Other researchers may suggest that for university student's their career starts in their first graduate role, but the insider knowledge of working with work placement students led this researcher to focus on this particular time period.

Why psychological ownership? Since a colleague first mentioned the term, it captured the imagination of the researcher, probably due to her own attachment to work place targets. As someone who has always felt attached to their job roles, she wanted to find out more. This, along with her insider

knowledge of work placements led her to be interested in exploring how PO manifests at the start of a career.

Therefore this research focusses on students from a British post-1992 university who are undertaking a compulsory work placement as part of their university degree. It is important to provide some background information to situate these individuals in their context.

PO research to date has mainly focussed on the establishment phase or middle career phases of individuals which, although useful when assessing the initial construct, misses the crucial stage when individual professional identities are still developing and are malleable. These participants are at the start of their career with relatively little previous work experience (other than part time work); are working in a job role with a fixed end date thus making their experiences transitory and are considered "net natives" as they are one of the first cohorts to enter the workplace having had access to the internet throughout their childhood and teenage years.

These WPS are situated in the organisational entry or initiation phase which is considered to last for the first two years of employment (Dalton et al. 1977; Cron 1984, p.86; Cron and Slocum 1986; Greenhaus and Parasuraman 1986; Greenhaus et al. 2010), managing the demands of transitioning from education into working life and moving from a "student" to "worker" identity, transferring knowledge to a different context (Ripamonti et al. 2018). As this is a period of professional identity building and work sensemaking, it is an important phase in developing workplace skills and offers researchers a myriad of opportunities to consider the factors that influence the development of PO in the workplace.

### **1.2.1 Work Placements**

Both the Dearing Report (Dearing 1997) and the Wilson Review (2012) emphasised the benefits of work-related-learning for university students, one form of which is a work placement. Work placements are a voluntary or compulsory part of an undergraduate degree (sometimes defined as a sandwich course); frequently positioned in year three of a four-year degree; lasting up to 56 weeks (thirteen months); with university staff (usually found within a placement team) acting as an additional supporter outside of the

organisation's own structure (Inceoglu et al. 2019). Whilst the participants in this study are only required to complete a 30 week placement, the nature of placement pipelines which require the current placement student to "hand over" to the next placement student means in reality most participants in this study worked on average for one year.

Terminology can differ across universities and organisations, with "internships" also a commonly used term (Aggett and Busby 2011), however, as historically this has referred to a shorter period of work experience, this study will use the term work placements. A Highfliers report (2020) suggested a record number of 13,514 work experience, internships, and work placements from the United Kingdom's top employers with many small and medium sized organisations also investing in student placements. The benefits to employers when hiring work placement students can include having an additional employee for a reasonably cheap outlay as well as providing an early career talent pipeline for the organisation. For students, in addition to income, work placements provide them the opportunity to see "how work works", clarify potential career options, build self-efficacy and develop a professional self-identity (Edwards 2014b; Inceoglu et al. 2019; Mele et al. 2021). The benefits post placement generally suggest WPS have a stronger academic performance in their final year of studies and improved graduate employment, including on average a higher salary (Mason et al. 2009; Crawford and Wang 2015; Brooks and Youngson 2016; Jackson 2017; Jones et al. 2017; Jackson and Collings 2018). These benefits are suggested to be partly due to an increase in confidence as well as being better able to link theory with practice. Work placement students are also able to return the "work structure" to education, which helps with the challenging aspect of final year studies and finding a graduate job role (Brooks and Youngson 2016). Nonetheless, some researchers suggest that whilst there are many positives, the post placement impact is less than often suggested (Wilton 2012; Inceoglu et al. 2019). Additionally, Jackson (2015) study of undergraduates on work placements in Australia did ascertain that some work placement students did feel anxiety and stress due to their lack of certainty on "how work works", whilst others lacked confidence due to their lack of mastery in areas such as IT skills, dealing with co-workers, mentors, and managers. The study does not indicate if these were long standing feelings or

if part of the initial transition into the workplace. Certainly, feelings of "how work works" and mastery of the job role suggest challenges of crossing boundaries. Odio et al. (2021) posits that this can be an important explorative, transformational period therefore such challenges are to be expected.

An individual's first job is often considered to be a "rite of passage" (Gennep 1960) in which individuals experience an identity change resulting in a move to adulthood. Liminality which has its roots in anthropology is characterised by a time bound period within this rite of passage whereby individuals feel "betwixt and between" identities (Turner 2011).

Different rituals can be widely observed in the workplace, with organisations that hire WPS and graduates often aiding that change of identity via induction and onboarding rituals. Other forms of rituals include rites of enhancement usually in the form of recognition of employees, be it award ceremonies, successes amplified in articles and company reports; rites of renewal strengthening department and organisational social structures; and rites of integration such as graduate programmes or Christmas parties (Trice and Beyer 1984; Islam and Zyphur 2009).

Whilst there is small body of research relating to liminality and work placements (Hart and Montague 2015; Mele et al. 2021), Hart and Montague (2015) suggest that a work placement is likely to trigger such a liminal process whereby an individual moves from a student identity to one of a professional. The pre liminal phase involves separation rites to help a new employee shed their student identity. These may be formal events during an onboarding process or informal interactions with their new peer group and line manager. The threshold, liminal or transitional period is one characterised by uncertainty, disorientation and ambiguity as the incumbent tries to fit into working life, undertaking intense sensemaking to help contemplate their new possible selves (Ibarra 1999). Mele et al. (2021) research with psychology students on an internship suggested the experience could be conceived as both a continuation and a separation from their student life. Whist this is a different experience as much academic learning still takes place to become psychologist, there is still considerable meaning-making and anxiety related to their new status. For business and marketing students, their new social

context allows them to disown their student self and experiment with new identities, in an inside-out-dialogic orientation or through being forced to recognise the changes that have taken place via an outside-in-dialogic orientation (Beech 2011).

Holmes (2015) suggests that when considering an individual's career-related identity, both self and social identity are emphasised. Individuals presenting an emergent identity may find this is upheld or denied by others, or their colleagues may ascribe an identity to an individual which the individual then affirms or rejects. This interplay between both the self and social identities help individuals develop and validate new workplace personas.

These socially invisible changes require reflection from the individual and are often aided by social rituals that help new employees reconstruct their identity. Workplace events such as inductions or workshops along with a guide in the form of a line manager or "placement manager" signpost, role model and encourage the move to a socially accepted professional identity. Nevertheless, whilst outwardly they may look like a professional, there may still be internal angst as individuals struggle to feel comfortable in their new skin. The completion of a probationary period or the successful execution of key tasks may be ceremonies of incorporation resulting in the individual moving to a post liminal identity of "worker" or "employee" and a greater feeling of belonging (Gennep 2013).

Entry into this new life phase is likely to involve both formal and informal social rituals facilitating the move into a "worker" identity. The liminal process may be both expediated and hampered within the social context via the organisational culture, the line manager and by peers. Some individuals may find themselves in limbo, neither seen as part of the team, nor no longer feel like a student, resulting in a loss of confidence or sense of belonging. Most individuals will make this successful transition, although time scales may differ depending on the individual and their circumstances.

The social construction of an early graduate career has been partly created via the narratives surrounding early career professionals, where change is a ubiquitous part of working life (Peeters et al. 2019), and universities need to prepare students for ongoing uncertainties and challenges in the workplace (Römgens et al. 2020). All of this continues the rhetoric that there is no job for life and that individuals will need to continually develop their skills to stay employable in a world where new jobs will be created using technology that doesn't currently exist (Kumar 2007).

In many ways, these assertions are not new with Milkovich and Hall (1977) conceiving that individuals should drive their own careers and DeFillippi and Arthur (1994) suggesting that through the boundaryless career individuals can cross employment boundaries. Nevertheless, rapid organisational changes, frequent technological innovation and more precarious work suggest a requirement for life-long learning to successfully manage recurrent transitions across different forms of work (Direnzo and Greenhaus 2011; Savickas 2011; Forrier et al. 2015; De Vos et al. 2021).

Donald et al. (2019) suggest by taking career ownership an individual in essence forges a protean career (Hall and Moss 1998) whereas those individuals who do not own their career are more likely to establish a more traditional work life. However, this form of career ownership suggests individual's control their career destinies rather than applying it to the theoretical concept of PO. Nevertheless, PO and career ownership have been briefly considered together suggesting PO could mediate individual's subjective career success and their work-related outcomes.

However, Donald et al. (2019) study does emphasise that undergraduates when questioned suggest their self perceptions of employability are influenced by human capital, career ownership and career advice. Thus, suggesting that whilst today's notion of graduate career has a clear social construction, graduate employment is also far more complex than often perceived (Tomlinson 2013).

Therefore whilst this study is related to the construction of PO, there is a myriad of social influences and expectations which encapsulate individuals at the start of their career. This period of great change when mindsets are malleable, but uncertainty can be high, situates participants in a different context to most previous PO research participants.

Despite the government's consistent interest in undergraduate's work related learning over the last twenty-five years, there is less research covering work placements than might be expected, with most studies focussing on strengths and weaknesses of work placements, transitions into the workplace and then back to university, employability skills and the development of professional identities (Ibarra 1999; Ibarra and Barbulescu 2010; Ibarra and Petriglieri 2010; Moores and Reddy 2012; Brooks and Youngson 2016; Jackson 2017; Tsai et al. 2017). This final area falls within the scope of this study, both relating to how early career professionals might develop feelings of ownership in the workplace and how this links to their identity.

#### **1.3 The relevance of the research**

The relevance of this research to the academic and practitioner community is the exploration of two under researched areas in PO, that of early career professionals who are working in a fixed term contract role. By using qualitative research there are opportunities to provide some rich contextual information highlighting the lived experiences of individuals who are discovering how "work, works". By interviewing line managers and university staff who visit these early career professionals whilst in the workplace, there is an opportunity to see different perspectives of PO through social interactions and work life stories. This research does not aim to generalise, rather to explore the complexities and transitory nature of work in order to contribute to the widening of the PO conversation.

#### 1.4 Aims, objectives and research questions relating to the research

The dearth of literature relating to PO in early career professionals/contract roles and the lack of methodological diversity has provided the direction of this study. Therefore the purpose of this research is to explore the development of work-related PO in individuals at the start of their career, using a sample of work placement students from a post 1992 university.

### 1.4.1 Objectives

- 1. To explore the formation of work-related psychological ownership in individuals at the start of their career.
- 2. To uncover the positive and negative outcomes suggested from the participant's experience of job-related psychological ownership
- To identify areas where qualitative research can add value to Psychological Ownership studies
- 4. To provide a framework for how PO may be utilised and developed by organisations and their key stakeholders (e.g., line managers, HR).

### **1.4.2 Research Questions**

- 1. How might psychological ownership of work-related targets be constructed by individuals at the start of their career?
- 2. What factors seem to influence the development of work-related psychological ownership targets in this specific group of individuals?
- 3. What do these individuals, line managers and Placement Development Advisors perceive to be the potential outcomes (both positive and negative) of developing job-related psychological ownership at the start of a career?
- 4. How might qualitative studies benefit our understanding of psychological ownership?

### **1.4.3 Thesis Structure**

This study is presented over five chapters. Chapter one has provided an introduction to the thesis with context relating to this particular group of participants.

Chapter two summarises the PO literature including PO's suggested conceptual core relating to theories of possession and ownership (PAO).

Chapter three outlines the methodological divergence of this study which used qualitative research to explore the development of PO in this target population.

Chapter four provides an overview of the findings from this study highlighting four key themes relating to the development of PO in this group of participants.

Chapter five outlines the conclusion of this study outlining the theoretical contribution of this study, recommendations and areas for future research.

# Chapter 2 Literature Review: Psychological Ownership

### 2.1 Introduction to the Literature Review

This chapter will consider the literature relating to Psychological Ownership, starting with a brief overview of some of the considered job-related PO targets before providing a definition and demonstrating the conceptual differences with other organisational related theories. Then it will move to an overview of the conceptual core of PO, theories relating to Possessions and Ownership (PAO). Further sections will provide more detail of why individuals might develop feelings of ownership within an organisational setting, the posited routes to PO development and what outcomes it may bring to an organisation and individual, be they positive or negative. The later sections of this literature review assess how PO is measured and sampled before providing a conclusion. The discussion will weave in theories relating to early career professionals on a work placement, serving as a reminder of the focus of this study, that is of inexperienced workers in shorter-term assignments. It is acknowledged that the majority of PO theories to date are based on research undertaken with experienced knowledge workers who are mostly in permanent roles. Rather than a hindrance, this allows us to explore alternative experiences thus helping us understand stable elements of the constructs and those which may be relevant to other groups. Given the complex nature of the topic these early pages provide crucial background information and therefore will not immediately relate to this context.

### 2.2 Theoretical Origins of Psychological Ownership

The notion of PO was first introduced to Management scholars suggesting a form of organisational ownership that went beyond legal requirements (Pierce et al. 1991). This emotional and behavioural ownership sparked interest from organisational scholars when Pierce et al (2001) seminal work on psychological ownership suggested individuals might have feelings of ownership for targets such as our job roles or the organisation itself. Four articles written by Pierce and colleagues (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004) laid the theoretical foundations for PO and whilst there has been some contrasting theorising (Avey et al. 2009; Avey et

al. 2012) most theorists have used the original conceptualisation as the basis for further research.

Pierce et al. (2001) posit that PO predominantly comes from the conceptual core of possessiveness and the psychology of "mine" with PO development highly influenced by work related to the psychology of possession including work by Furby (1978a) and Dittmar (1992) further explored in later sections. Pierce et al. (2001) were also influenced by anthropologists and philosophers, but given the stated feelings of possessiveness, "what is mine" there is an emphasis on the scholars in theories of possession and ownership.

In their early work, Pierce and colleagues (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004) suggest that as well as individuals having possessive feelings for their possessions, individuals can have similar feelings to non-physical items, stating previous research relating to; owning our labour (Locke 1690); children's ownership of nursery rhymes and owning our ideas (Isaacs 1933; Heider 1958).

They posit PO as a multifaceted phenomenon with a cognitive and affective element in which individuals feel possessive for certain workplace targets (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004). This early theorisation suggests that ownership is a human characteristic, part innate, part learned behaviour, part cognitive and part affective which gratifies several key motives resulting in feelings of attachments to targets such as our job role. These motives are: efficacy and effectance, self-identity and a sense of place/home and whilst independent of each other, their strength may be influenced by an individual's personality or values.

Pierce and Colleagues (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004) allude to the complexity of PO emergence suggesting a smorgasbord of motives, routes, individual and target factors required to facilitate PO development. They propose three key routes for PO development: control, investing self in target and intimate knowledge of the target. Again, these routes are distinct, however the greater the number of routes, the stronger the feelings of ownership. They suggest control and investing self are the most effective routes to PO partly due to their interaction with each other. To facilitate emergence, targets also need to be attractive to individuals, open and accessible with the opportunity for them to be manipulated. Thus, organisations will need to play their part in PO development with work environment structures facilitating or reducing barriers to PO. Whilst individuals may feel an immediate or strong initial feeling of ownership of targets because investing self in target and intimate knowledge of target take time to develop, Van Dyne and Pierce (2004) indicate that it will take time for these feelings to strengthen. It is posited that feelings may not last forever and decoupling or divestment rituals may help to facilitate this process, although to the author's knowledge, no research relating to PO and disinvestment has been undertaken.

The following sections provide a more detailed overview of psychological ownership and its conceptual core, theories relating to possession and ownership.

#### 2.3 Defining Psychological Ownership

PO is a multidimensional construct in which individuals and groups have an emotional and cognitive attachment to a particular target which results in feeling psychologically tied to it. This may be demonstrated by statements of attachment such as "I feel this job is part of me" and "I feel a sense of personal ownership" (Pierce et al, 2001, 2003). Such attachment has aroused a growing interest from theorists as it is often associated with a range of organisational and individual benefits, for example job satisfaction: (Mayhew et al. 2007; Bernhard and O'Driscoll 2011; Knapp et al. 2014); creativity, organisational commitment; (Van Dyne et al 2004); organisational citizenship behaviour; (Van Dyne et al 2004) organisational based self-esteem (Song et al. 2014). There are some studies pertaining to the dark side of psychological ownership although this has been less widely explored (Brown et al. 2005b; Brown 2009; Brown and Zhu 2016; Smith et al. 2018; Cocieru et al. 2019b; Zhang 2020).

However the interest in psychological ownership goes far beyond theorists in organisational studies with research undertaken in marketing and consumer behaviour (Zhang et al. 2014; Hillenbrand and Money 2015; Hulland et al. 2015; Jussila et al. 2015; Karahanna et al. 2015; Peck and Shu 2018; Reto and Jacob 2019; Pick 2021), family businesses (Rantanen and Jussila 2011; Mahto et al. 2014), co-operatives (Saila et al. 2012), technology (Henri et al. 2008; Brasel and Gips 2014), sport (Cocieru et al. 2019a), education and healthcare (Wood 2003; Md-Sidin et al. 2009; Asatryan et al. 2013; Mifsud et al. 2019), environmental psychology (Matilainen et al. 2017) and philosophical studies (Bermúdez 2015). Whilst this literature review will concentrate on the organisational aspect of PO research, other areas will be included to provide the reader with a greater breadth of knowledge.

Whilst Pierce et al (1991) first introduced the notion of PO in the early 1990's, the first frequently referred to definition was first provided in 2001:

"that state in which individuals feel as though the target of ownership (material or immaterial in nature) or a piece of it is theirs (i.e. it is MINE!)" (Pierce et al. 2001, p.299)

This definition suggests a feeling of being psychologically tied to a target and the notion that possessiveness is at the core of the construct. By using a possessive pronoun, the definition suggests that an individual dually feels ownership for something, but also ownership over something. Whilst individuals can intellectually or rationally consider a target, PO elicits an affective as well as a cognitive response resulting in an emotional connection which goes beyond cognitive boundaries. We have all seen a child state that this toy is "mine" when in an argument with another child, but it is also likely that adults have experienced similar feelings for example when they feel that someone has taken their idea. Whilst this may not have provoked a public utterance of "mine", it may frequently evoke a similar, but internal emotional response. By highlighting both the material or immaterial nature of PO the definition suggests that individuals can feel ownership for both physical objects

and immaterial notions/targets. Finally by indicating a piece of the target it is suggesting that PO can be shared.

Pierce et al. (2003) make a subtle change to their original definition in a subsequent paper to suggest PO is:

"the state in which individuals feel as though the target of ownership or a piece of that target is "theirs" (i.e. "It is mine!")"

(Pierce et al. 2003, p.86)

Thus withdrawing the "material or immaterial in nature". The bracketed section of the quote signifying PO's as more intangible than the previous definition suggests. Given that PO's conceptual core relates to PAO, it feels likely that this was to show a clear demarcation between feelings of ownership for something tangible such as a house, car, or piece of clothing and intangible targets such as our ideas, job role and organisations.

A later article by Van Dyne and Pierce (2004) includes the material/immaterial element again, but slightly alters the definition to emphasis the possessive core of the construct:

*"the state in which an individual feels that an object (i.e., material or immaterial) is experienced possessively"* 

(Van Dyne and Pierce 2004, p.442)

This definition provides greater emphasis on possessiveness and reestablishes the notion that PO can be both physical (object) and (something) intangible. By emphasising both material and immaterial objects, this definition suggests individual's may feel ownership for both physical objects such as work tools, as well as immaterial notions such as our ideas. This provides less distinction between psychological and other forms of ownership such as legal ownership, however it does highlight the multifaceted and complexity of PO as a construct i.e., work tools may be legally owned by the organisation, however an individual who uses them daily, may still consider them "mine." Conversely, we may legally own items such as a car, but never feel an emotional

connection of psychological ownership because it's only interest to us is its functionality.

However, the Van Dyne and Pierce (2004) definition has not been cited by Pierce or other scholars who have returned to either the 2001 or 2003 definition (Pierce et al. 2001, 2003). The "(it is mine)" element adequately portrays possession to scholars without the need to include the word "possession" which may also explain the exclusion of the bracketed "material or immaterial."

In 2010, Pierce and Jussila introduced a definition for collective PO which has been used by other scholars undertaking research into collective PO in organisations (Rantanen and Jussila 2011; Pierce et al. 2018; Gray et al. 2020; Su et al. 2020):

*"the collectively held sense (feeling) that this target of ownership (pr a piece of that target) is collectively "ours".* 

(Pierce and Jussila 2010, p.812)

There is a change in terminology from a "state" to a "sense" or "feeling" however this may be due to the collective nature of the construct. Throughout PO literature a state, feeling and sense of ownership are used interchangeably by scholars, however this seems to be personal preference rather than any underlying differences in interpretations.

Collective PO is outside of the boundaries of this study and therefore will not be discussed in any further detail.

For the purposes of this study, the Pierce et al. (2003) definition has been adopted as the researcher believes this most closely fits the key components of PO and concurs with other scholars that the bracketed "material or immaterial in nature" is not required.

#### 2.4 Conceptual differences to similar theories

There are other organisational theories which highlight the self and organisation with Pierce et al. (2001) providing some conceptual differences between PO and other organisational-individual theories such as organisational commitment and identification. At the core of PO is this notion of possessiveness relating to questions "what is mine/what do I feel is mine?" and the key motives; efficacy and effectance, self-identity and control which will be discussed later in this chapter (Pierce et al. 2001; Brown et al. 2014b). Organisational commitment differs as it is partly linked to social membership and considers the question "Do I wish to maintain my association with the organisation?" Whilst organisational identification is linked to social identity theory "Do I define myself in a similar way to the organisation?" (Pierce et al. 2001, p.305). There are however posited similarities between all these concepts as they relate to the notion of self-identity and reciprocal relationships (Pierce et al. 2001). Nevertheless, they are conceptually different enough for theorists to consider them to be distinct concepts (Dawkins et al. 2017).

Brown et al. (2014b) built on perceived differences between PO and other organisational constructs to include concepts which relate to the individual such as job satisfaction, experienced meaningfulness, and job involvement. Job satisfaction relates to a pleasurable appraisal of a job role "how does my job make me feel", Experienced meaningfulness refers to ones perception of the value of a job role "How important is the work I perform", whilst job involvement refers to a cognitive assessment of how important work is to the individual's self-concept (Brown et al. 2014b, p.322). Once again none of these constructs relate to this desire to possess that is the fundamental element of PO. Pierce and Jussila (2011) do nevertheless posit that job involvement is one of three key individual differences which effects PO emergence, a point rarely referred to since.

Given the nature of the attachment, core job-related psychological constructs such as a vocation or calling were also considered, however these provide a greater emphasis on a sense of purpose from an internal/external summons (Dik et al. 2009; Duffy et al. 2018) and are therefore significantly different.

### 2.5 The Development of feelings of Possession and Ownership

This section provides an overview of PO's conceptual core of possessiveness to provide context regarding its initial conceptualisation by Pierce and Colleagues (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004). Whilst this section may seem irrelevant to PO, as this section considers the origins of PO, it is helpful to provide an overview to establish how this concept has moved from feelings for material possessions to the immaterial.

Our daily lives are surrounded by possessions; some practical, some personally meaningful but financially worthless and some that feel central to our lives. Our relationship with our possessions can be a complex, multifaceted, multi layered phenomenon (Furby 1978a; Mittal 2006) with some possessions becoming part of our extended self and anchoring our identities (Belk 1988).

The process by which these feelings develop is open to debate, with some considering it to be innate (Burk 1900; Porteous 1976), whilst others suggest feelings develop as part of a socialisation process (Brooks and Lewis 1974; Furby 1978a, 1978b). However, having all seen a toddler describe a toy as "mine!" it is clear this connection starts early.

Feelings of ownership towards possessions strikes early with mastering and controlling our environment occurring in young infants as part of the competence/effectance motivation (White 1959; Belk 1988). By controlling objects rather than being controlled by objects, Furby (1978a) posits that individuals can build a more robust sense of self.

Ori et al. (2013) suggest that young children can infer ownership early, partly via a reasoning process which may include a judgement about previous ownership. Fasig (2000) posits that from the age of two children understand who an object belongs to, whilst nursery age children can ascertain that owners have greater rights to control (Kim and Kalish 2009; Rossano et al. 2011). Furby (1978a) suggests that the use and control of possessions

become important from an early age which extends to adulthood, although most frequently mentioned by nine to ten-year olds.

Furby (1978a) considers how our feelings and uses for possessions change during childhood; at age seven to eight possessions are considered for use or to enjoy; age ten to eleven when children develop responsibility for the care of their possessions and during high school when possessions shift to become objects of social power and status e.g., owning highly desired trainers. From this point onwards into adulthood, some possessions start to be considered as an extension of the individual hence one of the reasons why material goods and the construction of self are considered by some to be closely linked. Possessions thereafter can link to our sense of satisfaction, status, social power, and accomplishments displayed through individual's wealth, social standing, group affiliations, as well as career prowess. If we replace the term possessions with work in the prior sentence, it demonstrates the interchangeability of material and immaterial targets and how closely they relate to our identity (which will be discussed further in the comping chapters).

#### 2.6 Psychological Ownership Motives

Having laid out the conceptual core of PO, this section relates to PO in more detail starting with the suggested PO motives and the discussion around these. As previously mentioned, (Pierce et al. 2001b, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004) posit three key motives of why individuals may feel the need to possess organisational targets:

**Efficacy and Effectance** - Pierce et al. (2001b) posit that to feel in control of what we own, individuals explore their environment thus satisfying the need to be efficacious. By having the opportunity to explore and control one's environment, Pierce et al. (2003) suggest it produces intrinsic pleasure and extrinsic satisfaction for individuals.

Furby's (1978a) study of possessions in humans suggested that an oftenmentioned characteristic of ownership and a key reason why individuals are motivated by ownership and possessions is the instrumental function that

possessions provide defined as "makes possible some activity or convenience for owner" (Furby 1978a, p.58) This instrumental function links to White (1959) work on the competence motive and the motivation to have control of one's environment and thus feel efficacy. From these feelings links build between the object to the extended self.

Different motivations may mean that the same object can be used in the same way in an environment, but provides different motivational outcomes e.g., a bicycle may result in an adult feeling healthier and in control of their weight; for a child, it may be a way to spend time with friends thus satisfying a social need; or it may be a means of getting to school for a teenager. Therefore, several people may own a similar possession, but the desired outcome for everyone may often be different. Furby (1978a) suggests that this form of effectance motivation allows us to control and affect our possessions in a manner which is important to us.

Within the workplace, individuals may work in the same job role, but have a very different view to its instrumental function, depending on their desire for control, but also depending on what is important to them. Someone who enjoys working with others may take great satisfaction in building relationships with colleagues in other departments, whilst others in the same job role may gain more satisfaction in completing complicated tasks. Mastery will come in many different forms and people derive different levels of pleasure from different tasks. However, what will be similar for all individuals is the opportunity to control the target and a choice in how to work (Ferraro et al. 2011; Ye and Gawronski 2016a). One means of control may be via job crafting which will be discussed in section 2.7.

The efficacy or effectance motive has not been questioned by other PO theorists, although it has been renamed self efficacy by some theorists (Avey et al. 2009).

Whilst there is no current PO literature ascertaining the role of efficacy and effectance with early career professionals (ECP), it could be suggested that there may be different levels of efficacy as individuals try to build confidence

within their new positions. This may be facilitated by the amount of control given to WPS by their employers. There are studies suggesting that work integrated learning can have a positive impact on student's self-efficacy (Edwards 2014a; Jaaffar et al. 2019) although these studies include other forms of work placements including internships, volunteering and research assistant roles.

**Self-identity** is the second motive for PO (Pierce et al 2001) and there are clear links to the literature of possession and ownership. McClelland (1951); Prelinger (1959); Furby (1978a) and Belk (1988) suggest that often when an emotional connection is made with a target it becomes part of our identity or extended self.

As referred to previously, Locke (1690) suggested that as well as owning ourselves, we own our labour and what we produce from our labour. Today however, whilst some individuals in certain roles e.g., artists and farmers may be able to clearly see the fruits of their labour, for many it is a harder distinction to make. Often our output or ideas becomes the ownership of the organisation and whilst some may ensure that ideas contributed are clearly denoted as "my" idea, often the source of ownership within the workplace is harder to define.

Galvin et al. (2015) who considered narcissistic organisational identification suggest that in the same way that a material object can become part of the owner's identity, so to can an intangible object such as an organisation. Possessions often reflect who we are and how we wish to be seen by others and this interactive process helps us emotionally connect and maintain a sense of how we are. We hear phrases such as "They are a company man" which suggests that the individual and the organisational identity has become linked. Nevertheless, given that the organisation is a more distant construct to the job role with fewer daily opportunities to connect, is there enough time and access for employees on fixed term contracts to build this attachment and for the organisation to become part of a WPS identity? Is it important when building a talent pipeline for these young professionals to feel organisational PO and really identify with the organisation or is identifying with the job role enough?

Therefore if an individual feels self-congruity with the organisation and wishes to be seen as working for that organisation, is organisational PO still important?

Given that identity has a place of prominence in the literature related to psychology of possessions and "mine", whilst other PO theorists have acknowledged identity as a key motive, it has not been developed further. Dittmar (1992) and Hillenbrand and Money (2015), suggest we have an iterative relationship with possessions or other targets helping us to regularly define and reflect on who we are and allow us to show ourselves as we wish to be seen.

Hillenbrand and Money (2015) have considered identity and PO in more detail than other theorists, suggesting that there are four layers of self (core, learned, live and perceived) which relate to our personal and social identities. They suggest that when we make a statement regarding a target, we are also simultaneously making a statement about ourselves thus including something about how we perceive our own identity. They hypothesise that PO manifestation can occur in all four levels of self and occurs if a target allows an individual to act or live in a congruent manner. Therefore, if an individual believes they are altruistic, PO manifestation may occur when working for an organisation which shares their values for example working for a charity. The organisation may be congruent with their core and learnt self allowing a person to maintain their true self "living" their work life being "perceived" by others as giving back. However, using this example, if PO manifestation only occurs at the perceived self levels and an individual works for a charity because they want to look altruistic there may be incongruence resulting in dysfunctional behaviour. There has been very little analysis of these arguments however and further examination of the role of identity within PO may be useful.

Given early career workers are likely to experience a liminal period, identity is particularly pertinent for these individuals as they may still feel like a student but are learning and living "ought and ideal" drivers (Hillenbrand and Money 2015). Individuals may need frequent reiteration from "perceiving" others about their professional persona as without this security they may remain

"betwixt and between" identities and feeling PO for organisational targets may seem unobtainable.

**Sense of place or belonging** is the final motive/root suggested by Pierce et al, (2001) and provides both physical and psychic health (Porteous 1976). This isn't only a tangible item such as a house, but something that provides psychological security (Brown et al. 2003). The origins of this motive derive more from philosophers and the ethological concepts of territoriality, rather than research relating to possessions. Simone Weil (1952) suggests that having a place is a "need of the human soul" (Weil 1952, p.41), whilst Porteous (1976) proposes that if we control our territory or the space around us, it satisfies three key areas; identity, security, and stimulation. Individuals control their spaces in a similar way to animals by personalising and defending our territories.

There has been comparatively little research regarding sense of belonging/home within organizational studies. Whilst it has generally been of interest to scholars for generations, relatively fewer studies approach this topic from an organizational stance, and most have considered work and belonging from a negative stance (Hershcovis et al. 2017; Newheiser et al. 2017). Porteous (1976) suggests that our home is at the core of these feelings of identity and whilst our office may become important to us, because other people such as cleaners and workmen can gain regular access, the feelings of "other homes" are never as intense. This contagion could also relate intangible targets such as taking over a job role or tasks from another individual. Avey et al (2009) has suggested that belonging to a department, group or the organisation may mean that the socio-emotional feelings of having a place may be fulfilled. Given that Howell and Hill (2009) suggest that the need to belong is better satisfied by shared activities rather than via possessions, perhaps a sense of belonging also links more closely to our social identities?

Nonetheless as these individuals are new to the workplace and will be part of the organisation for around a year is their sufficient time for them to develop a sense of belonging and so is this motive as relevant?

Whilst Pierce and colleagues (Pierce et al. 2001b, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004) suggest a sense of belonging as a motive, there does not seem to be the same level of theoretical underpinning to link this to psychological ownership compared to the other motives. Nonetheless, Avey et al. (2009) does include three questions related to belongingness in his PO measurement which has been verified (Avey et al. 2012) and other scholars have accepted it as one of the three PO motives.

Whilst the above three roots/motives have been widely accepted (by for example Mayhew et al. (2007); Brown et al. (2014b); Hillenbrand and Money (2015); Dawkins et al. (2017); Zhang (2020)), Avey et al (2009) have posited that both accountability and territoriality are further PO motives. They used Higgen's (1997) regulatory focus theory as a basis for their proposals, suggesting that self efficacy, belongingness, self-identity along with accountability, are examples of promotion orientated feelings whereas territoriality would be a preventative focus. They posit that individuals with promotive PO would be more likely to share information, and hold themselves and others accountable, whereas a preventative PO viewpoint would result in territorial behaviour such as withholding information and ideas or showing defensive behaviour. Avey et al (2009) suggest that accountability is a PO motive because of our motivation to ensure we take control for what happens to targets which we believe are an extension of ourselves. In addition, because of our need to possess a space or place, Avey et al (2009) suggest that territoriality may sometimes be a factor in those people with a strong level of PO. This may be low level behaviour such as placing a jacket over the back of a chair to suggest it's "my chair" to more extreme behaviour where individuals withhold information or trying to claim a shared resource as their own (Brown 2009).

Brown et al. (2005a) in an early piece suggested that territoriality is a behavioural outcome and Alok's (2014) research concluded that promotion and preventative PO are different and could not be considered part of a multidimensional construct. Nevertheless, Avey et al (2012) used the same promotion PO scale in a further study exploring ethical leadership and concluded that the measures are substantiated. Olckers (2013) suggests that
Avey et al (2009) has focussed on the cognitive elements of territoriality rather than behavioural aspect and consequently should not be discounted (although this has not been supported by other scholars).

Stimulation (activation or arousal) was briefly mentioned by Pierce et al. (2003) in reference to (Porteous 1976) although less emphasis was originally attached to its potential role as a key motive. Nevertheless (Pierce and Jussila 2011; Rantanen and Jussila 2011; Jussila et al. 2012; Jussila et al. 2015) have included stimulation as a fourth motive during the intervening period. The desire of individuals to seek stimulation within their environments relates to activation/arousal. The desire to create, control, care for, master, and transform organisational targets satisfies our need for stimulation. This motive has not been widely discussed by other scholars who tend to refer to the early four papers (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004) when considering theory development.

It is important to understand these motives in such detail because Pierce et al. (2001) suggest that the motives for PO are the reason PO exists. If individuals and organisations understand what motivates us to feel PO for workplace targets (and see PO as a positive thing), then they may be able to facilitate access to these feelings in the workplace. Nevertheless, research to date has not ascertained if any of these motives are more important than others and if different motives are key at different times. This research may therefore identify which motives seem crucial in PO development for this group of ECP.

# 2.7 Psychological Ownership Antecedents/Routes

As mentioned in section 2:1, Pierce et al. (2001) initial conception conceived three key routes regarding the development of PO feelings for organisational targets. Whilst these routes can emerge separately, it is suggested that they may be interrelated. They posit that any route can facilitate PO, however the greater the number of routes available to an individual, the stronger the feelings of PO for the target.

# 2.7.1 Controlling the Target

Control within our environment is a fundamental human motive (White 1959). Control satisfies the effectance motive by allowing individuals to feel

that they are "the cause" (Pierce and Jussila 2011) of changes to their environment rather than being controlled within an environment. This then over time facilitates the link to the extended self (Belk 1988; Pierce et al. 2001).

Early PAO research (McClelland and McManus 1941; Prelinger 1959) suggests control to be an important determinant of ownership feelings which led Belk (1988) to posit that the greater our belief of possessing an object or being controlled by an object, the greater the likelihood of it becoming part of self. Furby (1978a) suggested that it is the element of control which can be the trigger to ascertain the likelihood of possessive feelings and the extent that possessions became part of self. People who hoard often experience intense control of their possessions such is the strength of their attachment, with Steketee and Frost (2007) suggesting that even others touching their possessions can feel like a violation of self. Belk (1988) suggests that both control by objects and control over objects may also be viewed as part of self. Control is therefore a significant element in both PAO and PO literature emphasising how it facilitates strong relationships with possessions and immaterial targets.

Individuals have greater opportunities to control their job role rather than their organisation due to the constant use and interaction involved with job tasks. There have been a number of studies looking at the relationship between job-related PO and control. In a study of 239 employees in New Zealand organisations, Pierce et al. (2004) found perceived control of job role mediated the relationship between work environment structure and job-related PO (Pierce et al. 2004). In particular perceived control in a job role can fully mediate the relationship between routine technology and job-related PO. Additionally, perceived control can partially mediate the relationship between routine technology and job-related PO. Pierce and Jussila (2011) concur demonstrating a positive relationship between job-related PO and experienced job control and in particular complex job design.

Job complexity has been described by Brown et al. (2014b) to include roles which involve breadth and depth of tasks, provide autonomy to explore,

personalise and adapt the scope of tasks along with understanding the intricacies of the overarching role. This in turn provides more opportunities for individuals to perceive control of their job role. Their two studies demonstrate that job-related PO has a positive relationship with job complexity and PO mediates the complexity/performance relationship (Brown et al. 2014b). Zhang (2020) meta-analysis of PO research concurs and suggests managers should provide opportunities for work autonomy which includes an element of control. As these participants are junior staff members, it is unclear if the WPS participants will have access to the depth and breadth of tasks suggested to be required in job complexity, nor if the control of others such as their supervisors will be extended to these participants. Also given the qualitative nature of this study, the research in these thesis will investigate if the experiences with control may differ from other groups of workers and to also consider how control is given to individuals. There is an information gap regarding the transfer of control in relation to organisational PO. By completing a qualitative study, there is an opportunity to explore the relational aspect of PO and how ownership may (or may not) be transferred between supervisor and employee.

Peng and Pierce (2015) posit that as organisations are more abstract than tangible elements of a job role, it is harder to develop organisational PO solely via control route especially within large organisations. Nevertheless, their results show that experienced job PO does positively relate to organisational PO. Other studies (Pierce and Jussila 2011; Peng and Pierce 2015) have also recognised job-related control's influence on organisational PO. Others such as Chi and Han (2008); Han et al. (2010) suggest a positive relationship between employee participation in decision making and organisational related PO. Liu et al. (2012) findings concur suggesting both participative decision-making and a self-managing team climate positively relate to Organisational PO.

#### 2.7.2 Intimate Knowledge of the Target

It is through a living relationship that we build an attachment to an object (James 1890) which Pierce et al. (2001) posit is true for workplace targets. Individuals may become psychologically tied to a target if there are

frequent interactions providing an opportunity for familiarity and a long-term association. For example, a gardener may feel linked to a garden they have continuously cultivated and spent time enjoying (Weil 1952). In our job roles, this may occur through the day-to-day completion of tasks over a period, but also the opportunity to talk about work to others. When considering the organisation as a target, organisational accessibility and interacting with colleagues to develop social networks may all help to facilitate PO towards an organisation (Zhang 2020) as will the quality and number of interactions (Jussila et al. 2015). Tenure within the job role or organisation is considered crucial by Pierce et al. (2003) to develop these feelings, although no guidance is provided regarding this time period. This study hopes to provide a greater understanding of the required PO tenure and if it is possible to develop feelings of PO when working on a relatively short-term assignment.

As the participants in this study will have worked in their organisations for less than a year, there is an opportunity to ascertain the importance of intimate knowledge as a PO route and if it is possible to feel this way about a target relatively quickly. If so, what impact, if any might it have on an individual's development of psychological ownership.

#### 2.7.3 Investing Self in Target

From the literature regarding ownership and possessions, it is important for individuals to have the opportunity to create, invest psychic energy and shape their possessions to aid and develop self-identity (Sartre 1956; Csikszentmihalyi and Rochberg-Halton 1981; Belk 1988). This builds on Locke (1690) work who conceived that we associate our work outcomes with our labour and our labour is part of ourselves. Pierce et al. (2001) and Chi and Han (2008) posit that the same is also true regarding the entwinning of self and organisational targets for the development of PO. The opportunity to invest self into our work via our mastery and through creating new systems, processes or products helps build that notion of "mine" (Pierce et al. 2003). Employee participation in decision making and self managing team environments provide opportunities to invest self in targets (Chi and Han 2008; Han et al. 2010; Liu et al. 2012). In an early paper, Parker et al. (1997) suggested perceived responsibility and concern for a target is found within

those with a strong ownership perspective. Pierce et al. (2001); Pierce et al. (2004) suggest that feelings of responsibility are an outcome of PO and therefore PO and responsibility are two different states.

Job design has been posited as a means of facilitating this investment of self in job role targets (Pierce et al. 2009a; Peng and Pierce 2015), however very few researchers have looked beyond job design which given the recent interest in job crafting by organisational scholars could be an oversight. Whilst job crafting is related to the work design literature, there is a clear difference; work design is a top-down process in which managers create job roles and tasks whilst job crafting is a bottom-up process in which individuals are motivated to adapt elements of their job roles over a period (Wrzesniewski and Dutton 2001)

Job crafting is a proactive means in which employee's change the physical and cognitive boundaries of their job role to enhance their overall job satisfaction (Wrzesniewski and Dutton 2001; Demerouti 2014). This personal job reframing, or remoulding is considered to relate to changing task, cognitive and relational boundaries and provides an individual the means of shaping a job role to fit their skills, knowledge, and motivation (Wrzesniewski and Dutton 2001). Literature in this area has focussed on either the predominantly gualitative side characterised above, or the crafting job demands, and job resources perspective (JD-R) focused on antecedents which has mainly been detailed from a quantitative perspective (Lazazzara et al. 2020). There are three elements to JD-R; seeking job resources, seeking job challenges, and reducing job demands. Given these elements require an investment of self, should organisations and line manager's encourage job crafting if they wish to facilitate job-related PO? Likewise, a recent addition to the job crafting literature, career crafting that asserts that in the same way individuals can craft the parameters of their job for a better personal fit (Wrzesniewski and Dutton 2001). Individuals may engage in career crafting which involves the same proactive behaviour, but related to their goals (Tims 2020). They suggest two elements in the career crafting survey which relate to proactive career reflections and proactive career construction (Tims 2020).

There have been three studies which develop the links between job crafting and psychological ownership, although two of the articles are not currently available in English. Naeem et al. (2021) studied employees working at a large pharmaceutical organisation in Pakistan using JD-R perspective. They ascertained that job-related PO mediates the relationship between job crafting and expansion orientated crafting such as seeking job challenges/job resources and transformational leadership. They suggest that the motives of PO may lead to proactive job crafting behaviours in which individuals look to seek out further job-related opportunities and resources such as autonomy or seeking feedback. However, they did not find a relationship between job PO and reducing job demands i.e., diminishing the emotional impact of job role or managing work life balance. Given very few studies have considered the dark side of PO, this element will be discussed further in chapter 2.5. Nonetheless, job crafting and in particular task crafting does seem to provide opportunities for control and self investment which are important routes to PO (Naeem et al. 2021).

#### 2.7.4 Safety

Following a meta-analysis of 141 PO studies, safety has recently been posited as an emerging antecedent (Zhang 2020) which allows individuals to be themselves without suffering any repercussions to their career or their perceptions of self. In this context safety is suggested to include trust, organisational justice regarding perceptions of fairness, relational closeness and perceived organisational support including feeling valued and supported. From the meta-analysis they suggest that safety is a vital component in PO emergence although do admit that a limitation of their study is that it is based on current PO literature. There are very few journal articles relating to organisational PO and safety or to any of the posited four elements of safety. Brown et al. (2014b) examined the role of trust in organisations and its moderating influence on territoriality and psychological ownership. Results suggest that whilst territorial behaviour can be reduced in high trust environments, for those individuals who still display territorial characteristics they may be perceived as self-serving and less favourably perceived by their peers. Elsewhere Olckers and Enslin (2016a) found trust mediates the relationship between PO and turnover intentions. Whilst, Chi and Han (2008)

study posited that organisational justice mediates the relationship between three forms of organisational ownership including PO. Given the recent release in (Zhang 2020) article, there are no current responses to this posited antecedent.

#### 2.7.5 Self Congruity

Self-congruity links to the mere and endowment effects (Heider 1958; Beggan 1992) whereby individuals look favourably on objects they own due to the psychological attachment made. Morewedge (2021) suggests that this attachment is stronger when the object links to an individual's social identity. As a recently suggested PO route, there are no studies relating to this proposed PO route/antecedent nevertheless, if we refer back to theories of PAO, there are similarities to Ferraro et al. (2011) self-worth links. In the same way that we build stronger attachments to objects that link to our self-worth, we may feel stronger attachments to our job role or our organisation if we believe it relates to our identity. In this study these are individuals at the start of their career, when social identity can be important, self-congruity may be important in building an attachment and developing job-related PO.

Two of the aforementioned routes to PO "intimate knowledge of the target" and "investing self in target" (Pierce et al 2001) suggest individuals need a time period in the workplace to form attachments, although the timings regarding manifestation have not been provided in this work or any other. However, if we consider our relationship with possessions, we are aware of examples of instant attraction (be it a new cuddly toy for a child or a car for an adult) which results in an immediate feeling of mine! Pierce et al. (2003) suggested that some individuals might build strong initial attractions, but for most it will be a lengthy, iterative experience. As mentioned previously, given this group of WPS will only be in the workplace for around one year, it is an ideal opportunity to provide greater understanding to potential timescales and ascertain its importance when discussing PO.

Nonetheless, whilst the routes are helpful in building our understanding of PO, as it is a complex, multifaceted construct, our attachment to organisational targets are more intricate and nuanced than current discussion of PO routes suggests. Whilst Pierce et al (2003) did highlight PO complexity in early

research and called for further work regarding factors leading to PO emergence, research since has negated to consider these intricacies in any detail although Brown et al, (2014) did consider the routes to PO alongside job complexity. However, this mainly concentrated on the role of job complexity within PO rather than the PO routes and therefore this notion has not been adequately studied.

Nevertheless, if we consider PO it feels plausible that some of us may have strong immediate attachments to certain organisational targets for a variety of different reasons. One explanation may be that prior building blocks from other facets of our life (such as being part of a university society or playing branded video games) facilitates feelings towards certain work-place targets. For example, if you undertake a marketing degree and move into a marketing role, have you already "invested self in the target of marketing" via your studies which may result in the development of early feelings of PO for the job role. Or through buying products from an organisation or having family and friends work at a company may this facilitate "intimate knowledge of the organisation?" Therefore, if an individual has "prior" related attachments to certain organisations is it possible to speed up feelings of PO for organisational targets thus providing a bridge to these new attachments?

It would be naive to suggest that this alone might facilitate feelings of attachment and scholars from theories of PAOs provide other potential factors. The Mere effect (Heider 1958; Beggan 1992) suggests that people are more likely to favour objects which they own and crucially their research found no time element influenced results. Additional research previously mentioned by Greenwald (2002) and Ye and Gawronski (2016) suggest that strong mental associations between the self-node and the object node can form if congruous, whilst the action-based model of dissonance (Harman-Jones 1999) and research by Brehm (1956) which suggests that behavioural commitment reduces dissonance. Therefore, ownership by choice may be key element when enhancing self-PO target congruity. Subsequently if an individual believes that they actively made a choice in choosing a particular workplace target and have a key desire to work for a particular job or

organisation, this may facilitate early development of PO towards organisational targets.

Furthermore, Ferraro et al (2010) suggested the self-worth link to possessions and therefore if our job role or the organisation is important to our self-worth, we are more likely to become self invested as it is important to us. For example, someone who is passionate about sport, may gain a self-worth match when working for a sports club and so is willing to invest more of themselves.

Conversely, some individuals may not invest as much of themselves into organisational targets as their self-worth is tied to other targets such as family or hobbies and/or the organisational target may not be as attractive to them. In these instances, feelings of PO manifestation and development may be slower or may not occur at all.

Rather than assessing if and how the aforementioned routes develop an individual's PO, scholars have more frequently researched further organisational antecedents with several studies concerning leadership including ethical leadership (Avey et al. 2012) and transformational leadership (Avey et al. 2009; Bernhard and O'Driscoll 2011; Park et al. 2013).

Additionally, the relationship between autonomy and job satisfaction is partially mediated by organisational PO (Mayhew et al. 2007), whilst the Henssen et al. (2014) study suggested the link between autonomy and PO also mediated the relationship with stewardship behaviour.

With regards to job-related PO, suggested antecedents include three work environment sources (technology, autonomy and participative decision making) mediate the relationship between experienced control and job-related PO (Pierce et al. 2004); autonomy partially mediating the relationship with job satisfaction and organisational commitment (Mayhew et al. 2007), whilst Kaur et al. (2013) suggested employee's spiritual and emotional intelligence predicted PO. Dai et al. (2021) study within the hotel industry suggested that PO mediates the relationship between regulatory focus, and work environment and Naeem et al. (2021) study found a link between job PO and seeking job challenges/resources. The three original PO routes have been accepted by scholars with studies showing the importance of control as a route to PO (Pierce et al. 2004; O' Driscoll et al. 2006; Liu et al. 2012). Investing self in target and intimate knowledge of target have generally been shown via employee participation in decision making and self managing team environments, although these studies have not been replicated. After a 20-year gap, safety has been posited as a new PO route (Zhang 2020) but will need further verification.

The initial three routes to PO are clearly established within the PO literature but feel less accessible to individuals at the start of their career as it will take time to have intimate knowledge of organisational targets, control targets or invest themselves in these targets. However, is it more complex than this and are there other factors which may contribute to PO development such as the culture of the organisation or the line manager? The following section will consider PO emergence and refer to the PO routes as part of this.

#### 2.7.6 Possessions and the Extended Self

The idea that possessions and our self identities are closely linked is not a new supposition with Locke (1690) suggesting that as well as owning ourselves, we own our labour and the products of our labour. This notion of linking self with possessions was also posited by James (1890):

"A man's self is the sum total of all that he CAN call his, not only his body...but his clothes and house...his reputation and works."

(James 1890, pp.291-292)

Sartre (1943) builds on these ideas suggesting objects aid the development of self and through understanding what we possess, we understand ourselves. This notion that having and being are closely entwinned allows individuals the opportunity to search and find who they are. Therefore, possessions go beyond their physicality – something that we may buy or own, to becoming things we believe we possess or believe is "ours." It allows us to mentally "try out" objects to see if they fit with who we believe we are or who we feel we would like to become (Csikszentmihalyi and Rochberg-Halton

1981). Sartre (1956) suggests control and mastery is important, along with creating and knowing. Csikszentmihalyi and Rochberg-Halton (1981) and Ahuvia (2005), concur, positing that an investment of energy or psychic energy is important. Watkins et al. (2015) suggests possession requires controlling, caring, and managing items linking to Locke (1690) idea of owning the products of our labour. This control and investment of energy is very ritualistic allowing individuals to turn an object into a meaningful personal possession (McCracken 1986).

Mittal (2006) also highlights forms of investment and energy when positing how possessions become part of self. Mittal more specifically divides the possession-self process into six key elements which occur prior and post-acquisition. Having choice in your acquisition and opportunities to invest resources occur prior to acquisition. This may include the investment of time, money, or energy. Sartre (1956) believed that buying possessions is also a form of creation because of the investment of energy.

Post-acquisition processes include investment by use and bonding which can also help an additional process of making memories. Finally, he signalled that collections are more likely to build stronger self-possession links due to the investment of time and effort in creating and maintaining a group of possessions.

Research by Ye and Gawronski (2016b) concurs with the importance of choice, having considered the relationship between implicit self-object and ownership. They posit that ownership by choice is important when enhancing self-object congruity. Building on previous research regarding dissonance (Brehm 1956; Harmon-Jones and Mills 1999) suggesting that behavioural commitment reduces dissonance, the act of choosing an object, helps strengthen our preference for it. This is also an act of control, so perhaps it is the dual action of releasing dissonance and feeling in control which helps develop person-object identity.

Whilst there may be a special bond between the individual and object, both Belk (1988) and Dittmar (1992) remind us that we are experiencing a three-

way relationship (self-object-other), partly from rivalry with others wanting what we have, but also because of our own desire to protect what is ours.

Additionally, we may gift, loan, or buy objects for others. Giving possessions to others may also be a means of extending self (Sartre 1956) and a special form of control. For the gift giver it may help connect the past, present and future creating a narrative which cements the extended self. However, for the receivers of the object this contagion can have both positive and negative connotations depending on the gift giver (Belk 1988). Csikszentmihalyi and Rochberg-Halton (1981) suggested that individual's favourite possessions were often gifts from loved ones because it becomes laden with memory markers. However, in other instances the receiver may struggle for the object to become part of their extended self due to beliefs about contamination.

Other writers such as Belk (1988); Watkins et al. (2015) see contamination as another means of integrating a possession into the extended self. Also referred to as contagion within the consumer behaviour literature, contamination can be a ritualist means of personalising an object so that it becomes part of the extended self. However as mentioned above, items shared with others may also feel contaminated as if there is a transfer of "essence" or "soul". There may be negative contamination such as clothes tried on by other shoppers (Argo et al. 2006); items owned by someone disliked (Rozin et al. 1989) or conversely sharing items with someone you dislike (Kramer and Block 2014). Alternatively, there may be positive contamination; items worn or owned by the famous (Lastovicka and Fernandez 2005) or touching an object owned by a high performer to increase their performance (Kramer and Block 2014) are two posited means. Whilst contamination has not been discussed in organisational PO, it could be possible for individuals to feel contaminated by the transfer of a job role, project, or task from another resulting in either positive or negative emotions depending on the individual.

Many individuals do however undertake divestment rituals which allow owners to ritually cleanse and transition meaningful possessions from one's life whilst still maintaining private meanings. In some instances, shared meanings will be

developed if the buyer and seller have commonalities facilitating a shared selfboundary, however in most instances, individuals are able to move objects from "me" to "not me" (Lastovicka and Fernandez 2005).

This notion of a possession narrative highlights the biographical nature of possessions and their active role in memory making. This includes both favourite possessions which may demonstrate someone's individuality or heritage and less favourite possessions which individual's feel are no longer part of them (Kleine et al. 1995; Ahuvia 2005; Watkins et al. 2015). This notion posits that narratives actively connect objects and people and act as memory markers for the stories in our lives. Objects have an active past, present and future often with a variety of owners and when we divest of objects, they often become part of someone else's life (Lastovicka and Fernandez 2005).

Whilst the nature/nurture arguments are still debated, it seems clear that possessions play a big part in an individual's life from childhood through to old age. Possessions and the self are closely entwinned providing a continuing journey between who we are, who we want to be and how we wish to be seen. In the PO literature, we would expect to see this close link with the extended self.

For our possessions to become part of our extended self, some psychic investment is required which often comes in the form of possession rituals such as creating, controlling, caring, contaminating, knowing, using, mastering, or buying items (Sartre 1956; Csikszentmihalyi and Rochberg-Halton 1981; Belk 1988; Ahuvia 2005; Mittal 2006; Watkins et al. 2015). Possessions symbolise something of the self, be it a gold medal won at a sports day or the title on your door which reads Managing Director. Possessions can be functional as well as denote status, social power, and accomplishments. The ones that signify who we are change status from being "mine" to being part of "me" (Belk 1988; Ferraro et al. 2011).

Within the PO literature, control and identity are likely to feature strongly with some consideration for how targets become part of "me" via rituals, choice, or the biography of the target. Finally, we would expect to see literature

regarding divestment rituals demonstrating how targets move from "mine" to "not me."

Whilst the discussion regarding possessions and the extended self mainly highlights continuity for a stable self-identity, societal changes along with new type of possessions are challenging our ideas and will be discussed in more detail later. The following section will return to PO and ascertain why we feel ownership and the ways in which it emerges.

#### 2.8 Psychological Ownership Emergence

Pierce et al. (2001, 2003) allude to the complexity of PO emergence in early PO research suggesting a combination of elements including PO motives, routes, individual and target factors are required to facilitate PO. This complexity however has been overlooked by many scholars with only a few scholars recognising the need for wider understanding of how PO emerges in organisational targets (McLntyre et al. 2009; Pierce and Jussila 2011; Brown et al. 2014b; Jussila et al. 2015).

Pierce et al. (2003) suggest that for feelings of ownership to emerge, targets must be attractive, accessible, with opportunities for individuals or groups to manipulate and develop them via the three posited routes (control, intimate knowledge, and investment in self). For example, a job role which includes lots of opportunities for autonomy and job crafting will be accessible and easily manipulated. Whether it is attractive will however depend on the incumbent's self congruence (Morewedge 2021) demonstrating the importance of understanding individual differences along with the role of the organisation in facilitating the development of PO towards organisational targets. Country culture has been explored (Peng and Pierce 2015), however organisational culture has tended to be seen as a by-product of areas such as the work environment structure (Brown et al. 2014b) rather than considering a more holistic view of the organisation. Given that PO can be a partly learned behaviour and is also a complex construct, it is likely that a variety of organisational touch points will influence how accessible, attractive and manipulatable a target is. Pierce et al. (2003) suggests organisational culture should be explored further, although this has not yet occurred.

As mentioned in the prior section, Brown et al. (2014b) suggests that the three original PO routes mediate the relationship between job complexity and jobrelated PO. Job complexity is characterised by a depth and breadth of tasks which allows individuals to take control and have influence over many tasks. The more complex nature of the tasks means that there are greater opportunities to invest self and physic energy into the job role, whilst to understand complicated tasks it is important to have intimate knowledge of the role. There are some differing opinions related to job complexity, however, with Jussila et al. (2015) highlighting that if roles are too complex this may inhibit an individual's desire to possess the target. Furthermore, in later research, Pierce and Brown (2019) suggest that weak situations better suit the development of PO in the workplace as it allows an individual's personality or attitudes shape behaviour rather than the current situation. They suggest that thoughtful job design provides opportunities for decision making which allows the individual to shape and control targets rather than feeling controlled by a situation. This raises questions regarding the importance of job complexity vs intimate knowledge and at what point it becomes detrimental to PO development. Is it actually the alchemy of personal characteristics, target attributes, PO motives, job complexity and routes combined with ingredients will slightly differ in shape and form for each individual?

Moreover, the aforementioned research suggests that individuals new to the workplace are unlikely to develop PO, nor would they be interested in its development because of the short-term nature of their tenure. Yet given PO has its basis in theories relating to possession and ownership which in turn suggest that individuals can have an instant attachment to possessions (Furby 1978a; Belk 1988), we could surmise that it is equally possible to have quick, yet strong attachments to workplace targets especially when related to PO motives such as identity or sense of belonging. From the points raised in section 2.5, the PO literature diverges from theories of possession regarding this point. Whilst this may be true, there is no research to confirm or disprove these suggestions.

Consequently, individuals participating in a work placement are an interesting group to study as they perform a variety of job roles in different departments,

within a range of different industry sectors and are often new to the full-time work force. Their roles involve a reasonably short tenure and may lack the complexity and diversity of tasks compared to other populations studied. This research may help us answer some of the many outstanding questions related to PO development in ECPs.

Pierce and Jussila (2011) book on Psychological Ownership discuss what they believe are three key individual differences:

Individualism/collectivism may mean that some individuals with a more individualistic self-concept may talk about "my" versus those with collectivist mindsets talking about "our". Individualism/collectivism may have an impact on the self-object-other relationship and how individuals deal with the rivalry and territorial feelings that can arise from strong related feelings of PO. Wang et al. (2018) believe that whilst individuals with high levels of job-related PO may undertake more organisational citizenship behaviour and be committed to achieve in their job role, they may also engage in claiming and defensive behaviour with others to ensure the boundaries of what they believe is "mine".

**Job Involvement** is posited as the degree to which work is central to the life satisfaction of an individual or how important an individual perceives work in their life or for their self-esteem or their identity (Lodahl and Kejner 1965; Lawler and Hall 1970; Saleh and Hosek 1976). Pierce and Jussila (2011) suggest that job involvement may be an important boundary condition in the development of job-related PO, although once again, no further work has been undertaken to assess these claims.

There is conceptual ambiguity regarding Job involvement, however, with Solana and Alicia (2019) suggesting there are at least four conceptualisations of job involvement. Their systematic review suggests that since 2008, most theorists have used the definition and instrument introduced by (Kanungo 1982) relating to both job and work involvement "A cognitive or belief state of psychological identification" (Kanungo 1982).

It is likely that job involvement, work design and/ or job crafting will facilitate possession rituals such as those utilised when building attachments with targets (McCracken 1986; Russell 1988; Pierce et al. 2001, 2003). PO rituals

may allow individuals to claim something as "theirs", providing them the opportunity to cultivate, control, care for, contaminate, know, master, or claim potential PO targets. Whilst possession rituals have been alluded to by Pierce et al. (2003) as a means of expediating PO emergence, it has not been explored in any detail. As discussed previously, rituals may be particularly prevalent for this group of individuals as they are at the start of their career. Ritualised behaviour which facilitates shared organisational and departmental culture and meanings often become visible to new starters in both formalised and informal means. The transference of a "handover book" from one work placement student to another may be a symbolic form of shared meaning, facilitating the new placement students transition into the workplace, but also providing the outgoing student with an opportunity to parade their "intimate knowledge of the job role" via this divestment ritual. Whilst many rituals are aimed at attaching the target to the individual's extended self, there are also divestment rituals designed to disengage with targets when an individual leaves a role or organisation (Pierce et al. 2003) and this will be discussed in section 2.8 regarding the demise of PO.

Lastly, **Organisational based self-esteem** is the final posited individual difference suggested by Pierce and Jussila (2011) which relates to individuals' perception of how important, worthwhile, and meaningful they are within the domain of the organisation. Once again there are limited research in this area, however Xiao-Fu et al. (2014) research suggested that organisational based self-esteem mediates the relationship between organisational psychological ownership and employee's positive organisational behaviours.

To conclude, because of the complexity of the PO construct, our understanding of PO emergence is still limited. (Pierce et al. 2001, 2003; Van Dyne and Pierce 2004) provided early discussion points for theorists to assess, which unfortunately has never really caught the interest of scholars. Given the rich, nuanced theorising that could have taken place at the early stages of a new construct, it feels that there is so much that we still need to understand about PO emergence and development.

#### 2.9 Measuring PO and Theories of Possession and Ownership

Organisational Studies have principally used the survey method for PO research, with two tribes of measures having formed. There is the seven-item measure designed by Van Dyne and Pierce (2004) emphasising feelings of possession in general statements such as "This is MY organisation". Dawkins et al. (2017) suggests several methodological concerns regarding measurement in particular the lack of clarity concerning item selection and the way in which questions relate to the three PO motives. As statements are unspecific the association with the efficacy-effectance motive is unclear. This can also be seen with the connection to sense of place/home which needs further refinement. Furthermore, whilst the Van Dyne and Pierce (2004) measure is a favoured choice of many theorists, poor face validity and factor loading concerns has resulted in partial replication on occasions (Mayhew et al. 2007; Bernhard and O'Driscoll 2011; Park et al. 2013).

Avey et al (2009) produced a sixteen-item measure reflecting the five items which they believe make up promotive and preventative orientated PO (belonging, self efficacy, self-identity, accountability, and territoriality) and is based on both inductive and deductive processes to generate items. The first four items form promotion PO scales, whilst territoriality is used to measure preventative PO. Concerns have been raised suggesting that promotive and preventative PO are two different dimensions (Alok 2014) and if territoriality is a motive or an outcome of PO (Brown et al. 2005a; Olckers and Schaap 2013; Dawkins et al. 2017) However, this survey has been validated Avey et al. (2009); Avey et al. (2012) and in comparison, to the Van Dyne and Pierce (2004) survey, clearly shows connections between items and PO motives.

Olckers (2013); Olckers and Schaap (2013) built on Avey et al. (2009) survey developing a thirty-item scale using promotion and prevention orientated PO to create the South African Psychological Ownership Questionnaire. They included two further dimensions (autonomy and responsibility); however selfefficacy was withdrawn at scale development stage due to the sample group differing interpretation of the term. Olckers has since returned to the Avey measure (Olckers and Enslin 2016b; Olckers and Zyl 2016). In addition selfidentity and sense of belonging loaded onto one factor. In China four items

from the Van Dyne and Pierce (2004) scale was adapted into Chinese by Chi and Han (2008).

Given the concerns regarding the Van Dyne and Pierce (2004) measure and the failure of other alternatives to capture the attention of scholars I would suggest further work needs to be undertaken to substantiate measures before further work considering antecedents is undertaken. Moreover, given that PO is a complex topic and much work is still required on its initial conceptualisation, would not some methodological diversity also be beneficial?

Law (2004) suggests that real life is "messy" and those in social sciences studies should accept that simplicity isn't always possible. Whilst dominant methods may provide understanding at one level, they often are unable to capture the complexity and ambiguity found in human life. If we compare research in PO compared to those in possession and ownership, there are far greater examples of methodological diversity within theories of possession and ownership. This allows greater opportunities to be "messy" as well as use dominant methods which in this instance are qualitative approaches. Overall methodological diversity in PAO is found via; theoretical articles (Furby 1978a; Belk 1988; Mittal 2006; Belk 2010, 2013; Belk 2014; Watkins et al. 2015; Molesworth et al. 2016; Bardhi and Eckhardt 2017; Denegri-Knott et al. 2020);semi structured interviews (Ahuvia 2005; Denegri-Knott and Molesworth 2009; Bardhi et al. 2012), phenomenological interviews (Karanika and Hogg 2013); photo elicited interviews (Tian and Belk 2005); surveys (Ferraro et al. 2011); a mixture of studies including scenario based online experiments with surveys (Fritze et al. 2020), experiments and questionnaires (Morrison and Johnson 2011; Kramer and Block 2014; Ye and Gawronski 2016a); participant observation (Lastovicka and Fernandez 2005) and participant observation and projective techniques (Kleine et al. 1995; Masset and Decrop 2016).

By utilising a diverse array of methods, it might be said that theorists can unravel the messy (Law 2004). Yet scholars within PAO have recently questioned the topic's methodological diversity during the Interpretive Consumer Research Conference in 2019, contributing to a Journal of Marketing Management article, (Denegri-Knott et al. 2020). It was posited that whilst there has been diversity of methods within this area, most scholars have

relied on qualitative approaches and this underlying historical context might benefit from recent advances in technology. An example provided would be data tracking to demonstrate the "doing" of digital possession work. They also suggest that the historical context of PAO research may influence theoretical underpinning used in comparisons between digital and tangible possessions, whereas "bolder theorising and methodological innovation" (Denegri-Knott et al. 2020, p.943) might ascertain digital possessions as a stand-alone topic not a poor relation to the past.

If we look at PO studies across all areas, methods utilised include; Wood (2003) study of students PO which measured feelings of PO via a survey with both numerical evaluations and open ended; Kirk used a variety of methods which included laboratory experiments followed by a survey to assess individuals PO territorial responses to certain tangible and intangible objects; Gray et al. (2020) research regarding collective PO in creative teams used a mixed methodology; whilst Matilainen et al. (2017) study of forest owners and Cocieru et al. (2019a) study of football club's supporters are rare because of the use of interviews. This is compared to the 141 quantitative studies used by Zhang (2020) in their PO meta-analysis and clearly demonstrates the lack of methodology diversity currently found in PO research.

Given the reliance on the survey method with the resultant PO measurement ambiguity I suggest that PO is at a methodological junction that would benefit from utilising alternative methods when developing and substantiating current claims. By exploring lived, messy experiences concerning the intricacies of PO, there is an opportunity to examine more closely the undoubted complexities of the concept and provide "heterogeneity and variation" within social science subject areas (Law 2004). Whilst qualitative studies may not be generalizable, the rich data provided would undoubtedly benefit PO scholars thus providing a much-needed detailed tapestry of individual's feelings.

#### 2.10 Psychological Ownership Sampling

Alongside the measurement of PO, PO sampling has tended to rely on homogeneous groups.

The literature regarding theories of possession and ownership provide an overview into how feelings for our possessions develop and change over our lifetime (Belk, Furby etc), however PO sampling has commonly focussed on full time knowledge workers with several years' work experience within an organisation (for example: Pierce et al. 2003; Pierce et al. 2004; Van Dyne 2004; Avey et al. 2009; Bernhard and O'Driscoll 2011; Avey et al. 2012; Brown et al. 2014; Wang et al. 2018). This narrowness in sampling has resulted in gaps in the PO landscape especially concerning key periods such as initial PO manifestation along with ignoring a significant proportion of organisational populations. Sampling the target, intimate knowledge of the target and investing self in target (Pierce et al 2001) which suggest some level of autonomy and longevity within the role/organisation which we will now consider in more detail.

PO research has focussed mainly on the establishment / middle career phases of individuals (Cron 1984; Cron and Slocum 1986; Greenhaus et al 2010; Griffin et al 2014; Johnson and LaFrance 2016; Super 1975, 1980) often with a mean/average age of samples in their thirties (including Pierce et al. 2004, average age of respondents 35.6; Van Dyne and Pierce 2004, average age 34; Bernhard and O'Driscoll 2011 mean age 39; Avey et al. 2012 mean age 41; Alok 2014 average age 31.76; Brown et al. 2014 average age in study one 36.7 years and in study two, age 32; Wang et al. 2018, average age in study one 33.77). As scholars have mainly considered antecedents of PO the intention of prior studies has not been to consider how PO alters over time and the factors that might induce a change in those feelings. However, this does mean that the academic community has very limited understanding of how PO develops and perhaps more importantly no clear understanding of PO manifestation. Given the large numbers of young people employed in our workplaces who have the opportunity to develop PO at a time point when their professional identities are still developing and malleable, this gap in the psychological ownership literature seems to be a pertinent place to investigate further. During the organisational entry or initiation phase considered to be the first two years of employment, (Dalton et al. 1977; Cron 1984; Cron and Slocum 1986; Greenhaus and Parasuraman 1986; Greenhaus et al. 2010)

individuals are managing the demands of transitioning from education into working life assessing their future "possible selves" (Markus and Nurius, 1986). By researching individuals at the start of their career there is an opportunity for longitudinal studies which would provide us with a greater understanding of the way in which PO might appear then wax and wane during an individual's career.

This age group is of particular interest due to the changing nature of their relationship with possessions compared to previous generations. This group have had a greater exposure to brands via online channels which may influence the attractiveness of some workplace targets. Certain organisations with very desirable products or image may result in individuals having already built mental relationships with said organisation. Organisations within areas such as sport, clothing or technology can be very prevalent in a young individual's life which results in said individual already feeling as if they have an intimate knowledge of the organisation or have invested themselves in the organisation. These changing forms of ownership may therefore have implications for the development of PO in individuals which will be discussed further in the following section.

#### 2.11 Changing nature of Possessions

In the 20 years since PO research have gained traction, research regarding theories regarding possession and ownership have also progressed. This section will provide an overview of these changes.

As individuals move away from physical possessions to newer forms of ownership such as digital goods and shared access services, ownership of possessions is becoming less clearly defined and thus our preconceptions about possessions and the extended self are also being questioned (Belk 2013). There are a range of different terms used to describe such services including amongst others; access-based consumption (Bardhi et al. 2012), sharing and collaborative consumption (Belk 2010; Botsman and Rogers 2010; Belk 2014), access-based services (Fritze et al. 2020), co-production (Humphreys and Grayson 2008), digital consumption objects (Molesworth et

al. 2016), digital virtual goods (Watkins et al. 2015) and digital possessions (Denegri-Knott et al. 2020). Some of these terms overlap, but they generally refer to items such as smartphones and laptops, digital items such as social media and gaming sites, bits of code and access services such as bike or car sharing. They differ from more traditional forms of possession in that they are may be immaterial, accessed and consumed via devices, co-created or may not involve a formal transfer of ownership (Belk 2010; Bardhi and Eckhardt 2012; Belk 2013; Belk 2014; Molesworth et al. 2016)

Early research regarding digital products posited concerns for the possessionself relationship; consumers saw digital products as intangible and easily replaced, (Watkins and Molesworth 2012), less easy to control and lacked social visibility which provided less certainty for consumers regarding the nature of ownership. All of which raised concerns by scholars regarding the relationship between these new possessions and the extended self (Siddiqui and Turley 2006; Petrelli and Whittaker 2010; Watkins and Molesworth 2012; Belk 2013)

Later research has however demonstrated that our relationship with these possessions may be more nuanced than originally posited (Denegri-Knott et al. 2020). This may be partly due to experience as they become part of daily life and so accepted as possessions, but also the original viewpoint may have been too simplistic. For example, Orth et al. (2019) study of owners of digital products suggested owners feel meaning associated to the digital contents they can access via their phone rather than the actual tangible phone itself, reversing the view that it is easier to feel ownership for the tangible vs the intangible. Belk (2013) suggests that self is now more jointly constructed via online sites such as Facebook allowing for affirmation of individual's identity via likes and shared photo's/experiences. The Facebook timeline allows individual's memory markers providing symbolic anchors of key experiences and documenting when they purchased new possessions. All of which can reiterate an individual's values and sense of self. These digital platforms encourage attachment to sites via this perpetual game of posting and liking. The psychic energy involved in establishing a presence commodifies

the self (Molesworth et al. 2016), but simultaneously also merges having and being (Sartre 1956).

However Digital Virtual Goods (DVG's) such as social media and video games consoles also fragment ownership as often technology organisations own the Intellectual Property. Whilst they provide individuals the rights to own the information they post or curate (Watkins et al. 2015), they are unable to move their "Spotify" play list to another site nor move their avatar across games thus limiting the same level of control that tangible possessions provide individuals (Molesworth et al. 2016). Nevertheless, there is freedom on these sites to disembody, reembody, cultivate, co-create and provide continuity (Belk 2013; Watkins et al. 2015)which bear some similarities to the possession rituals of old.

#### 2.11.1 Liquidity of Possessions

Based on Bauman (2000); Bauman (2007) work on liquid modernity, Bardhi and Eckhardt (2017) have provided a different perspective about the changing nature of consumerism. They suggest that liquid consumption brings less stability and more uncertainty in life which may result in the loss of opportunities for possessions to anchor our lives and aid identity development. They suggest that a consumption continuum or spectrum now exists with solid consumption and liquid consumption at either end. As society shifts towards liquidity via digital consumption items (DCO's), fast fashion and shared services there may be some change in the way in which we view our possessions, with millennials and younger generations having more frequent experiences of liquidity than previous generations.

They suggest that ephemerality, access-based consumption and dematerialisation make up this new form of liquid consumption in contrast to the solidity, stabilised, long-standing commitments of the past. In this new liquidity, fluidity and nimbleness is required to deal with the instability and uncertainty that may occur, and precarious workplace positions or industries may result in greater use of the sharing economy and liquid consumption. From an earlier study of global nomads (individuals who have lived in many different countries and were not anchored to one location) Bardhi et al. (2012) discovered that some people have a more detached and fluid relationship with possessions which they call "liquid relationships" (Bardhi et al. 2012). They concluded that objects that may seem important and part of our extended self during one part of our life may seem less important or irrelevant during another part. This may be due to the situational factors of the owners who are an atypical group, however Masset and Decrop (2016) concur with the suggestion of liquid relationships having studied the relationship tourists have with items bought on holiday. Their results intimate that whilst some relationships with possessions endured over a long period of time, with other possessions the relationship changed. Possessions aid identity construction, yet how we perceive ourselves is not static and during key moments of change (such as a house or job move, or a pandemic) we realise that some possessions no longer reflect who we are.

As individuals move along the consumption continuum toward liquidity, their consumption habits may follow a similar path to the global nomads involving more transactional relationships with fewer emotional attachments for shorter duration. Bardhi and Eckhardt (2017) suggest that consumers now desire ephemerality because it allows them the temporality to justify the constant upgrading that has become synonymous with modern day Western life. Bellezza et al. (2017) concur having discovered an "upgrade effect" that occurs with mobile phones where individuals consciously or unconsciously mistreat objects such as phones to justify the latest upgrade.

Liquid experiences also fulfil the desire for ephemerality and provide a new take on Sartre (1956) having and being, Having experiences rather than possessions become more important to self because of their fluidity and easier links to demonstrate self-identity especially via social media (Carter and Gilovich 2012).

Research relating to PO has not considered how the changing use of possessions might change the nature of our relationship with organisational targets. If self is being anchored by digital services showcasing our lives, does this have an impact on the connections we have with our job role or

organisation? For example, will individuals want certain job roles or be attached to certain organisations because of their online profile? Whilst this may not be a concern for all individuals, those at the start of their career, mainly millennials or generation Z are net natives and have grown up in a digital world with far more of their possessions being further along the liquidity spectrum than previous generations. Traditional ownership rites of passage regarding cars and homes may be delayed or may not occur at all with "shared ownership" for this group becoming increasingly important.

Moreover, given the posited ephemerality, is it possible that the strength of attachments to PO targets within the organisation may differ for this sample group as their relationships with targets are built on greater fluidity.

PO research has not yet considered these possible implications; therefore this could be a key area of future research.

# 2.12 Psychological Ownership Outcomes

PO is suggested to have several positive outcomes/effects related to both the organisation and the individual, hence why it has become a topic of interest for scholars.

Due to the pleasure producing elements of PO, individual outcomes relate particularly to job satisfaction, self-esteem, affective commitment, in-role performance, pro-active work behaviour, joy and job engagement (Van Dyne and Pierce 2004; Mayhew et al. 2007; Avey et al. 2009; Avey et al. 2012; Knapp et al. 2014; Peng and Pierce 2015; Wang et al. 2018; Zhang 2020; Henssen and Koiranen 2021).

Positive organisational outcomes include organisational commitment, organisational citizenship behaviour, work engagement, intentions to stay and stewardship (Van Dyne and Pierce 2004; O' Driscoll et al. 2006; Mayhew et al. 2007; Avey et al. 2009; Wang et al. 2018; Zhang 2020).

One area touched on in consumer PO is feelings of pride in this instance for items individuals own (Di Muro and Noseworthy 2013; Kirk et al. 2015; Ahuvia et al. 2018). Pride is suggested to be a two faceted emotional response often activated by personal evaluations and reflections of self (Tracy and Robins 2007; Carver et al. 2010). Authentic pride is most positive form often relating to achievements and self-worth, whereas hubristic pride, which can still be authentic relates to our perceptions related to being superior to others (Tracy and Robins 2007; Carver et al. 2010; Kirk et al. 2015). This demonstrates the dual faceted nature of pride as on the one hand we can be proud of our achievements, yet there are negative connotations related to "pride before a fall". Within the consumer behaviour and PO literature pride (Kirk et al. 2015; Ahuvia et al. 2018) relates closely to identity which given identities close links to PO therefore makes it surprising that it has not been considered more in the organisational PO literature.

Henssen et al's (2014; 2021) studies of CEOs of family-owned businesses, suggested the joy of ownership. Whilst a very different level of participant, this is one of the few studies that have indicated the pleasure producing side of PO first mentioned by Pierce et al. (2001).

There are also negative outcomes of PO which will be discussed further in the following section.

# 2.13 Shadow side of Psychological Ownership

There have been significantly fewer studies regarding the negative side of PO, with most research focussing on the notion of territoriality.

Early PO research (Pierce et al. 2001, 2003; Van Dyne and Pierce 2004) alluded to possible negative implications for those individuals with high levels of PO in particular. They posit that there may be a mix of conditions that result in some individuals developing negative feelings some related to the organisational context and in particular unwanted change and some related to individual personality types such as those who need to feel high levels of control. The sort of "shadow" side behaviours suggested including individual outcomes such as sense of burden, personal sacrifice, job strain, alienation, burnout, and frustration along with deviant and territorial behaviours such as knowledge withholding (Pierce et al. 2001, 2003; Avey et al. 2009; Pierce et al. 2009b; Brown et al. 2014b; Peng and Pierce 2015; Dawkins et al. 2017; Cocieru et al. 2019c; Zhang 2020; Henssen and Koiranen 2021). Whilst the burden of PO has been mentioned occasionally, only Pierce et al. (2009b)

have provided more of an overview of this weight of responsibility suggesting it may be a means of protecting their identity and sense of self.

A series of studies have investigated the relationship between territoriality and PO (Avey et al. 2009; Brown 2009; Graham and Sandra 2011; Brown et al. 2014a; Brown and Zhu 2016). Brown et al. (2005a) argue that territoriality is a behavioural outcome to a target that an individual feels attached to. These feelings warrant an individual to claim, maintain, signal, and defend said target in a similar way that an animal may mark out and defend its territory. If individuals feel PO for a target, they may engage in marking behaviours such as identity or control orientated behaviours which allows an individual to communicate what they believe is "theirs", for example, individuals in the workplace may personalise their desk or add their names to stationery items. Furthermore, fear of infringement of said targets may result in anticipatory or reactionary defences such as locking the door to your office or glaring at someone should they breach the boundary of your territory (Brown et al. 2005a; Brown 2009).

Marking and defensive behaviour has both positive and negative implications for individuals and their organisations. Whilst territoriality can mediate the relationship between turnover intentions and PO (Lu et al. 2017), facilitate organisational commitment and reduce workplace conflict, it may also make those individuals who are perceived to engage in territorial behaviour more isolated from colleagues or they become preoccupied with marking their territory and thus lose focus (Brown et al. 2005a). Team members are more inclined to negatively judge them (Brown et al. 2014b), believe they have less power in the workplace or consider them poorer performers (Brown and Zhu 2016). In addition, whilst those individuals with high levels of PO who work in high trust environments are less likely to engage in territorial behaviour. If they do, they are more likely to be self-serving and their colleagues may underestimate their team contributions.

Other studies have noted the duality of PO ascertaining that whilst knowledgebased PO when moderated by territoriality was likely to cause knowledge hiding, organisational based PO reduced the effect and therefore individuals with strong feelings for their company were less likely to knowledge hide (Peng

2013). Moreover (Wang et al. 2018) suggested that whilst job-related PO mediated the relationship between job engagement and OCB's, those with high levels of avoidance motivation were also likely to engage in negative implications such as pro job unethical behaviour, knowledge hiding and territorial behaviour. Finally Dai et al. (2021) study of hotel workers ascertained that promotion and prevention regulatory focus can reduce job burnout, although for individuals with low levels of PO, promotion focus is more likely to reduce job burnout than prevention focus.

Zhang (2020) meta-analysis of PO research along with Dawkins et al. (2017) PO review both call for further work to be undertaken regarding the shadow side of PO suggesting that areas such as stress and burnout warrant further investigation. They posit that those individuals who feel high levels of ownership may invest too much of themselves into the target which may result in feelings of anxiety or stress.

Given there has been no research work undertaken regarding the shadow side of PO in individuals at the start of their career or on fixed term contracts, there is a gap in our PO knowledge which this study aims to fill.

# 2.14 Demise of Psychological Ownership

Pierce et al. (2003) initially suggested that PO can disappear via a decoupling process like possession divestment rituals. The reasons for the disappearance of feelings may be due to a variety of factors such as changes to PO routes including loss of control within a job role; target characteristics changing e.g., a new organisational strategy or targets becoming less visible and malleable. They also posit that those from short term orientation cultures will find it easier to decouple than for those individuals with longer term orientations.

As this area has not yet gained interest from other scholars, there is a lack of empirical research to concur, although it does seem likely that PO will follow a similar trajectory to theories of PAO. Given the costs involved in employee turnover, a more nuanced understanding of this area is important generally. This study will not assess this area in detail, however given the participants have fixed term contracts, divestment rituals may be identified.

# 2.15 Conclusion

This chapter reviewed theories relating to psychological ownership and its conceptual core theories of possession and ownership thus providing an understanding of how PO was conceived and developed. Figure one demonstrates this overview:



Figure One: Psychological Ownership Overview

Nevertheless, whilst the development of PO and its related antecedents have been researched in some detail, research to date has concentrated on experienced workers who have spent some time in the workplace, leaving a large gap in our knowledge regarding how PO might actually develop in individuals at the start of their career.

It seems clear that the self and material / immaterial targets are closely entwinned providing a continuing journey between who we are, who we want to be and how we wish to be seen. The strongest attachments signify who we are, changing status from being "mine" to being part of "me" (Belk 1988; Ferraro et al. 2011) or in the case of collective PO to being part of "us". These links to our identities seem particularly pertinent for those individuals experiencing liminality and being "betwixt and between" student and professional identities (Turner 2011). With possible and future selves on the horizon emergent identities are socially constructed via an interplay of rituals, role modelling and experimentation (Beech 2011; Holmes 2015). Yet can attachments form in individuals who are working in a particular job role within a particular organisation for a short period of time? This is not entirely clear and so this study will consider not only if PO can be developed in this group of WPS, but also if there is evidence of strong attachments.

For our possessions to become part of our extended self, some psychic investment is required which often comes in the form of possession rituals such as creating, controlling, caring, contaminating, knowing, using, mastering, or buying items (Sartre 1956; Csikszentmihalyi and Rochberg-Halton 1981; Belk 1988; Ahuvia 2005; Mittal 2006; Watkins et al. 2015). Control of objects and targets seem to play an important part in this process, and we have a stronger connection to items which we choose and which we believe link with notions that we value.

This psychic investment has been translated into PO routes in the PO literature with control and investing self heralded as key routes and intimate knowledge of the target being the three widely posited routes. Zhang (2020) and Morewedge et al. (2021) have expanded the list of routes to include safety (which consists of organisational justice, trust, perceived organisational support and relational closeness) and self-congruity, although given these are recent addition, there has not been further discussion amongst scholars. Trust has been considered to play a part in developing feelings of ownership (Brown et al. 2014b; Olckers and Enslin 2016a), although other elements have not been considered in more detail by theorists.

Pierce et al. (2003) suggested that whilst all individuals may be able to develop feelings of ownership for organisational targets, these may emerge at different times and in different ways being heavily influenced by situational factors. PO emergence is complex and suggest that whilst at the cognitive level individuals might make quick attachments, to feel full ownership may be a long reiterative process. However, this has not been fully explained and no other scholars

have examined this area in more detail, nor explained PO's demise or regrowth. If we consider our relationship with possessions, it is true that some of our most treasured possessions have been with us for some time and their story entwines with our own. But there are also possessions which instantly become important to us either because they were given to us by someone special, they were self-made, or we can see someone physic energy. Therefore, both control and investing self may provide us with opportunities to quickly feel attachment. Only coming to intimately know the target really requires an investment of time.

Thoughtful Job design by organisations allows individuals more opportunities to control and adapt their job roles and thus has been heralded as a means by which organisations can facilitate opportunities for individuals to develop job-related PO (Pierce et al. 2009a; Peng and Pierce 2015) via a top-down approach. However, if we consider two of the PO motives investing self in target and intimate knowledge of the target, these are also partly controlled by the incumbent, in a bottom-up form of job designed which is more closely aligned to job crafting. Therefore, is job-related PO influenced by job crafting as well as job design?

Whilst investing self in target and job crafting could be seen to have similarities, one is about making the job role "mine", whilst the other is about changing the boundaries of the job role to provide greater job satisfaction. Job crafting allows individuals greater control and opportunities to invest self which are key routes to developing ownership and therefore could be more important in the development of PO than job design. This study allows us to consider perceived examples of job and career crafting and its possible impact on Jobrelated PO and the rituals undertaken to develop it.

Pierce et al. (2003) posited that possession rituals facilitate the development of PO and whilst this area is often referred to in the literature relating to possessions, it has been overlooked in the discussion regarding PO emergence and development. Given the suggestion that intense sense making helps border hoping into a professional self, rituals may be particularly key to facilitating PO in early career professionals. As this group of

participants are working on short term contracts, divestment rituals also seem likely to occur.

Moreover, our relationships with possessions are changing from a physical, tangible items to those which may be digital, shared or access based. There is growing evidence to suggest that we may have a liquid relationship with possessions (Bardhi et al. 2012; Masset and Decrop 2016). Research suggests that despite initial concerns that these items were less likely to become part of the extended self, digital items in particular are still considered meaningful providing powerful anchors to our extended self (Belk 2013; Molesworth et al. 2016). Sites such as Facebook, Instagram and iTunes mean that we no longer own items yet still having feelings of attachment via our creations (pictures, news, timelines etc). Those born at the start of this century may have a very different view of ownership as they are more likely to share ownership (Facebook, iTunes) and rent/subscribe rather than buy (Netflix).

Furthermore, as liquid consumption brings ephemerality, access-based consumption, and dematerialisation (Bardhi and Eckhardt 2017) will this mean that younger millennials and generation Z have less certainty in their lives resulting in posited short term transactional relationships with possessions (Bardhi and Eckhardt 2017) which could follow through to their careers. Within this research should we expect individuals to consider their careers as Protean careers or to career craft? PO research has not considered the implications of our changing relationship with possessions and how this may affect individual's ability to form attachments to organisational targets which this study will consider in more detail.

Most of the above suggests that what is missing from PO research is more granulated detail regarding PO development especially at the start of an individual's career. As observed in section 2.5, early conceptualisation (Pierce et al. 2001, 2003; Van Dyne and Pierce 2004) provided a myriad of potential areas for scholars to research, question and develop (including boundary conditions, target attributes, personality, personal values and cross-cultural contexts) yet research outputs have been focussed in certain areas such as antecedents and its links to other organisations constructs. If we look at the PO research landscape there is uneven coverage and whilst it is natural that

some areas take precedence over others, given PO conceptualisation is over twenty years of age this limited investment in certain areas is ultimately affecting the application of this concept in the workplace.

This may be partly due to studies regarding Organisational PO traditionally being framed by a positivist perspective. Dawkins et al. (2017) review of PO highlighted the need for refinement of both the conceptualisation and measurement of PO, yet some years later, many of these concerns remain.

Moreover, given the reliance on the survey method with the resultant PO measurement ambiguity I would suggest that PO is at a methodological junction that would benefit from utilising alternative methods when developing and substantiating current claims. By exploring lived experiences concerning the minutiae of PO, this study will have the opportunity to examine more closely the undoubted complexities of the concept provoking further discussion between theorists along with a more intricate tapestry of PO manifestation and development.

Finally, PO has been linked to positive and negative outcomes for both the individual and the organisation. Research has tended to veer towards positive outcomes, although there has been more recent interest in negative outcomes in particular territoriality. This study will consider both positive and negative implications of job-related PO.

To conclude, the above section has provided an overview of PO research demonstrating its close links to theories of possession and ownership, but also its divergences as well as areas that require further investigation.

The following chapter outlines the research methodology of this study, before reviewing the results of this qualitative study.

# **Chapter 3 Methodology**

# **3.1 Introduction**

This chapter will outline the methodological approach undertaken for this study concerning the development of PO in early career professionals. As you may recall, the predominant perspective in PO (Psychological Ownership) research to date has been a positivist stance, with the survey method being the dominant method utilised. As research to date has neglected the social element of PO, concerning how individuals, managers, and colleagues construct PO together, there is an opportunity to broaden PO methodological perspectives. Therefore, this study will approach PO differently, to provide the opportunity to add colour and richness, breadth and depth to causal relationships, providing a multidimensional perspective regarding why individuals might have the propensity to develop PO.

This chapter will outline the philosophical assumptions of the researcher in terms of what they believe is the nature of reality (ontology), epistemology or *"how we know what we know" Crotty (1998, p.3),* and values (axiology), before aligning these to the researcher's theoretical perspective and methodology. All of these inform the method utilised in this study, providing the opportunity to understand the lived experiences of PO development in individuals who have just entered the workplace.

Table one on the following page highlights the researcher's own philosophical assumptions, theoretical perspective and research approach:

Philosophical assumptions / Theoretical perspective /Methods	Researcher Assumptions-Perspectives- Methods
Ontology	Multiple realities
Epistemology	Constructionist
Axiology	Value-laden and Co-constructed
Paradigm	Interpretive
Methodology/Research Method	Qualitative/ Semi-structured interviews
Thematic Analysis Approach	Reflexive Thematic Analysis

Table One: Researcher's philosophical assumptions - theoretical perspective and methods

# 3.2 Research Aim, Objectives and Questions

As stated in the introduction chapter, the aim of this study is to explore the development of work-related PO in individuals at the start of their career, using a sample of work placement students from a post-1992 university.

# 3.2.1 Objectives

- 1) To explore the formation of work-related psychological ownership in individuals at the start of their career.
- 5. To uncover the positive and negative outcomes suggested from the participant's experience of job-related psychological ownership.
- To identify areas where qualitative research can add value to Psychological Ownership studies.
- 3) To provide a framework for how PO may be utilised and developed by organisations and their key stakeholders (e.g., line managers, HR).
# 3.2.2 Research Questions

- 1. How might psychological ownership of work-related targets be constructed by individuals at the start of their career?
- 2. What factors seem to influence the development of work-related psychological ownership targets in this specific group of individuals?
- 3. What do these individuals, line managers and Placement Development Advisors perceive to be the potential outcomes (both positive and negative) of developing job-related psychological ownership at the start of a career?
- 4. How might qualitative studies benefit our understanding of psychological ownership?

### 3.3 Philosophical Assumptions

According to Creswell and Poth (2018) it can be important to understand how our philosophical assumptions influence the research process that we choose to follow. Individuals often make assumptions in the following areas:

**Ontology** considered to be the study of what exists and the things that exist (Lawson et al. 2007) or what we consider to be reality (Creswell and Poth 2018). This researcher considers that multiple realities exist amongst individuals and that this array of perspectives and differences is important to be recognised, especially within their own context (Braun and Clarke 2013; Creswell and Poth 2018). This differs from those who believe there is one objective truth, an identifiable reality or that this notion of reality exists on behalf of those in power (critical) (Denzin and Lincoln 2018).

*Epistemology* is "the theory of knowledge" Crotty (1998, p.3) allowing individuals to ascertain, understand and justify their beliefs about their knowledge – and what counts to us as knowledge. Whilst Crotty (1998) suggests there are a number of epistemological perspectives, he proposes three as the most frequently used: objectivism, constructionism and subjectivism.

*Objectivism* which is closely aligned to a positivist theoretical perspective suggesting a clear separation between meaning and meaningful reality and "belief in total objectivity" (Denzin and Lincoln 2018).

A second widely used form of epistemology according to Crotty (1998) is subjectivism in which a perspective is impressed on a subject/object without any interaction. This is not a world view that the researcher subscribes to however, as they believe that knowledge develops via an interplay between the researcher and subject.

Constructionism suggests that there is not one form of knowledge and therefore that our meaning-making does not rely on one truth, but many different truths depending on how they have been constructed. Via our engagements with others, be they people or objects, knowledge is constructed, and therefore those with this perspective appreciate different ways individuals construct their meanings and reality (Crotty 1998). This is the belief most closely aligned to the researcher. For example, by completing interviews with individuals at the start of their career, and by asking them questions regarding their experiences of the workplace, a researcher can appreciate how they have built their perspective of the workplace whilst still understanding that people can experience the same phenomenon and perceive it in very different ways. For example, individuals in the same department may attend the same meeting but construct totally different perspectives regarding the tone and the intent of that meeting. We are aware of these differences in humans and so it seems appropriate to try to understand what some of these differences about a phenomenon might be. Real life is messy (Shaw et al. 2008; Braun and Clarke 2013), but there is a joy in considering how individuals construct perspectives on different phenomena and by sharing some of their stories, their reality of multifaceted concepts can be highlighted.

Finally, **axiology** relates to the role of values in our research. As someone who suggests that knowledge is constructed, the researcher also acknowledges that meaning may involve the researcher's own values in their interpretation and so therefore needs to be aware of their own role in this research. The researcher subscribes to the perspective of Braun and Clarke (2019) centring researcher subjectivity as part of the reflexive thematic process

and applying transparency around the process, rather than suggesting that any individual can be fully objective. Reality is co-constructed between researcher and participants, highlighting the multiple perspectives that can exist regarding any phenomenon (Creswell and Poth 2018). This subject will be considered further in the reflectivity and data analysis sections.

# **3.4 Theoretical Perspective**

Leading on from a researcher's philosophical assumptions, Crotty (1998) suggests the researcher's theoretical perspectives inform the methodology used within a study. Other researchers, such as Creswell and Poth (2018) include these interpretive frameworks as part of the philosophical assumption section or paradigms (Denzin and Lincoln 2013), which this researcher concurs with.

This study has adopted an interpretive paradigm acknowledging the role of the researcher within the study who wishes to understand and interpret individuals' experiences of psychological ownership. By interpreting individual and group meaning within specific contexts there is the opportunity to build knowledge regarding how individuals construct meaning at a particular point in time. There is an acknowledgement that truth differs between people and over time (Braun and Clarke 2013).

Other theoretical perspectives include positivism and post-positivism, relating to traditional scientific methods used by those who believe there is a logical and objective truth (Braun and Clarke 2013; Creswell and Poth 2018), critical inquiry, feminism, and post-modernism (Crotty 1998).

### 3.5 Methodology: Qualitative Approach/Research Methods

This study provides us with an opportunity to engage with the lived experiences of early career professionals and see the factors that may strengthen / inhibit PO development in this particular context and understand how it affects these individuals' working lives. Maxwell (2012 cited by Mertens 2017 p237-8), suggests that qualitative research allows individuals to add processes, differences and structure to causal relationships, sometimes termed as "the meat around the bone", allowing us to see how individuals' experiences differ, whilst still sharing some similar characteristics. Meaning making is often formed with others, be it from their upbringing and early socialisation to the cultural norms that influence how we live and work. In the context of this study, the researcher believes that these early career professionals are joining an organisation with their own individual history of work beliefs and attitudes (shaped and influenced by family and friends' perceptions of work, experiences within part time work and general career-related rhetoric, which lecturers, placement staff and the media frame around their lives). Qualitative research allows us access some of these many individual realities, allowing us to co-create knowledge with them in their specific context (Braun and Clarke 2013; Creswell and Poth 2018).

Brinkmann and Kvale (2015) use the analogies of a miner and a traveller to link interviews and epistemology with the miner representing a positivist approach of seeking gold or truth, whereas the traveller in a new country will ask questions and seek out stories to build knowledge, thus entwinning information and analysis to construct a sense of place. The traveller may have some existing knowledge of a country at the start of their journey, but like interpretivist researchers, it is through interactions with others that knowledge and analysis entwine, are developed, sometimes providing the unexpected.

Interviews have been described as "professional conversations" with "structure and purpose" (Kvale 2007, p.14; Brinkmann and Kvale 2015, p.3) that try to ascertain meaning and perspectives on a particular topic. Semi-structured interviews are the most dominant form of interview (Braun and Clarke 2013), although Miller and Glassner (2011), in Silverman (2011), suggest that some are wary of the likelihood of it truly capturing the social world. Whilst an interview may not mirror our social world, it does provide opportunities to collect meaning, knowledge, and perceptions.

Interviews are best used for the exploration of "understandings, perceptions and constructions of things that participants have a personal stake in" (Braun and Clarke 2013, p.81) and therefore fit well with the topic of PO. This will be discussed further in section 3.8.4

Other research methods were considered. Ethnography in its initial form using direct observation (Silverman 2011) was considered because of the chance to

observe individuals in their natural environment and the daily rituals that become part of their working life. This was rejected for two reasons; the main reason being that this was the participants' first job role and to be watched by an academic from their university might have been unnerving with an uneven power dynamic. Secondly, many junior business roles consist of work which is mainly online and, in some companies, working from home was the norm, thus limiting occasions to observe the social construction of PO. Nevertheless, organisational ethnography may be a suitable method to use in the future, with a different research population, and using organisations where there is greater social interaction, such as hospitality, retail, and healthcare.

### 3.6 Data Collection

The data collection process used in this study will be discussed in more detail below. However figure two below provides a timeline of the data collection and analysis:



# Data Collection and Analysis Timeline

Figure Two: Data Collection and Analysis Timeline

The following sections describe the data collection process for this study including data sampling, sample composition, the data collection process of interviews and finally triangulation.

### 3.6.1 Data Sampling

Sampling is often contested within qualitative research with Emmel (2013) suggesting that it is not useful terminology, whilst other debates concern sample size numbers and the notion of saturation (Patton 2002; Braun and Clarke 2021c; Staller 2021). Some of these debates seem to refer in part to epistemology and how the overriding perception of objectivity and positivism still skew the perspectives of qualitative researchers.

Purposeful sampling is suggested to be a strong element of qualitative research, as researchers are purposefully looking for certain individuals to participate in research, rather than trying to eliminate bias (Patton 2002). Information richness is the desired outcome, rather than a desire for objectivity and generalisations. In this study, a large sample of individuals with different levels of work experience would not help discover how PO might develop at the start of a career and so, the decision was made to focus on a particular group of individuals in the workplace.

As referred to previously, some researchers may suggest the first graduate job as the pertinent time to undertake research into early careers. However, working in a post-92 university where work placements are the norm, and having insider knowledge of the impact of work placements on individuals, this felt a more significant place to ascertain how PO might develop in individuals (further discussion on page 83). As with all studies, there are limitations, and these will be discussed in S3.8.3. A strategy of purposeful sampling utilising maximum variation and intensity sampling was desired within this particular university population.

Maximum variation purposeful sampling tries to provide a range of different participant perspectives, whereas purposeful intensity sampling offers strong examples (Staller 2021). Palinkas et al. (2013) suggests that combining forms of purposeful sampling can be beneficial if it fits with the aims of the study. This study wishes to explore the formation of work-related PO and therefore the researcher wishes to see a range of perspectives, including those of participants who are towards the extremes of this participant group, where experiences might be quite different (Patton 2002). The researcher therefore

chose to utilise a purposeful sample combining both maximum and intensity variation of WPS participants. This was felt to be achieved by the completion of participant WPS 19, although as an additional participant had already been scheduled, it felt pertinent to complete this interview as well

### 3.6.2 Sample Composition

Sample sizes for qualitative studies have been debated with Patton (2002) suggesting there are no rules and Creswell (2015) stating that often qualitative research sampling takes a random form, as researchers have a prior idea in their mind regarding the sample size, which they stick to throughout the research. Often the sample size is related to saturation, which is suggested to be the point where no new information is yielded or there is information redundancy (Lincoln et al. 1985). This point is disputed by Braun and Clarke (2021c) for use in thematic analysis (TA) and reflexive TA, arguing that reflexive thematic analysis (TA) differs from coding reliability and codebook TA in part because of its "messy", organic form. They suggest that the move to defining saturation in one's sample (be it 12 interviews (Guest et al. 2006); 9 interviews (Eynon et al. 2018) or 7 interviews, as suggested by Constantinou et al. (2017) may have been a well-intentioned form of proposing validity in qualitative research and a pragmatic approach to funding requirements (Sim et al. 2018). Yet by focussing on such a form of validity does not take account of the different forms of TA and if saturation is thereby an effective form of validation. Sample size debates in qualitative studies seem to occur regularly with proposals of solutions (Fugard and Potts 2015) or questioning current practices (Sim et al. 2018). Debates are spirited, yet there is no congruence, with a mixture of ontology, pragmatism, and rules of thumb (Blaikie 2018; Sim et al. 2018; Braun and Clarke 2021c). Emmel (2013) suggests that asking about sample size is asking the wrong question, which Guetterman (2015) concurs with, suggesting "how" and "why" might be stronger approaches.

In addition to the WPS, triangulation was desired to provide different perspectives regarding the experience of workplace PO. Therefore, three participant types; Work Placement Students, employers and Placement Development Advisors, were interviewed, providing an opportunity to see ways

in which PO might be constructed at the early stages of a career from a variety of different perspectives.

Whilst the PDA may be an outsider to the organisation of the WPS or employer, their role situates them within the placement experience through their structured meetings with the WPS / employer and through providing onplacement support. In addition, they all had experience of working with several cohorts and could see the patterns that emerge in ECPs.

One of the study limitations is the lack of employers willing to participate in the study. Those WPS who were interviewed asked their supervisors if they would be willing to be interviewed and the researcher sent an email follow up to five employers who expressed an interest, but only three employers agreed to be interviewed. This will be further discussed in the limitations section, but does demonstrate the challenges faced by researchers accessing some participants.

Tables two and three show the composition of the sample of participants in this study in more detail. The main participants are the WPS, due to the desire to ascertain how PO develops in early career professionals. However employers and PDAs were also included to provide different perspectives of the construct:

- Work Placement Students (WPS) 20 participants from 15 different organisations
- Placement Development Advisors (PDA) 3 participants
- Employers (E)- 4 participants at 3 organisations.

The university participants involved in this study comprise few international students and consist mainly of white students, and so neither ethnic diversity nor nationality are included in the table. Whilst there is maximum variation of sampling within the study in terms of gender, different organisation types and different job roles, it must be acknowledged that one of the limitations of the study is the lack of racial diversity (this will be discussed further in the limitations section). The decision was taken not to include race or nationality in the table, as those participants might be immediately recognised and so for

confidentiality reasons this has not been included. Two students identified their Additional Learning Needs during the interviews and this information has also not been categorised in the demographic information for similar reasons to those above. As Dodgson (2019) suggests, we need to be transparent about the context of the participants, including demographics, power relationships and our responses to participant. Therefore, it is important to be clear that this samples are not representative of university students in the UK and because of the level of their education, it is not representative of all early career professionals. Rather this study provides an in-depth snapshot of a particular group of individuals during a particular time period.

As highlighted, a purposeful maximum variation and intensity sampling was employed. The researcher continued her interviews over a second academic year to try and build the maximum variation in the WPS participants with some success. Intensity sampling was successful across the range of participants. It was challenging to find employer participants, with the majority of employers stating that they were too busy to be interviewed. Nevertheless, even within these participants there is a variation and intensity to provide the required complexity and richness of data, and so a pragmatic decision was taken given they were not the main focus of the study.

The employer participants worked in the South of England for a large technology company, a small Finance house and a University. Three participants were female, one male and all were white. Two participants were relatively senior in their organisation, whilst two would be considered middle management. All had worked at their organisation for at least six years. The PDAs consisted of two female and one male participant, two born in the UK and one overseas. The researcher had worked in the same department as one long-established PDA, but had little prior knowledge of the others. Prior to the interviews, whilst the researcher was excited, as this quote demonstrates there were also some nerves:

"These are different types of participants and I am using different questions and so have that initial anxiety about starting again. I am completing all three interviews on the one day almost back to back and am quite excited about what they will have to say. I used to work with one of the

participants, but do not know the others particularly well, so I am sure there will be a different dynamic."

Researcher thoughts prior to PDA interviews

However, the concerns proved to be unfounded and the interviews proved to be helpful to triangulate some on the perspectives of the WPS and employers.

Participant Number (1-10)	Gender	Course	Participant Number (11-20)	Gender	Course
P1	М	Business Studies	P11	F	Business Studies
P2	М	Business Studies	P12	F	Business Studies
P3	F	Comms & Media	P13	М	Business Studies
P4	М	Comms & Media	P14	F	Business Studies
P5	М	Business Studies	P15	М	Marketing
P6	М	Business Studies	P16	F	Business Studies
P7	F	Business Studies	P17	М	Accounting & Finance
P8	F	Business Studies	P18	F	Business Studies
P9	М	Business Studies	P19	М	Business Studies
P10	F	Business Studies	P20	М	Business Studies

Table Two: WPS Participants

Participant	Gender	Organisational Sector	Role Area
E1a	F	Education	Project Management
E1b	F	Education	Project Management
E2	F	Technology	Recruitment & Engagement Lead
E3	Μ	Finance	Director
PDA1	F	Cover all sectors	Placement Support
PDA2	Μ	-	Placement Support
PDA3	F		Placement Support

Table Three: Employer and PDA Participants

### 3.6.3 Data Collection Process

The WPS were contacted via an email asking them to participate in a study on psychological ownership and all those who replied were offered an opportunity to be interviewed. The PDAs relating to these courses were also approached and agreed to be interviewed. As mentioned previously, it was more challenging accessing employers and only four employers from three different organisations agreed to be interviewed. As discussed above, there were variety and intensity within the employer sample which did allow for a range of perspectives. The result of which was the data collection took longer than expected, starting the pilot in December 2018, with completion of all interviews by February 2020.

### 3.6.4 Interviews

Interviews are suggested to be a craft by Brinkmann and Kvale (2015) which requires learning best practice. Whilst the researcher has been a recruiter in a previous job role and so has experience of drawing information from individuals in a formal setting, prior research regarding research interviews was invaluable.

Interviews are often categorised in three forms:

- Structured interviews, in which pre-determined questions are set by the researcher.
- Semi-structured interviews, in which there are pre-determined questions, but there is flexibility for the researcher to ask other questions, if the participant raises an unanticipated point.
- Unstructured interviews, where the direction of the interview is led by the participant, although the researcher may steer the interview at times to cover key themes.

As mentioned previously, semi-structured interviews were chosen for their rigor and their flexibility. The researcher based their interviews around responsive interviewing Rubin and Rubin (2004), due to the flexibility of the process in which steps within the process are not fixed.

An interview guide was developed in the suggested brainstorming fashion (Smith et al. 1995). The questions were then considered against the research questions, with questions grouped with potential follow-up questions, and special attention paid to phrasing and opening and closing questions. The questions were then tested via a pilot with a WPS who had recently completed their placement and was about to enter their final year of studies. One of the key benefits of a pilot study is to assess your research methods and make adjustments where required (Morse 2015) and this proved to be effective when assessing the interview schedule. The pilot confirmed that the broad open-ended questions were effective, but resulted in some adjustments, mainly relating to the inclusion of follow-up questions to ensure greater depth of answers, and to ensure the researcher didn't miss any key areas (appendix 3). The results in this pilot study were not included in the final study, as on reflection it was clear that this pilot lacked depth in comparison to later

interviews. It was however beneficial in ensuring a more robust interview schedule.

Constantinou et al. (2017) suggest that qualitative research can be deductive as well as inductive. Whilst this study demonstrates a far greater degree of induction, by asking broad semi-structured questions, which allow the interviewee's responses to emerge, such as "Tell me about your job role", there were also questions which implied a level of deduction involved such as "As you know my study is about psychological ownership. What does ownership in the workplace mean to you?"

Because the researcher has completed a literature review on the topic before the interview schedule was compiled, there is prior knowledge and beliefs about the topic, which should be acknowledged by the author (Braun and Clarke 2021c). These include knowledge regarding PO motives and routes. There are also prior beliefs that need to be acknowledged, related to feelings that there were gaps in the literature regarding the overall conception of PO development, particularly in early career professionals. Whilst the researcher did not have beliefs about what this missing information might be, they felt that the role of others in PO construction may have been underplayed (other than in discussing leadership). The researcher also believed that PO might have manifested itself in individuals prior to the start of their placement, hence question 12 in the interview schedule (appendix 4). Nonetheless, whilst the researcher had completed a literature review in advance, because previous research has predominantly used a positivist approach, the researcher was able to develop mainly broad questions to ascertain the lived experience of psychological ownership with the intention of hearing participant's individual stories. In particular, the "what" and "how" questions regarding PO development fell into this category?.

Interview schedules were also adapted for both the Employer questions and the PDA questions (appendix 5 and 6). In both instances, the questions were like those of the WPS and tended to be broad in nature initially, before narrowing down specifically to questions about PO.

Participants were interviewed in a location of their choice, be it their workplace, the university or, if a great distance was involved, online. Those who were

interviewed online were used to using workplace technology (even before the Pandemic) and as you can see from the extract below from the researcher's journal, whilst there were some initial concerns about including online interviews, there were many very positive experiences:

"This was my first Skype interview and so I was concerned beforehand, if it would record properly and if it would be as personal. I didn't need to worry however as both elements worked well. In fact in many ways this interview felt more personal." Researcher

Nonetheless, for all interviews this is an unnatural setting, in which some participants may exaggerate claims or make statements to please or satisfy what they believe the interviewer was looking to hear, rather than giving an accurate overview. Some of these interviews could be defined as "acquaintance interviews" (Garton and Copland 2010) in which there is some prior relationship. The researcher taught some of the students, although they were part of a large cohort (P1, 2, 5, 6,7, 8, 9, 10, 11, 12, 13, 14, 16, 18, 19, and 20) of 300 students and so the degree of prior relationship with the majority was quite limited. The researcher did have further interaction with P7, 9, and 14, who had been Seminar Reps. The researcher had taught P15 in a smaller cohort, but had not taught or met P3, 4 or 17 in advance of the interviews. The researcher has considered the question of "acquaintance" interviews" in some detail, to see the effect, but feels that in this study there was very little difference on the whole between acquaintances and "interviewing strangers" (Braun and Clarke 2013). What feels more pertinent was participants' self-confidence and chattiness. So, P 1,3, 4, 9, 10, 11, 12, 13, 15, 17 were very talkative. In fact, P4, with whom the researcher had had little previous interaction with, talked continuously, whilst P6, 8, 14 and 16 were more challenging. Below is an extract from the researcher journal regarding P16:

"This participant was very nervous and fidgety - I don't feel I did enough to relax her in the first instance. When doing Skype interviews, must remember to take the time to do this first of all. I was tired and that may have come across" Researcher

This experience was unusual, as Skype interviews had been successful to this point and the researcher had interviewed a lot of participants (both face to face and online). It demonstrates the importance of settling your participants when opening interviews, to set the scene and make everyone comfortable. However, Braun and Clarke (2013) suggest that even the most skilled interviewers can have bad days, and this unfortunately was one of them.

It must be acknowledged that there is a power dynamic in any interview with the researcher as an academic and interviewer had greater power. Therefore, it was the researcher's desire to make the participants feel comfortable. Some of this was completed in advance of the interview, by asking the participants to choose where they would be interviewed and by reminding them about confidentiality and research ethics. Then at the start of the interview, the researcher spent time on "small talk" and the first question which was an ice breaker question. Whilst part of the reason why P6, 8, 14 and 16 may have been less talkative may have been due to the power relationship, the researcher believes that it was mainly due to the artificial setting. Whilst some individuals are very comfortable talking about themselves, others may struggle, partly due to the unusual context, but as mentioned previously partly due to their own confidence.

The interviews with the PDAs and the employers all went smoothly, perhaps because they were interested parties and there was less of a power dynamic. One employer interview contained two participants which could have been a challenge. However, both were very respectful of each other's views and it was interesting to see the areas of agreement / disagreement and where they reminded each other of WPS involvement. Below is an example of this in the transcript for employer one:

E1a: "we would have been completely imbalanced, so unfortunately I explained to XXX why I couldn't involve her in the interviews"

E1b: "XXX did the questions though, didn't she?"

E1a: "Yeah. She was involved in everything except the interview".

All participants were sent an email after the interview to express gratitude and ask if there were any additional points that they wished to raise. No additional information was forthcoming. Transcribing interviews proved slow, as the researcher was a part-time student, and it felt too late to send a follow up email with the interview transcript. There are positive and negative outcomes to sending out interview transcripts to participants. On the one hand, it gives the participant the opportunity to review the conversation and make any adjustments. However individuals can also be dismayed by their lack of fluency and/or want to make changes, which alter the data integrity (Hagens et al. 2009; Irit 2011). It is suggested that participants reviewing transcripts can present as many challenges as advantages.

#### 3.6.5 Research Ethics

The university lays out an ethical code of conduct which the researcher followed, as well as regularly referring to other sources such as Creswell and Poth (2018) who provide an overview of ethical issues at each stage of the research process. Early University ethics approval was gained by submitting the ethics checklist (appendix 1), including a Participant Information sheet disclosing the nature of the study and implications relating to management of data and the Participant Agreement form (appendix 2) that are provided to participants prior to the start of the interview. There were different interview questions for the WPS, the PDA and the employers and so ethics checklists were submitted separately for each as denoted in the timeline (figure two). All participants were aware in advance that the interviews would be recorded. All participant agreement forms were signed and returned before the start of the interviews to ensure informed consent and participants given the opportunity to discuss any questions either by email or before the interview. No questions or concerns were raised. Whilst this study was deemed "low risk" by the University Research Ethics process, it was important to always be sensitive and aware that individuals are sharing what could be personal and sensitive information to them. Data protocols were followed regarding data storage and anonymising data such as names and organisations was important. Pseudonyms were used in the transcribed data, this thesis and other research publications.

As highlighted previously, information relating to nationality, ethnicity and additional learning needs have not been included because that might make participants identifiable.

Further information relating to the trustworthiness of research can be found in section 3.9.

### 3.6.6 Triangulation

There are suggested to be four forms of triangulation (Denzin 1978; Patton 1999), although this study used only one form: data source triangulation, in which participants from different groups were interviewed, in this case WPS, employer's and PDAs. Whilst in many instances triangulation is used to corroborate viewpoints to find a single truth, in this instance it was used to build a greater depth of understanding of PO from multiple perspectives (Braun and Clarke 2013). The PDAs have experience supporting many WPS during their placement year, helping them reflect and articulate their skill development, whilst the employers have experience of managing early career professionals. Therefore, both parties provide another perspective of PO, which may occasionally corroborate viewpoints with the WPS, but can also suggest alternative viewpoints which add to the richness of the data. As mentioned previously, the researcher struggled to engage further employers in the study which would have been ideal. Nevertheless, the intention was always to focus on the WPS participants and their experiences of PO.

### 3.7 Data Analysis

This study used reflexive thematic analysis (Braun and Clarke 2019, 2021a) to interpret the collected data. Thematic analysis (TA) is an umbrella term used for a broad range of approaches, techniques, or methods (Terry et al 2017), which allows for different philosophical assumptions. The commonality between all is the desire to make sense of patterns in data.

When considering data analysis, various approaches were deliberated. Template analysis (Brooks et al. 2015) was initially considered

due to its easy adaptation to different studies. However, one of the key features of template analysis is the use of "priori themes" which whilst perhaps providing some direction to the research, do undermine this study's desire for a mainly inductive approach. Discourse and Conversation Analysis were also rejected, because, whilst knowledge is constructed especially within discourse analysis, its critical realist stance and its rigid coding and counting (Potter 2011) provide a very different philosophy to that of the researcher. Conversation analysis is more aligned to natural discourses within everyday life and therefore was not appropriate for an interview situation (Braun and Clarke 2013). Content analysis was rejected due to its focus on some of the more positivist perspectives towards data which can often involve coding data numerically Wilkinson and Kitzinger (2017 cited by Willig and Stainton-Rogers 2017).

Braun and Clarke (2006, 2013, 2014, 2016; 2017, 2018; 2019, 2021b) have written prolifically about thematic analysis, raising the profile of TA, but also helping to categorise various forms of TA. Kidder and Fine (1987) initial categorisation indicated two forms of TA: Big Q, whose basis is more inductive and interpretative, falling within a "qualitative paradigm" (Braun and Clarke 2021a, p.329), whereas Little Q relates to more "hypothetico-deductive research" Locke (2004, p.303) closer to a positivist paradigm. Due to the recent increase in popularity of TA, Braun and Clarke (2019) have sought to further develop these categories, suggesting three main typologies:

- Coding reliability TA, emphasising an objective/unbiased, very structured form of qualitative research often with an emphasis on deduction.
- Codebook TA, considered the broadest TA category with a qualitative paradigm and early theme identification.
- Reflexive TA, which relates to an interpretative reflexive approach where subjectivity and context are acknowledged as part of the process. Coding is more fluid and organic with themes actively created encapsulating numerous facets. Whilst this is a mainly inductive form of analytic process, it is recognised that it is almost impossible for researchers to come from a purely inductive stance as

they may have read literature on a topic and thus accept some level of deduction may have informed their research.

These typologies demonstrate the different perspectives towards TA, suggesting a broad umbrella term, where different epistemological stances result in very different ways of analysing qualitative data, albeit emphasising that all TA analysis involves seeing patterns. Reflexive TA was chosen however because of the clear links to interpretivism, where the researcher plays an active part in theme creation or via building interpretive stories. Themes in this instance are described as "themes as patterns of shared meaning, cohering around a central concept" (Braun and Clarke 2019, p.551) and have been compared to a sculptor actively shaping a block of marble. It is the researcher who shapes the themes, acknowledging their personal subjectivity, but seeing it as a resource, rather than a hinderance. Reflectivity is acknowledged as being a key engagement within data analysis and interpretation.

Whilst they suggest six phases of TA (Braun and Clarke 2006, 2013; Kim and Beehr 2017; Braun and Clarke 2021c), at times these stages can merge, because of the iterative nature of this form. Transparency on why researchers do what they do is important because of the subjective nature of reflexive TA. Terry (2017) suggests therefore that coding can be strong or weak, not right, or wrong. Below is an overview of the process for this study:

**Phase One: Data familiarisation.** An initial read through of all hard copy transcripts was completed, to start early familiarisation and make notes on areas of interest. An example of data familiarisation on a transcript relating to participant 11 can be found in appendix 7. During data familiarisation, there is no coding, but rather a consideration on the possible meanings of conversations, along with early identification of commonalities and differences emerging. All WPS transcripts were read together followed by the PDA and employer transcripts.

**Phase Two: Generating codes** was then undertaken via NVIVO. Complete coding was used (Braun and Clarke 2013) in which pertinent codes to the research questions were identified and data was attached to multiple codes (see an example of multiple codes for one passage for Participant 11 in

appendix 8). Both small and large chunks of data were coded, to one or multiple codes. There was a mixture of "semantic", surface codes (Braun and Clarke 2013, 2021a) such as "thrown in the deep end" which related to the language used by participants and latent more "conceptual" conceptual codes (Braun and Clarke 2013, 2021a) such as self-efficacy, which were influenced by the researcher's knowledge of the topic. To ensure a thoroughness of approach, data were multi-coded where appropriate. So, for example data concerning "trust" was initially coded to a code called "trust", but also to any code which related to its environment, such as "line manager", or "organisational culture". This allowed an understanding of the context of the data and how participants had constructed their view of "trust" and to provide a "checking system" to ensure a robust and thorough approach. In total, 131 codes were developed (appendix 8), although some of these codes were very broad, because of the multi-coding. There were some surprises, particularly around the challenges of PO, as these were somewhat different to previous studies. This mainly related to the context of this group of participants. For example, "territoriality", which is a widely posited outcome of PO, was rarely identified, whilst feeling "the weight of ownership" was mentioned more frequently, compared to the PO literature. Nevertheless, there was always a possibility that differences might be identified, given these participants were situated within an early career stage of their working life, unlike most participants in previous studies who were more experienced.

**Phase Three: Generating initial themes** involves actively identifying patterns from the codes. In appendix 9, you can see how the codes were clustered together or combined where necessary. As this was a reflective process, the following stages were not clearly defined in a codebook form, but are more aligned to reflexive TA's interactive questioning of the data, demonstrated through the researcher's journey to move from Phase 3 through to phase 5. Some codes such as "psychological safety" were listed across three categories early on, denoting how challenging the process can be at the early stages of analysis. On paper the process sounds very logical, but actually it was perplexing, frustrating and challenging at times. The researcher did follow Ripley's (2011) advice "on returning" and left time during analysis to contemplate, often via long walks which is the researchers own "restorative

space" (Rhew et al. 2021). Whilst juggling a PHD alongside work can be challenging, sometimes the gap between stages was helpful thinking time! Some codes, such as "additional learning needs" resulted in very little coding and did not relate to PO, but rather to the placement and so were removed. From this point. It became easier to see some potential themes and an initial thematic map was produced (appendix 10) with the themes. Whilst reflexive TA does not usually include tables, for ease of understanding, I have included a table to show how the codes link to themes.

Nonetheless, after careful consideration, the themes in appendix 10 showed some overlap with "Socialisation" "Organisational Influence" and "What is Ownership", demonstrating codes that could fit into any theme, such as the code "rituals".

Phases 4 and 5 involves **reviewing and defining themes**, as it is suggested to be rare for first attempts to result in a final thematic map (Terry 2017). This was certainly true in this instance. When reflecting on codes, it became clear that there was a social construction to ownership for these early career professionals, which involved them being "given ownership" in some instances and then "accepting ownership" or "taking ownership", if it wasn't given. At this point these became two important themes. In addition, on consideration, it was clear that PO Routes and Motives seemed to have been lost in phase 3, so they were reintroduced. "Strength of PO feelings" was also introduced to capture the different levels of feelings that individuals might have. PO outcomes and Future Careers remained, due to their clear links to the research questions.

Appendix 11 demonstrates the final stages of actively generating themes. The PO motives and routes fall under the themes of "taking ownership", so they were combined. PO outcomes was renamed "Pleasure and Pain" as it was felt to better represent participants feelings towards ownership. Finally, future careers were renamed "longevity or liquidity", because the picture is far more complicated than the theme "future careers" suggested.

Phase six relates to **writing the report**, which is seen as the final stage of the process. Eldh et al. (2020) suggest that there are limited journal articles which define the purpose of including quotations or a justification for their inclusion.

Consequently, there is very little guidance regarding a framework for quotation inclusion. Patton (2002) suggest quotes are essential to hear the participants' voice and concurs with Sandelowski (1994) that they are illustrative of findings. There is also concerns relating to the ethics of anonymising quotes and stating a participant number, along with research integrity in not picking controversial quotes if they are not relevant (Eldh et al. 2020). For this research, the author thought participant voice was incredibly important and has included some longer quotes if they illustrate findings. The intention was to include all participants, but inevitably there are some participants who provided better illustrations of the phenomenon than others, because of their experience of PO. The researcher was careful to ensure that quotes were not overused and their words were not changed. Participants were given a number and their names were never used (Eldh et al. 2020). The final thematic map can be seen in figure two below (a more detailed version, with the codes linking to themes, can be found in appendix 11 and links to the research questions as follows:

- How might psychological ownership of work-related targets be constructed by individuals at the start of their career? Relates to themes "giving and taking ownership"
- 2. What factors seem to influence the development of work- related psychological ownership targets in this specific group of individuals? Relates to themes "giving and taking ownership"
- 3. What do these individuals, line managers and Placement Development Advisors perceive to be the potential outcomes (both positive and negative) of developing psychological ownership at the start of a career? Relates to theme "pleasure and pain"
- How might psychological ownership influence future career development plans in early career professionals? Relates to theme "liquidity and longevity"

# Phase 5:Refining, defining and naming themes



Figure 3: Final Themes

These themes will be discussed in more detail during the analysis and findings section.

Whilst the overall structure of this analysis looks ordered, it was time consuming, messy, iterative and perplexing at times (see appendix 12 for one example of the many hand-drawn mind maps that I put together). Nevertheless, it was also fascinating, thought-provoking, inspiring and sad to remind myself of their stories, and I feel privileged that the participants were so honest and generous with their time.

# 3.8 Trustworthy Research

Trustworthy research is one of the cornerstones of academic research and differs significantly in qualitative and quantitative research due to different theoretical perspectives and methods (Williams and Morrow 2009). They suggest three elements relating to trustiness of qualitative research:

- Data integrity
- Balancing subjectivity and reflectivity
- Clearly communicating findings

This section and the following section will discuss these three elements starting with balancing subjectivity and reflectivity.

### 3.8.1 Reflexive Thematic Analysis: Researcher Perspective (subjectivity)

The researcher's own subjectivity plays a part in qualitative research and particularly within Reflexive TA. Reflexivity:

"means turning of the researcher lens back onto oneself to recognise and take responsibility for one's own situatedness within the research and the effect it may have on the setting and people" (Berger 2015, p.220)

This 'looking at oneself' relates to the whole research process from the type of research questions formed, through to the final analysis and writing up of papers. By continually reflecting throughout the process, individuals can be aware of their biases and use them as a frequent reminder of one's own complexity.

Within this study, my own prior experience of working with individuals at the start of their career on placement and graduate programmes means that I have some knowledge of early career professionals. I was therefore interested in studying this group of participants, because I had seen some ECPs quickly demonstrate ownership feelings and I was intrigued on how some individuals might develop PO, whilst it was less observable in others. I expected other researchers to share this interest, but to date this is still an under-explored area. In a way, this was exciting, as it was an opportunity to do a different study, but also frustrating, as I had to complete this doctorate reviewing research from a very different perspective.

I realised very early on that I was interested in the social construction of PO and that influenced my research questions, to emphasise the interest in the initial building blocks of PO. I was interested not in the existence of PO, because my own background and experiences told me it did exist, but rather how it is constructed in the workplace. What elements might be important and how does it manifest in people and between people?

I have always had strong feelings of ownership (even though I was not aware of the theory of PO) starting with my "Saturday" job at the age of sixteen when I was made a supervisor and so when I first read about this topic it resonated clearly. I know it had to be "mine!"

What was interesting however was to hear about the challenges of PO. I have been situated in PO all my working life, but it was only after hearing others' stories that I realised the burden of PO. Whilst I am at a different point of my career, some of the stories reverberated uncomfortably in my thoughts. I started this research with perhaps rose-coloured glasses framing my personal thoughts about PO. By reflecting throughout the process and having the time to think, I now believe I have a better-rounded view of PO. This study isn't about me, but from the participants' words, I have learnt more about my own experience of PO. They have a slightly different context, but I believe that some aspects have resonated at different time periods throughout my own career, including now. Nevertheless, there are also differences which continually emphasised to me regarding the complexity of PO. There seems to be a smorgasbord of influences and rather like a recipe, if you add different weights of components at different time periods, the outcome will vary.

I positioned myself during the interviews as an "insider-outsider". My experience teaching many of the participants about the placement process made me conscious of my insider status and positioned me somewhat as an "expert", whilst at the same time, I was an "outsider" in their world of work, and I was grateful the WPS welcomed me into their world. Early research regarding insider-outside roles in qualitative research was often characterised by three roles: Peripheral, active, and complete members (Adler and Adler 1987). This doesn't quite characterise the position I found myself in, whereby I was a "powerful insider" due to my teaching role, a "knowledge insider" due to my experience of the workplace, yet a "workplace outsider" due to my lack of knowledge about their WPS experience and an "experienced outsider" because I was at a different point in my career. I could also have been seen as an "ownership insider or outsider", depending on the WPS, because of my own experience of ownership. However, I chose not to share these experiences. Dwyer and Buckle (2009) have defined this duality as "the space

between", whereby dialectical differences / similarities are not ignored, but not over emphasised, and this is the approach I tried to harness.

By acknowledging my subjectivity, the researcher followed Clarke and Braun (2018); Braun and Clarke (2019, 2021a) six phases actively, trying to balance participants' meanings and the researcher's interpretations. The most challenging aspect occurred during the data analysis and the subsequent writing up of the themes from an experience told to me by P15. He had in his words organised a successful event, but on the way home had a car crash. I struggled considering if I should add this to the analysis or if it should be left out, because it felt quite ghoulish. It took me some time to realise, it was the participant's decision to include it, not mine, and it was my interpretation that felt it was ghoulish to include, not his. My role is to examine and interpret, whilst understanding by doing this, my values may creep into the research. This was a good lesson to learn about letting the data talk, rather than for me in the role of researcher to make value judgements. I have always prided myself in believing that I am objective, but like everyone else my previous background and experiences have shaped my thinking, which I need to be as aware as a researcher.

The following section relates to the other posited elements of trustworthiness; data integrity and communication (Williams and Morrow 2009).

### 3.8.2 Data Integrity and Communicating Results

Data integrity relies on dependability and adequacy (Williams and Morrow 2009) which this study has worked to achieve. Adequacy in reflexive TA, as mentioned previously, does not rely on fulfilling sample sizes, but instead focuses on comprehensiveness and variations in the sample (Levitt et al. 2017). As mentioned in section 3.7, there is a suitable level of breadth and variation of experiences within this sample, which includes triangulation from employers and PDAs.

Dependency relates to the clear articulation of research methods and strategies, which this chapter has fulfilled in the data collection and data analysis sections. Clear communications and application refer to opportunities to link relevant literature to the topic and situating the participants, to ensure the reader understands the operating context (Williams and Morrow 2009). This is required for work to create impact. Hopefully by reading this thesis, this will be achieved.

### 3.8.3 Limitations

This work is context-bound, relating to the experiences of a particular group of individuals at a particular time period. Therefore the intent is not to generalise to the wider population, rather instead to widen the conversation regarding PO development and offer a different perspective. Mayring (2007) asks if generalisation is important or required in a study result, whilst others including (Lincoln et al. 1985; Carminati 2018; Denzin and Lincoln 2018) suggest than within an interpretivist paradigm, the emphasis is understanding individuals, especially within a particular social context. This is a limitation of the study, though future research could be undertaken with other population groups, so that transferability is achieved (Lincoln et al. 1985; Carminati 2018). Hopefully, this is a starting point regarding early PO development and further studies relating to early PO, from both qualitative and quantitative perspectives, would be of interest.

Nonetheless, given the demographic similarities between individuals in terms of education levels, location and race, further studies with greater diversity in these areas would be beneficial. This might include studies with other groups of early career joiners, such as those who work straight after college or recent graduates or individuals from different education systems. Some studies relating to individuals in different industries away from office work would also be beneficial. As these WPS are only working at the organisation for a short time period, it could be said that their mindsets may be different compared to those in a permanent role. Nonetheless, a number of WPS and employers saw their experience as an opportunity to lead to a graduate role via the organisation's 'talent pipeline', so there were reasons to do well alongside the obvious point of gaining some work experience. Nonetheless it would be interesting to see commonalities with any future studies relating to temporary, contract or permanent workers.

There are also geographical and cultural limitations related to early career professionals. Work Placements are not undertaken worldwide and so different countries may have different routes for early career professionals. That said, given PO has seen to be transferable across countries, there may be similarities in other early career professionals.

The researcher would have welcomed more employers to be involved in the study, to provide greater opportunities for triangulation and for providing different perspectives. The challenge in this study was trying to reach participants and employers from the same organisation, which proved very challenging. Morse (2015) suggests that triangulation can occur with two or more data sets and thus with the PDA interviews, the researcher believes they have achieved triangulation. The three employers who participated, demonstrated a variety of perspectives towards ownership, which was partly due to the organisational culture, but also related to their own viewpoint. Notwithstanding, it is acknowledged that whilst this study was keen to ascertain the perceptions of early career students, further studies with both employees and supervisors would be beneficial in the future.

Future research would benefit from multiple interviews with participants over a longer time period to see how PO development may wax and wane. It seems unlikely that PO is entirely stable and it would be interesting to see the factors which influences these changes. This was not possible for this research and is also one of the study limitations.

### 3.9 Summary

Chapter three has outlined the researcher's philosophical assumption as one that believes there are multiple realities with multiple truths. Their theoretical perspective is most closely aligned to that of an interpretivist and both this perspective and their philosophical assumptions led them to develop a qualitative study using semi-structured interviews as their research method. The participants in the study were students from one university undertaking a work placement; supervisors who support work placement students and Placement Development Advisors who visit students in the workplace. Data was analysed using reflexive thematic analysis which acknowledges the

researcher's role in the interpretation of data. The trustworthiness of the analysis was considered essential with data integrity protocols utilised and a balance between subjectivity and reflexivity. The final element of trustworthiness will be ascertained via the Findings section in chapter four.

# Chapter 4 Psychological Ownership in Early Career Professionals; Findings and Discussion

# 4.1 Introduction to Chapter

This chapter aims to explore the findings relating to the development of Psychological Ownership in individuals at the start of their career, using a sample of students from a post-1992 university. Four key themes were developed from this study (giving ownership; taking ownership; pleasure and pain; longevity or liquidity) providing an opportunity to hear the WPS discussing their PO experiences, with employers and PDAs providing additional perspectives and corroboration. The section starts with a reminder of the participant demographics and an overview of how individuals described ownership in the workplace.

Because this predominantly inductive study used qualitative research to understand the real-life experiences of PO, the results do not clearly map to previous PO research, which is mainly quantitative research, but rather reflects the experiences of this group of individuals at the start of their career in a fixed term job role.

# 4.1.1 Theme Overview

The four themes generated from the interviews reflect an active interpretation of the interviews with WPS, employers and PDAs:

*"Giving Ownership"* relates to the ritualistic, relational, and organisational process that permissions an individual at the start of their career to take job-related PO. It is an active, dynamic process situated within organisations relating to three key components: organisational culture, supervisory support and trust, resulting in the permissioning of job-related PO.

*"Taking ownership",* the second theme, can be developed in two instances. The incumbent who has been permissioned ownership, accepts job-related ownership from the supervisor / line manager and takes ownership of the relevant task, project or job role. Secondly, in instances where ownership have not been given or permissioned, the WPS may still actively choose to initiate job-related PO. Forms of "taking ownership" rely on the individual feeling self-efficacy, linking to WPS self-identity or sense of belonging, and having a clear PO route to taking ownership via routes such as control, self-congruity, investing self, some elements of psychological safety or a desire to build intimate knowledge of the target.

Most PO research has focussed on organisational outcomes; however, the "*pleasure and pain*" theme relates to the feelings and outcomes of PO according to this group of participants. There is a duality to PO which in this study seems to imply that the stronger the feelings of job-related PO, the stronger the strength of both positive and negative emotions and outcomes. Because these individuals were at the start of their career, those with strong feelings for PO targets often felt unable to discuss the weight of ownership that they felt on their shoulders.

Finally theme four: "*longevity or liquidity*", discusses how job-related ownership feelings may impact on the future graduate career plans of this group of early career professionals. Within this theme, there will be a discussion regarding an additional PO target which may compete with job-related and organisational PO, that of 'career PO'.

### 4.1.2 Participant Context

In section 3.6.2, the sample composition revealed the demographics of the participants. However there are also differences in terms of the organisational sector, the types of job roles and the size of the company, be it a global organisation or a SME based in the UK. Table four provides an overview of this information. Some organisations hire WPS as part of a formal intern scheme, which often consists of a programme of events together, such as onboarding, group projects and mentoring. Informal schemes are often when there is only one WPS at the organisation and/or they joined the organisation on a fixed term contract. The mixture of job roles reflects the breadth of placement roles often available to WPS:

Participant Number	Organisation Sector	UK or Global Org	Formal/ Informal Scheme	Job Role
P1	Technology	Global	F	Resource Coordinator
P2	Construction	UK	I	Project Manager
P3	Technology	Global	F	PR Intern
P4	Technology	Global	F	Legal Intern
P5	Car Leasing	Global	F	Management Placement
P6	Engineering	UK	I	Operations Assistant
P7	Education	UK	F	Project Support Administrator
P8	Technology	Global	F	Sales Development Intern
P9	Beverages	UK	I	Territory Sales Manager
P10	Retail	Global	F	Manager
P11	Technology	Global	F	Marketing Campaign Executive
P12	Technology	Global	F	Early Professional Talent Acquisition Coordinator
P13	Media	Global	F	Business Continuity intern
P14	Technology	Global	F	Employer Branding & Marketing Co-ordinator
P15	Technology	Global	F	Early Professional Attraction & Engagement Events Coordinator
P16	Transport	Global	F	Human Resources Advisor
P17	Finance	UK	F	Branch Support
P18	Service	UK	I	Commercial Support Executive
P19	Engineering	Global	F	Commercial Intern
P20	Technology	Global	F	UKI Field Marketing Specialist

Table Four: WPS Organisational Information

Participants typically applied for between 3 to around 50 placements with some (P3,P9,P10, P11, P14, P15, P17,P18 and P19) undertaking multiple interviews and assessment centres. Whilst some individuals (P9,P10,P15 and P17) had offers from more than one organisation and therefore had an element of choice, most accepted the first offer made to them and were pleased to have been made an offer. P14 provided a typical example of a WPS who was very focussed on securing an early placement:

*"I was really keen to get a placement early so I remember not being that fussy...I applied to 17 in total...But I secured this role...by Christmas Day" (P14)* 

Some students decided to move away from the traditional list of organisations and looked locally for job roles (P7 and P9) or made an active decision to work for a smaller organisation (P2).

Some participants were very proactive in their placement search (P1, P2, P3, P5, P9, P10, P11, P14, P15, P19), whereas others were less zealous *"I started the search, quite late"* (P6)

Very few had a clear idea of what type of job role or organisation they wanted to work for. Whilst students spoke of frustrations with the recruitment process (feeling the process was impersonal or employers not responding), by the time they joined the organisation, they were positive about the job role / organisation.

Many participants had a formal induction either with other new starters within the organisation or with other WPS students if the organisation hired large numbers of such. P9's induction included a formal rotation around other parts of the business, which was unusual. Within some organisations, roles are continuously filled by WPS students, which often results in a short handover / shadowing period with the former incumbent (P1, P3, P4, P7, P12, P13, P14, P15, P17, P19, P20). This allowed individuals to "learn the ropes" and ask questions informally to their peers.

Other participants who were interviewed for this study are shown in table 5 and are either supervisors who work at organisations that hire placement students or Placement Development Advisors who visit and support the WPS whilst in the workplace, providing pastoral care where required and helping the WPS

reflect on their experience and skill development. These other participants allow us to triangulate the data to provide some different perspectives within the context of this study.

	Organisational	
Participant	Sector	Role Area
E1a	Education	Project Management
E1b	Education	Project Management
E2	Technology	Recruitment & Engagement Lead
E3	Finance	Director
PDA1		Placement Support
PDA2	Cover all sectors	Placement Support
PDA3		Placement Support

Table Five: Employer and PDA roles

### 4.1.3 Participant's interpretation of Ownership

Participants in the study were asked, "what does ownership in the workplace mean to you" Previous studies have neglected a personal perception of ownership, which negates the social experiences of how ownership is lived, but also the different perceptions which permeate people's lives.

A commonly provided answer from the WPS revolved around responsibility, with P7 providing a typical observation:

"Ownership for me I think is having responsibility of someone or something or a project and making sure that it's that the best it can be." P7

Ownership was mainly mentioned in relation to responsibility for tasks (P1,P2, P3, P5, P8, P11, P12, P13, P14, P15, P16), projects (P1, P7, P19), people (P2, P7, P10, P13) including themselves and their career (P1, P5, P8, P15, P19 and P20).

Positive outcomes, such as achievement (P8, P12, P15), adding value (P6), fulfilment (P4) and pride (P3, P8, P12, P20), were included within these perceptions, along with the more negative connotation of owning up if things go wrong (P5, P15, P20).

Four WPS provided slightly different perspectives (P9, P16, P17 and P18), emphasising a sense of belonging or "we" more than other WPS, as indicated here:

"to me it's a sense of belonging in the company, that I contribute directly to the company... it is very much a sense of how I feel connected to everyone else" P17

The PDAs also used the term responsibility to define workplace ownership, although PDA 2 also suggested *"it's basically running your job as though it was your business*" to emphasise a higher level of responsibility.

Responses from employers E1B, E2 and E3 were similar to those of many WPS, referring to the responsibility element *"taking responsibility for a task, for a project, for something and wanting to carry it through to the end"* (E3) although employer 1A referred to the sense of belonging:

"For me it's more about, I'd say it's more about loyalty. So you know being loyal to be you, being loyal to the department, your boss, your work colleagues, that kind of thing. That's what I see it as" (E1A)

Such similarities between participant groups indicate the relational, sociallyconstructed nature of PO, which appears value-laden with societal norms. Furby (1978) in her research of children's development of feelings of ownership for possessions indicated children often role-modelled adults taking care of their possessions and it may be that experiences of ownership with others and especially those more senior may play a part in how PO is viewed.

Sense of belonging is a key motive for PO (Pierce et al. 2001, 2003) and therefore for some individuals (P9, 16, 17 and 18), ownership for them required them to be part of a group. Responsibility has been seen by Pierce et al. (2001); Pierce et al. (2004) as an outcome of PO, rather than as part of the construct. Nevertheless, this does provide some context regarding how PO may be perceived by this group of participants. Whilst these responses are slightly different to Pierce et al. (2001, 2003) definition, what many of the participants acknowledge is the personal demarcation of targets, but also allude to a number of targets which can be both material (a project) or immaterial (the organisation).

### 4.2 Theme One: Giving Ownership; Job-related PO

This section will discuss a key theme regarding the process of job-related PO development, that of "giving ownership" to early career professionals. There were three key elements to its successful formulation (shown below in figure three): organisational culture, the line manager, supervisor, mentor, or other person responsible for the day-to-day management of the WPS (hereon in named as the supervisor) and trust. These will be discussed in more detail below.



Figure Four: Elements that Contribute to Giving Ownership

Job-related ownership was constructed via a relational social process with some supervisors "giving" ownership to their WPS. The act of giving task / job-related ownership was mentioned by both employers ("So literally I give ownership" E3) and WPS ("I think that was due to my manager actually. Full on giving me that ownership" P11) often initially via a building block of tasks. This provides a
ritualistic signal and permission to the employee that ownership has been transferred to them, even if initially on a small scale. Watkins et al. (2015) in their work on digital possessions refer to this as 'transferability', in which ownership is passed, exchanged, or gifted to another.

"Ownership" was a key organisational value for employer 2, and they provided early signalling to participants, via the recruitment process, regarding ownership expectations. During assessment centre activities and interviews, ownership was a measured competency and thus potential new hires were required to demonstrate and provide examples of previous ownership experiences during interviews, as well as showing ownership in group activities:

"during an exercise we would be looking at, is that student....taking ownership of something...ownership of an idea they've had, ownership of being the timekeeper, the mediator, the leader...taking ownership and then following it through "E2

Other organisations also signalled the "giving" of ownership via their website and within the recruitment and selection process. Participant 10 chose an organisation which consistently stated that WPS would undertake challenging work and great responsibility rather than (in her words) a *"nice"* organisation where she also had an employment offer. She perceived their indications online and during the assessment centre to be:

"this year is going to kill you...if you can't do it, that's fine... but we need people that can...I was sort of like...I think I can do that and if I can't, I definitely want to try." P11

This was a highly regarded placement with considerable social status, which allowed a participant with high self-efficacy and who was achievementorientated to relate a role back to their motives / desires.

Consequently, is seems possible for individuals to use the PO route of selfcongruity (Morewedge et al. 2021) even before they join an organisation. For some individuals this early signalling provides opportunities for the WPS to build self-congruity with an organisational target prior to joining and in this instance allowed a WPS to activate the PO motives of self-efficacy and selfidentity (Pierce et al. 2001, 2003).

One large technology organisation was particularly clever with a ritual providing an informal ceremony of incorporation during the WPS induction, which helped individuals understand more about the company's products during a group activity:

"they gave us a song and a XXX product or service and we had to rewrite the lyrics for that which was quite funny to see everyone's attempts, but it just meant that you felt really relaxed" P1

Providing opportunities to attach self and company together at such an early point can only help to speed up the onboarding process and allow individuals to feel that they belong.

In organisations that had an ongoing rotation of WPS, the first "giving ownership" rituals often lay with the outgoing student. During their last few weeks in the organisation, the previous incumbent had often carefully crafted a handover document which specified the role, key contacts, and systems. P1 provides a representative example of the experience of those who had a handover with the outgoing WPS:

*"it was one of the interns, XXX she put together a slide deck and one note file that was all of the information that we could possibly need and we spent a lot of our time with her just going through it and making sure we knew everything"* P1

This is simultaneously part of an initiation ritual for the new postholder but is also part of the divestment ritual for the outgoing WPS and thus is populated with symbolism and meaning. "My role" is transferred from one person to another, providing opportunities to share "their" perspective, "their" insights, and "their" territory. This is intentional ownership transfer (Kalish and Anderson 2011) providing a clear demarcation of one individual gaining ownership rights and another losing their ownership rights. However, whilst this handover could be fraught with territorial, contagion and other negative emotions, most participants found this a positive experience. They gained "insider" knowledge from the current WPS, opportunities to learn the ropes of their new job role and ask, "stupid questions." The outgoing WPS also felt a sense of pride in handing over something that was perceived as theirs "I was leaving...but because I.. really cared about it, and I wanted it to succeed for the next intern" (P11). P14's experience was a common example of the peer-to-peer support found in the student handover process:

"loads of advice. About their year, ups and downs...saying it's okay, don't be nervous, you will get into it...how they handled things, and not to take things personally" P14

New WPS also benefitted from seeing their possible selves and settling any nerves (Markus and Nurius 1986). Employer 2 suggested however that it can be very daunting to see the amazing things that an outgoing intern is undertaking and so reassures the WPS that they will quickly be in a similar position *"3 months and I promise you will feel …where you don't have to check something before you do it."* (E2).

Generally, the perception of these handovers was considered positive by participants, although P13 did suggest that the picture painted by the outgoing incumbent was not realistic.

Once the prior student had given role ownership to the new incumbent or for those individuals who didn't benefit from this form of handover (P2,6,9,10 and 16), it was often the supervisor and occasionally the team who managed the process of "giving" ownership. P2's experience was unusual as he shadowed one of the Managing Directors *"for the first two-three weeks I was just shadowing the MD"* (P2).

P6 by his own admission started his placement search quite late and eventually found a placement via a friend who had worked for the organisation previously. Whilst everyone in the organisation was very welcoming, after the first week he *"hated it"*. With retrospect he believes it was the shock of starting a work placement, but it wasn't uncommon for these new workers to have a crisis of confidence or identity. Both P4 and P10, who were both confident gogetting types, suggested they struggled in their first week *"I think I went home for four out of five days that week in tears"* (P10). The separation rites of a liminal process are often characterised by disorientation and uncertainty, but expediate the sense-making process (Ibarra 1999). As suggested by Mele et

al. (2021) the border zone between education and the workplace can be scattered with meaning-making activities, as the participants come to terms with their new lives and changing personas.

Even P19 who sought out an organisation which would provide him with an elevated level of responsibility demonstrated that fear:

"Yeah, it was scary especially the first few calls and things...it was scary but it was one of those things that I felt ... I am appreciative of it now because I feel like that's why I've managed to learn so quickly" P19

The context of these participants is different from most PO studies and so it is worth emphasising the challenges faced when starting your first key role. Participants mentioned difficulties such as the lack of structure or access to their supervisor hampering their ability to settle in and feel comfortable. P18's manager worked remotely, and she didn't meet her for the first four months with all contact by email *"my colleagues did say recently…..for the first week you looked absolutely petrified"* (P18).

This is a liminal period where the WPS are between two different identities and seek safety and in particular trust and organisational support (Zhang 2020) as reiterated by the PDAs: *"A handover....is important....it's amazing how many companies that doesn't always have that"* (PDA3). This feels particularly pertinent for employers hiring young new employees during / post pandemic as working from home may result in further feelings of isolation and will need a greater level of structure / access to a supervisor to feel safe within their environment.

Ownership of tasks was found to be a means of constructing job-related PO via an ongoing development of the range and importance of tasks, often directed by the supervisor or team. This "building block" approach employed by many organisations allowed supervisors to demonstrate trust in their new charges and to help their new employees understand the business environment with which they were now operating in, using a psychologically safe form. In many ways this process is akin to that described by Furby (1978) when considering the development of feelings of possession and ownership in children. Initially dependent on others to provide them with possessions, they

gradually assert themselves to take control of the acquisition process, before becoming responsible for these possessions.

When considering how "giving ownership" developed, ownership of tasks allowed new employees to gradually take control and cultivate tasks often investing self.

Employer two, whose organisation values included ownership explained the process that was commonly utilised in the ritual of giving ownership after the previous WPS has departed the organisation:

"Although we hand-hold them, they do it, so I will be sat with them, but they'll be the one writing the email...So you have that bit ... where the old intern's left, I then have to kind of bridge the gap for a little bit of really hand holding them through, how would you do this, what would you say." E2

These are life cycle rituals for those organisations who have a continual WPS in roles, with the supervisor demonstrating at an early point that the WPS needs to control and invest themselves into their job role. This was also confirmed by the students within this organisation and other organisations (P1, 5,10, 11, 12,13, 14, 15, 19, 20) demonstrating the pride felt when given larger tasks *"I actually got given for the first time, quite a big task. I had my own job that was just me"* (P11).

Using the building block of "giving ownership" of tasks culminated in WPS and supervisors suggesting that job-related PO took between three to six months. Some organisations (such as employer 2) embedded ownership in their culture facilitating the supervisor / team to "give" ownership, resulting in their new employees developing job-related PO within three months *"But yeah, three months I think broadly"* (E2), whereas in some organisations, where ownership was less embedded, it was closer to six months *"probably about the middle of the placement"* (P7). P17's experience was typical of how WPS built knowledge of their job roles:

"a couple of months.in the first month, getting my footing, in the following month it was developing what I'd sort of grasped...the third month, it was not mastery...I'd learned how to do areas competently by

myself...responsibility fell on me in the workplace and I sort of owned that section of what we do in the office" P17

This broad overview encapsulates how in the minds of WPS, they built knowledge, understanding and confidence during the socialisation's period. Whilst this theme mainly focusses on "giving" job-related ownership, there are also opportunities for organisations to simultaneously facilitate the building of organisational PO in this process. The coming sections will continue to demonstrate how job-related ownership was developed by highlighting three elements which seem crucial when "giving ownership" to this group of WPS; role of supervisor; organisational culture and trust. Nevertheless, there will also be references to instances that inadvertently seemed to expediate the development of organisational PO.

#### 4.2.1 Supervisor Role in "Giving Ownership"

Supervisors, or in some instances the mentor, were instrumental in the development of job-related PO in this group of participants as they often permissioned, gave or denied ownership. When asked about the role of the supervisor, the PDAs suggested it was "huge" (PDA1 and PDA2):

"The best examples... the managers are supportive but not micromanaging um... encouraging the students to take responsibility and to have confidence...and that does come a little from being hands off but there when the student needs it" PDA3

Previous PO literature regarding relational factors influencing PO is scarce, relating mainly to ethical and transformational leadership (Bernhard and O'Driscoll 2011; Avey et al. 2012; Shouse 2017). Given the context of this study relates to early career professionals, studies regarding senior leaders are less relevant to these WPS participants, who lack direct access to individuals in this position. Instead, supervisors or mentors provide a relational link to the organisation and are the most likely individuals to "give" ownership and to act as a role model or possible selves (Markus and Nurius 1986). The process described below, was typical of the process described by WPS on how they were given ownership:

"She slowly started testing me...it started with answer a few emails, you get back to them, you do this and then slowly like, oh, this is the big document you can monitor it" P11

Task ownership helped integrate individuals into their role and build on the handover provided. Task ownership built relational trust between the supervisor and the WPS, with both testing each other regarding the boundaries of the job role and the relationship. *"He sent me to cover an area...on my own...he was hard on me, but in being hard.. I worked a lot harder"* P10

As mentioned previously, employer 2's organisation had ownership as a core cultural value expected of all staff. She demonstrated the building block approach that was used with new WPS:

"So it gives them more tangible things and bigger projects, I think... It will get to the point when they're probably nine, ten months in, that they are owning massive things and they are the person doing it. It's nothing to do with me at all...be part of my team, but actually they have owned from start to finish." E2

Nevertheless, these building blocks were not just about constructing ownership to larger building blocks, but also to provide context so that the WPS could see how tasks joined together as part of a bigger picture. E3 suggests how they manage this process:

"we think it's very important that as they learn, they have context, so they have a framework on which to build the task, so that no task is just a little silo of its own." E3

E2 concurred suggesting she would often bring the WPS into meetings or onto calls to shadow her regarding *"things that aren't their direct day job, but...things that affect us as a team"* (E2).

Context-setting along with building ownership helped WPS confidence and feeling part of a team. These were key relational foundations and, when "giving" ownership worked well, helped move the individual from owning tasks to owning their job role. P14 suggested her supervisor gave her so much independence that she realised that the work was her responsibility. But this ownership went beyond just opportunities to control, craft and cultivate the job role and included setting her own deadlines *"there's a lot of freedom with how I work and when I do the tasks, as long as I'm meeting deadlines it's completely fine"* P14. Other participants at employer two (P12,15, 19) particularly mentioned the opportunities to choose where to work, which again linked back to feeling trusted by supervisors and the organisation. It wasn't a case of just deciding how to do their job but having ownership for their working life – another building block to job-related PO. Nonetheless building ownership of the job role simultaneously facilitated boundary and threshold-crossing from student to professional (Mele 2021).

Context-setting and feeling that they were given greater opportunities to be part of the team also provided a connection to the organisation. Pierce et al. (2004) suggest that job-related ownership is frequently stronger than organisational PO, which this study would concur with, yet frequently the company were praised by the WPS, when actually it was mainly due to the way in which they were managed by the supervisor / team. Participant 15 provides a typical comment demonstrating how that connection with the organisation is often built by staff members:

> "So I think that it's a, it's a great company to work for, first off. I think that since I've been here, I've been treated incredibly well" P15

Therefore, whilst the process of "giving" ownership is mainly pertinent to the job role, the organisation may benefit by being nested in this positive relationship (which will be discussed further in 4.5.4). Whilst sometimes structural elements such as being able to work from home, were important, actually the relational element of belonging was as crucial.

Bernhard and O'Driscoll (2011) study of family business owners' leadership styles suggested both transformational and transactional forms of leadership had a positive impact on the PO of non-family employees. Whilst leadership was not part of original aim of this study, this research does strongly suggest that for these early career professionals the supervisor role appears to be critical. Whilst there were indications of both transformational "*XXX has been such an inspiration to me, such a role model…..he's had such an impact on my life.*" (P2) and transactional leadership "*my Director was very rulebook focused, everything had to abide by the rules…very strict*" P13, in most instances, the WPS worked with lower-level managers and had a very different experience. Role relationships did however serve both task-related and sociopsychological *needs* "So it's that support whenever I do start to question things. She goes, No, you're an idiot you'll be fine. She's been incredible for me" (P4) to facilitate ownership development and confidence with the supervisor or mentor taking a key role.

Supervisors were conduits ensuring that ownership ran all the way through the WPS work. E1 discussed a conversation with P7 after she had taken responsibility for an office move. Whilst she did a fantastic job with the move, she then left the empty crates in the centre of the floor rather than arranging them to be removed. E1 talked to her about "hinching it" and effectively owning the entire process rather than 90% of this large task.

E2 suggested that supervisors were the "safety net" that would catch the WPS if they fell, but conversely would also give them a great deal of work to push them a little out of their comfort zone. The participants at this organisation did generally concur regarding the perceived organisational support (Zhang 2020) *"They obviously put a lot of effort and a lot of trust in placement students"* (P20), although P12 did have several challenges working at the organisation which are discussed in the shadow side of ownership section.

A coaching style appeared successfully used by supervisors in the development of PO, especially during the newcomer socialisation period, with employer 3 suggesting how he coached WPS to create, control and invest selves in their work. "it was a really good article, but I wanted to get her to start to think critically...And she crafted it completely differently...she's now created something that is so much better ...So things like that... showing them how to work...how to construct things" E3

This was an iterative process in which the supervisor was permissioning ownership, autonomy, self-investment, and control, whilst providing developmental support. However, as an experienced manager, E3 was also aware of his potential for giving too much ownership to individuals and that due to their background, their motivation, or their lack of direction in terms of the task, they might not accept it:

"So my potential for over-delegation which I recognise, doesn't sit well with people who don't like ownership, but it sits very well with people who do." E3

Some WPS did struggle to develop early ownership, as there was limited transference of ownership from their supervisor. As mentioned previously, P18's supervisor worked remotely (pre-pandemic interviews), and she did not meet her in the first 4 months of the placement. The opportunities for transference were confined to emails and the occasional telephone call which left the individual lacking in direction and confidence "it's guite hard not knowing you know, what it is to have an actual Manager and to be managed, basically" (P18). In this instance, the role of *"giving"* ownership moved to another team member and over time the WPS did develop greater confidence and task ownership. Nevertheless, by her own admission she still maintained an intern identity and didn't feel clearly permissioned to take ownership in many situations: "Make decisions for the company, which is quite hard to do sometimes because...as an intern you think do I have the power to do that?" (P18). Whilst it could be suggested that by considering herself as an intern, she had transitioned through liminality away from her student identity, there is an incapacity or hesitation regarding decision-making. Odio (2021) and Garsten (1999) suggest that internships may leave participants feeling powerless because of their transitory role and low status. Whilst this was rarely observed in this study, the lack of supervision seems to have contributed to a "limbo" like

state for much of the placement, leaving P18 questioning how she is seen by others (Mele et al. 2021) rather than feeling a professional who belonged in the organisation.

In some organisations, there was a frequent turnover of managers, which then impacted on the supervisor's ability to "give ownership" and fill the gaps in the WPS knowledge.

P5 worked for in a branch of an international organisation. His supervisor changed on three occasions due to staff turnover, which had an impact on his understanding of the job role and his development of job-related PO:

"What I really needed at that point was a manager to go, "this is what you can do next in terms of the build-up of tasks" you learn and do one task and you expand it...after Christmas, I got a manager who was really good at educating in terms of this is, this is the business and basically learning and teaching the employees and he was really good at that." P5

Whilst the WPS felt confident in some tasks, in areas where he had less selfefficacy, his development slowed, and he became frustrated and then bored. A new supervisor facilitated building job-related knowledge, which resulted in self-efficacy, eventually resulting in indications of both job and organisational PO. He was however left considering what might have been and before the end of his placement was considering a new career direction on graduation.

Finally, the PDAs highlighted how often supervisors were themselves often in their first managerial role and so were learning the managerial ropes themselves. PDA1 suggested that this was now becoming a more frequent cause of miscommunication:

"The companies will let the managers cut their teeth and get their first management experience on a student, um, and that can be really difficult because if the manager doesn't quite know what they're doing in everything as well, the student doesn't quite know what they're doing, then automatically there's this clash" PDA1 Employer 2 concurred, suggesting early in her career, she found it challenging to know how best to manager a WPS: *"I just had no idea how to delegate, wasn't sure what I should be doing versus what was I allowed"* (E2). Nonetheless, at the most basic level, poor communication or inconsistent delegation may leave the WPS struggling to complete tasks correctly. Yet, if ownership is not clearly given and permissioned, it can leave the WPS in limbo, lacking the confidence or expertise to know how to manage the situation. P15 had two different managers and could see the contrast:

### "XXX, there's less of that delegation, there's more of that participation in terms of how do we work best together because I don't feel like she has a clear enough sort of strategy and set out to give that delegation down" P15

In conclusion, this study highlights how supervisors / mentors played a crucial part in the development of job-related ownership in the early career participants of this study. A supervisor who "gives" ownership, provides challenging work alongside coaching and socio-psychological support (trust, helping the development of a professional self) seems important to help facilitate early feelings of ownership. This study highlights elements of social construction in job-related PO and recommends further work to be undertaken to consider how social interaction by supervisors/leaders may impact the development of PO in staff members. In particular, understanding how this process of "giving" ownership may play out at other levels in organisations.

This section has briefly indicated that organisational culture also plays a role in this process, and this will now be discussed in more detail.

#### 4.2.2 Organisational Culture

There has been some research relating to the way in which PO might manifest in individualist and collectivistic cultures (Peng and Pierce 2015; Wang et al. 2018), however, far fewer studies have considered how PO is nested within an organisational context and therefore the role of organisational culture in the development of PO (Dawkins et al. 2017). The findings from this study indicate that organisational culture plays an important and significant role in PO development with supervisors then facilitating this culture to early career professionals.

Within this study, there were some organisations who embedded ownership in their culture, which was communicated to the WPS by the previous incumbent, supervisors, and other team members. WPS 4,5, 10, 11, 12, 14, 15, 19, 20, all gave examples of an ownership culture. P5 suggested that from day one he was aware of his responsibilities and that the organisational culture encouraged individual problem solving, self-management and autonomy, as well as taking responsibility. *"I don't think I had a choice of deciding where I'm going to own this...from day one...it was drummed into me."* P5

Participant 19 looked for organisations that provided work placements with a great deal of responsibility, although was surprised when he was asked to be the European lead on the weekly global strategy call, whilst his manager went on holiday during the second week of his placement. When asking for support, he was told that people generally learnt from experience. *"I'd much rather you jump in at the deep end, learn the hard way, and then going forwards you'll benefit."* (comment from line manager, recounted by P19). Ownership was given and employees were expected to be accountable and so he accepted and cultivated ownership at this early stage, aware that he had a 'safety net'. This was a similar experience to P12, 14, 15 and P20, who worked for Employer 2:

*"I'd like to think we really push them and we really enable them to have ownership because we … it's not sink or swim but it's … we will like throw you in and your safety net is down there but basically go off and make your mistakes"* E2

and also suggested by E3 "encourage them to run with it, but knowing that at all times, I am there to support them" (E3).

As mentioned previously, this specific group of WPS working at E2 were given ownership over their workloads, their time, their calendars, and deadlines. This provided them considerable opportunities to take control, craft and invest themselves in the various building blocks that joined together to form their overarching job role. P12 ran assessment centre days and was given overall responsibility for them: *"I loved the responsibility of it, I loved running the day and I felt in control"* (P12). Control has been posited as a key PO route (Pierce et al. 2001, 2003; Pierce et al. 2004) and will be discussed further in the "taking ownership" section. However, whilst this study concurs, organisational culture and supervisors were also crucial in facilitating opportunities for control, such as the example above and what Liu et al. (2012) describes as a self-managing team climate (in which individuals have autonomy, discretion, and independence). Pierce et al. (2004) research suggested control mediates the work environment structure (technology, autonomy and participative decision making). However, I would contest that it is organisational culture which really embeds PO in an organisation and the work environment structure relates to part of "how" organisations implement this. E2 provides an example below suggesting how ownership is ritually embedded in appraisals within the organisation, not just for WPS, but beyond this to their full-time staff:

"So they are constantly being tasked with noting down when they've taken ownership for things...it's very much you start with those competencies, but they still follow you through and they're still things we talk about with them, and myself, years later." E2

P2, 4, 5, 10, 11, 12, 14, 15, 19 and P20 all worked in organisations where they felt that they were 'thrown in the deep end' and were required to learn by their mistakes. However, most crucially, these were environments where individuals mainly felt supported (P5 less so, due to supervisor turn over). They were granted the autonomy and freedom to manage their work and themselves, but they were also encouraged to ask questions, make mistakes and to learn independently.

"they did really chuck a lot of responsibility on you in the first day and the second day, they sort of it was sort of how did you get on? ...So one aspect, it's chucking you in at the deep end and second aspect is sort of honing your skills nice and early and so straightaway, you're teaching yourself how to do things." P5

Whilst a sink or swim mentality may not suit all individuals, for those with greater self efficacy, it proved the quickest way to develop ownership and a new professional identity.

The WPS in these organisations did develop early ownership feelings, particularly for their job role, which was an attractive, open, accessible and manipulatable target (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004). These individuals also tended to exhibit self-efficacy within their environment and wanted to take control, *"I kind of thrive off autonomy as well"* (P4). Therefore, both organisational support in being "given ownership" and having WPS who developed early self-efficacy helped develop the confidence to accept this form of working. Employer 2 emphasised that they hired the sort of individuals who could manage this form of self-ownership and therefore the selection process may be crucial as a means of assessing suitability to these workplaces.

Nevertheless, as individuals experience a liminal period from the start of their WP, it was still a period of identity–change, bringing uncertainty and selfquestioning. Those organisations with strong ownership cultures helped facilitate this process by providing structure, support and clarity, *"So we're encouraging them to ask for the help"* (E3).

P19 suggested he was immediately treated as an employee who had a voice within the organisation *"we are not viewed as students…We are all viewed as employees and…. have an equal voice."* (P19), which was a sentiment shared by other participants such as P2, P4, P12 and P15. Employee voice and participation in decision-making have been suggested to have a positive relationship with organisational PO (Pierce et al. 2004; Chi and Han 2008; Avey et al. 2012; Li et al. 2012), possibly because it allows individuals to feel that they are the causes of such changes (Pierce and Jussila 2011) and this study concurs. Morewedge (2021) suggests attachments are stronger when linked to social identity and by being "given" ownership; the WPS felt part of the team. Whilst a 'sink or swim' company culture may not suit all individuals, for those with self-efficacy it seemed the quickest way to develop a professional identity and develop job-related PO:

Role-modelling was also encouraged in a number of organisations (P1,3,4,12, 14, 15, 19, 20) with 'lunch and learn' or opportunities to request a 'coffee meet up' with more senior staff. P1 worked for an IT organisation and was initially concerned about his lack of technology know-how. However, he learnt that the

company culture facilitated growth: "it's a real 'learn a tool' culture, not a 'knowit-all' culture" (P1), which gave him the confidence to try out new technologies. Whilst some of these events were as much about the WPS exploring career options and building knowledge of roles and responsibilities, there was frequently a more subtle form of rites of renewal, socialisation or cultural integration, which built a sense of belonging and as well as organisational rolemodelling. For organisations trying to build talent pipelines, this seemed effective on several levels, from building an immediate sense of belonging and strengthening social structures to providing 'future selves' (Markus and Nurius 1986) and career paths. P15 felt highly motivated after a talk from a VP within the organisation. The level of appreciation for someone at that level proved to be hugely gratifying and motivational and built that sense of belonging suggested to be important as part of the integration process (Gennep 2013) and taking it beyond the team to feelings for the organisation:

## *"when you have people like that...it motivates you...you think...l've got something in this company"* (P15).

Such events, including onboarding and one-to-one opportunities for engagement, highlight the biographical nature of an organisation in a comparable way that we have with our possessions. Organisational stories and experiences are shared with interns to transfer the essence of the organisation or department. The WPS then add their own experiences, thus building and developing the biography of an organisation (Kleine et al. 1995; Watkins et al. 2015). E1A talked about an Alumni board that the WPS built to demonstrate that history of their team, emphasising their individual and collective identities and how this can also develop a sense of belonging:

"they set up their own little Alumni board. They gave themselves all nicknames...XXX was Kindness Guru and XXX... oh, he was Green wasn't it? He was Sustainability Guru...So it was like they were building their own little community really". E1A

Whilst this is also simultaneously a form of "taking" ownership, it does demonstrate the importance of that sense of belonging for some individuals and

that strong desire to feel part of something which will be discussed further in the "taking ownership section." Moreover, by ascribing titles to each other, they are upholding each other's identities (Holmes 2015) and anchoring identities within the team or department.

Working for an organisation with a strong ownership culture and being "given" ownership did also seem to assist attachments beyond the job role to other organisational targets, such as the team and the organisation itself. Those working at Employer 2 (P12,14,15 and 19) all suggested returning to the organisation after graduation

Nonetheless, whilst some of the organisations did display a culture of ownership, at times there were situations that occurred which inevitably impacted on the WPS. As mentioned previously, P5 had several different supervisors across the year and periods of no supervision. Whilst initially he felt able to take ownership because he was supported, when this support disappeared, his feelings of ownership at times reduced, because he did not have the knowledge or skills required to move to the next level "*What I really needed… was a manager "this is what you can do next in terms of the build-up of tasks" P5* 

P10 worked in various retail branches and had a particularly poor experience in one branch during a branch rotation. Feeling unsupported working with a team who she felt were setting her impossible challenges, she reflected, *"That was horrible. They were so hard on me" (P10). It* was only when she moved locations that she felt in control and perceived organisational support (Pierce et al. 2001; Zhang 2020). P2 suggested within his SME that whilst he perceived ownership across the organisation, some teams demonstrated more ownership than others, with the Project Management team protecting their projects as if it was *"their baby"*.

E1a did not believe that her organisation had an ownership culture: *"if it was cultural…I think you'd see it in all staff and I don't…think I see it in all staff." (E1a),* yet paradoxically believed the WPS felt differently, because it was "*their XXX" (E1A).* As I have already referred to the student-initiated alumni board, it does feel as though ownership was felt by the WPS involved in the creation of this board. From the participant's interview (P5), however, I would suggest that

ownership was embedded in the department and so the behaviour in the team and via the outgoing WPS meant that there was a microculture of ownership, "we did the handover through the students... the previous student embedded it in the new student" (E1). It seems plausible that if ownership is not embedded throughout an organisation, there may be pockets of ownership behaviour coming through departments and teams. However, this area needs to be further researched to consider organisational culture and PO in more detail.

P13 suggested ownership felt embedded in some departments who had autonomy within the team, but not within his own team., although this was mainly due to the relationship with his supervisor, *"I think she was very stubborn and umm didn't like other people's ideas"* (P13), which will be discussed further in the coming section. P6 did not demonstrate many examples of workplace ownership within his department. However, his organisation ran "Kaizen" events, which is a form of continuous improvement. People could come up with an idea and individuals volunteer to be part of a project team to develop the idea. This form of ownership is neither solely job- nor organisation-related PO, but does demonstrate a form of ownership, although the boundaries of the target are not completely clear.

In another example, P8 worked for an organisation where there had been a quick organisational growth and organisational directors were keen to establish new values. She noticed a change in culture during the duration of the placement, *"it felt like a completely different company at the end to the company that I started with"* (P8). The change of culture also moved from one where interns were not valued "she's the intern, she can do it"... that kind of stigma attached to it" (P8), to one where the CEO expressed an interest in intern programmes. This, in her eyes, resulted in being given ownership of tasks and projects in a way that wasn't possible at the start of the work placement. Luckily this WPS grew alongside the cultural changes herself and ultimately felt valued by the organisation.

Overall, the participants of this study indicate organisational culture to be a key element in facilitating ownership in individuals at the start of their career:

*"I don't think as a staff member can actually fully take ownership, unless it is given as well by management, you know, because it's the* 

## kind of thing in a working environment with is a pretty much two-way street." PDA2

In particular when WPS were given responsibility for the entirety of their job role, including where and how they work, it allowed those individuals looking for responsibility to blossom. They generally developed ownership feelings for the whole role, rather than just tasks and with early PO development more common. However, this research stresses the importance of support, which links to the notion of safety (Zhang 2020). If organisations do not consider fully the mechanisms to support these early career professionals, there is a chance that they stop learning and developing or lose their confidence. Many of these individuals still needed a support mechanism in place and/or a supported "sink or swim" mentality, where mistakes were encouraged.

To conclude, those organisations who embed job-related ownership within their culture provide their WPS and other staff opportunities, which not only benefit the job role, but also the company. Nevertheless, further research needs to be undertaken to consider this in other organisational settings with other groups of employers. For researchers, this could however be a fruitful area of further study.

#### 4.3 Trust

The third key element in this study, which facilitated "giving" ownership, was for individuals to feel trusted when undertaking their job role. As new employees working in their first professional job role, it was a quite different experience to prior work experience in part-time job roles. P2 explained what he perceived to be the difference:

"There's two different types of way of working, one is basically, you get told what to do, you clock in, clock out and that's it. And there's another model... I'm in now is where... what it is, you're given an aim to achieve, and you have to go about it" P2

This change in working practice is exhilarating, but can be nerve-inducing and feeling trusted by those giving tasks was crucial, due to the initial fear factor felt

by many participants, "I've been...left to run meetings, something that I was really scared about doing at the beginning" (P16). Trust relies on an individual being vulnerable to another without fear of their intentions and behaviours (Rousseau et al '98, Mayer et al '95). These early career starters were already vulnerable because of their lack of workplace experience and understanding of how "work works", but confidence and feelings of being trusted built over time "So yeah, the more trust and the longer I have been there, the more I understand what's going on" (P16)

Within those organisations where an ownership culture was embedded (4,5, 10, 11, 12, 14, 15, 19, 20), trusting staff to complete their work was generally the norm. The "building block" approach employed by many organisations allowed supervisors to demonstrate trust in their new charges and to help their new employees understand the business environment with which they were now operating. Initially, small tasks were completed before the WPS was given increased responsibility and was trusted to complete the work in the way that best suited them, *"that post three months…I trust they can do things without needing me to check everything" (E2).* WPS felt as if they were tested by their supervisors and if they took responsibility for a small task, then this was a building block to being permissioned larger, more complex tasks:

"slowly she started showing me signs that she wanted to push me to the point that I feel comfortable in just going to her and saying, I would like to do this." P11

Brown et al. (2014b) recognise the interpersonal nature of trust and it was also clear in this study that the supervisors who signalled trust were more likely to be rewarded with early job-related PO from their WPS. P14 spoke about how she got to a point where she was autonomous:

"my managers just kind of expect me to be getting on with my job – I don't have to tell them what I'm doing every time I do a task... they trust me to do it." P14 For WPS to build confidence and trust to accept ownership, the PDAs suggested that the WPS needed to be within a structured, supportive environment where they were given responsibility, but also provided with clear ownership guidelines which also included feedback, *"I think ownership also comes from good feedback from managers as well"* (PDA1). This links to the work of Zhang (2020) whose meta-analysis of PO research suggested safety was a key PO antecedent. Safety consists of four components (organisational justice, trust, perceived organisational support and relational closeness) and whilst trust was most strongly identified in this research, both organisational support and relational closeness were observed as part of the "giving ownership" mentality.

Trust did allow the WPS to feel safe to complete job role tasks, because it demonstrated their supervisor's confidence in their abilities, which spurred them on to undertake increasingly more complex tasks and own their job role:

"because I don't think my manager would have trusted me if she didn't think that I could have handled that.....So the whole process, she was there with me, but it wasn't handholding...that was definitely a step for me to then be able to do bigger things later because I felt like, okay, she trusted me with this" P11

Olckers and Enslin (2016a) study of participants in professional services in South Africa found a positive correlation between PO and workplace trust and in particular trust in the organisation. The participants from this study also indicated the role of organisational culture influencing the level of trust given to WPS. Employer 3 felt that trust was key to make WPS feel that they were valued:

"just let them get on with it, because if we've chosen them right..they're going to be listened to, they see that the work is valued, they see that they can contribute, it's pretty much my work here is done, just let them get on with it and gently guide." E3

Nevertheless, whilst an ownership culture tended to embed trust, supervisory trust often only came with experience, as mentioned by Employer 2, who suggested that it took a few years of supervision before she had the confidence to trust those working for her. This view was also expressed by the PDAs who, as previously mentioned, suggested first-time supervisors were given a WPS as

the first person they would manage. PDA 1 suggested a lack of trust manifesting in micro-managing was noticeably common in "millennial" managers, although this may also be due to the new supervisor being desperate to succeed. PDA1 suggested the number of managers who struggled with delegating the current cohort of WPS "probably about six or seven, and they've been more micromanagers" (PDA1).

Certainly, those WPS with experienced supervisors (P1,2, 4, 7, 9, 10, 11, 14, 15 and 19) felt more noticeably trusted by their supervisors *"She was very, very liberal, she allowed me to just do everything"* (E11), although this was not true in all instances (P3 and P13). E13 struggled with his supervisor because whilst ownership was often given, he perceived that there were always boundaries around this ownership resulting in little trust or interpersonal closeness. The "giving" of ownership came with caveats and was more akin to lending, causing doubts in the WPS mind, limiting their self-efficacy and ultimately affecting their level of ownership:

# *"It was always a bit difficult...my Manager had already preconceived ideas about what needed to be done and when but wouldn't tell me what it was and when it was to be done for."* P13

There was a strong perception of unfairness related by P13 suggesting that a lack of interpersonal and procedural justice can inhibit opportunities to develop PO (Chi and Han 2008). Trust has an element of the past in that our experiences shape our future trusting behaviour (Rousseau et al. 1998) and P13's experience is almost the opposite of P11. Where P11's confidence grew, because she felt trusted and safe, P13's confidence eroded, "*Umm, I was confident at the beginning, but then the confidence sort of dipped*" (P13) and the relationship suffered, never to recover. Ownership was therefore sporadic, occasionally given, but more frequently borrowed with hidden caveats "*What I struggled with is that you really had to be careful asking a question of how to do something or umm putting forward an idea*" (*P13*). The essence of the "giving ownership" process had become contaminated (Rozin et al. 1989; Argo et al. 2006; Kramer and Block 2014).

P3's story has similarities in that both WPS experienced a lack of interpersonal trust with their supervisor and in both instances, others in the

organisation acknowledged challenging supervisory behaviour, "My line manager was particularly, (pause) he, loads of people say... he can be very strange" (P3) and

*"previous intern said...might come back five or ten times umm but wouldn't tell me how to do it right"* (P13). P3 showed a great deal of ownership in a previous part time role and talks about the differences between that role and her work placement role in relation to PO:

"I've always been able to go that extra mile, see something that needs to be done...to be changed and just do it. And they've encouraged me and empowered me...I don't think I've always had that here. It's it's been a lot of hand holding..I get that ownership there. If I want to do something, they trust me to go and do it" (P3)

The key differences relate to that of interpersonal safety and trust, but because of this, there are no opportunities to try out the role to see if it fits resulting in a limbo like state.

Brown et al. (2014a) suggested whilst there are many positives to forming high trust environments, there may be occasions when individuals engage in territorial behaviour that impact on team dynamics. Whilst there was little evidence of territorial behaviour found in this study, P10 did suggest that because of previous experiences of group-work at university, she was used to claiming and withholding knowledge, *"it's not like you're in first year of uni where people don't care"* (P10). She initially continued this behaviour in the workplace before realising that she needed to trust other's capabilities.

The biographies of our previous ownership experiences have implications for future psychological ownership, and therefore those who have been less inclined to trust in the past may need role modelling and support on how to build trust in a different environment. In addition, organisations need to consider who best to line-manage early career professionals, as their trust behaviour can have a significant impact on an early career professional's opportunity to develop PO.

To conclude, this study builds our understanding of the role trust may play in PO development and in particular its role in "giving" ownership alongside the influence of the supervisor and organisational culture. As these study

participants are early career professionals, trust may be more significant for PO development than for other, more experienced employees, although given the outcomes of previous studies (Brown et al. 2014a; Olckers and Zyl 2016; Knapp et al. 2019; Zhang 2020) trust and organisational safety may always be crucial in the giving ownership relationship.

"Giving" ownership to WPS relates to the clear permissioning of ownership in facilitating the development of job-related PO. This study's results suggest the importance of three key elements that expedite this process alongside the relational element of PO, which has often been missing from many previous PO studies. The importance of these elements seems crucial to the development of PO in these early career professionals. However, it could also be contended that these elements can play an important part in the development of employees PO at any level. Whilst the permissioning element may be of less importance to experienced staff, the active relational process of "giving" ownership to employees via the organisational culture and through supervisors and leaders could be an important signalling tool aiding effective communication. Nevertheless, there also needs to be interpersonal justice and organisational safety (Knapp et al. 2019; Zhang 2020) mainly in the form of trust, to quickly develop an individual's confidence to build job-related ownership.

The themes in this study are interconnected and so the picture of PO development will continue to be shaped in the coming sections of this chapter, displaying a complex tapestry of feelings relating to PO.

#### 4.4 Theme Two: "Taking Job-related Ownership"

Theme two considers the second element of this relational ownership process, that of "taking ownership", which related to PO of the job role. How this may then influence organisational PO will be discussed in section 4.5.4. The analysis of data from this study indicates two forms of ownership being taken. One form is an acceptance of being "given" ownership by a supervisor, mentor, or team member and how that facilitates future opportunities to take ownership. The second form is when ownership has not been offered, but the WPS decides to take ownership, nevertheless. Both forms are active, relational and rely on suggested opportunities to fulfil one or more motive, access to PO routes and personal factors, and PO target attributes.

Pierce et al. (2003) suggested that whilst individuals may quickly recognise a target as "theirs", there is an interactive, lengthy process to develop full cognitive and affective feelings of attachment. Although no guidance was given on their definition of "lengthy", they suggest that PO emerges at the confluence of at least one of the posited routes (control, intimate knowledge and investing self) with target attributes (visibility, attractiveness, accessibility, and malleability) and individual factors, such as traits and values (Pierce et al 2003; Pierce and Brown, 2020). However, beyond these assertions, no studies have developed, concurred, or disagreed with this original notion, and therefore this research provides a more definitive starting point which future studies can build on. The following sections will consider posited PO routes (Pierce et al. 2001, 2003; Zhang 2020; Morewedge 2021) which expediated "taking" ownership by this group of early career professionals. This section will then discuss the two forms of "taking" ownership in more detail.

#### 4.5 Control

This study's results concur with previous studies regarding the importance of the control route in contributing to the development of job-related PO (Pierce et

al. 2001, 2003; Liu et al. 2012). Nevertheless, there were some notable differences which allude to the complexity of PO in different contexts. In this study, ownership permission allowed WPS to take control, which as new employees they were not all used to doing in the workplace. P12 provides a typical response to the differences in control from a work placement to a previous part time job:

"Every week...no one was saying do this, do that...Whereas at XXX... okay you're on this section, you're doing this tonight and you'll do that. I was directed more" (P12)

Permissioning and trust were very important in WPS feeling able to take control of tasks which organisations and supervisors managed in different ways. In organisation 2, P12,14,15 and 19 were expected to take control via the supported "sink or swim" method. Li et al. (2012) discuss control as a building block to PO, but in this study, the array of responses suggests that whilst control is important, other routes, such as opportunities to invest self, safety and self-congruity, as well as confidence, also play a part.

P15 liked opportunities to take control and get involved, so from the moment that control was given, he took control and developed his job role considerably:

"I think I've been able to give it my own, like, twang, if you like... to have a bit more freedom...some people like to report into quite a lot saying, is it okay to do this... I probably should do it more than I do, but I just quite like to go about stuff" P15

This was a participant who actively wanted to control and invest himself to give the role his own stamp. Safety was less crucial to him and when ownership was permissioned, he took it. His line manager (E2) was a great admirer, suggesting she had seen his potential during the recruitment process, and he didn't disappoint, *"Whereas he's only five months in now and he is outstanding"* (E2). In contrast, P14 who worked in the same team, initially borrowed ownership, having some hesitancy around permissioning, which did seem to relate to confidence: "I would be telling umm my...Manager...when I was working from home, or umm updating them when I'd completed a task and I kind of realised that all they said was, oh that's great. They kind of didn't really need to know" (P14)

A number of other WPS such as P6,16, 17 and 18 were similarly lacking in selfefficacy initially, but the differences in strength of ownership between P14 and P15 felt significant. P14 suggested that by the placement mid- point, she had a great deal of control and enjoyed the level of autonomy and feelings of trust:

"I would say I very much do have control...I don't have to tell them what I'm doing every time I do a task or when I've finished it, they trust me to do it...there's a lot of freedom with how I work and when I do the tasks...where I work as well, I didn't have to be here today, technically I could work from home." P14

Ultimately, P15 seemed to "take" ownership of the job role, pushing the boundaries and 'job crafting' (a topic to be discussed in S4.4), as far as he was able to: "anything to do with recruitment in XXX... I feel responsible.. (thumps the table)" (P15), Conversely, it seemed as if P14 was happy to control within her job-role parameters, but, whilst she developed job-related PO, there didn't seem to be a strength of ownership feelings nor the desire to craft that was perceived in other WPS. Pierce and Jussila (2011) suggest that boundary conditions such as individualism/collectivism, job involvement and organisational self-esteem demonstrate the differences between individuals, and this study partly concurs that job involvement may be important. Nevertheless, it could also be surmised that P15's self-worth is linked to work. Therefore, it feels that many factors play a part in these individual differences and it is the opinion of the author that it is this complexity that needs to be better understood by theorists.

Whilst most individuals were permissioned some form of control, the differences between some contexts were stark. P10 worked for an organisation that stated in their recruitment process that their placement would be tough. In this instance, permission wasn't granted in an obvious way, rather there seemed to

be a deliberate policy, which required WPS to seize opportunities and take control:

"we would be told what to do...if you want to change something, or you want to say... would you mind, if I don't watch you today, I'd like to go and do it myself. You can but you almost that, you have to want it. They won't just really give you all the opportunities because they want to see if you're going to get them yourselves. P10

This organisational culture was different to most, but for participants such as P10, they understood expectations in advance and wanted that sort of challenge even if at the start of the placement this seemed particularly demanding (this was the WPS, who went home and cried for the first week of her placement). At the other end of the spectrum was P13 (discussed in the trust section), who had some control, but did not feel trusted. In this instance, the "giving" of ownership proved to come with caveats and was more akin to lending, causing doubts in the WPS mind, limiting their self-efficacy and ultimately affecting their level of ownership. Taking ownership was challenging because:

"I think she was very stubborn and umm didn't like other people's ideas...f something was suggested, she'd say, no, not doing that. So yeah, it was difficult" P13

This indicated a "limbo"-like situation, second-guessing expectations and unable to take control and invest self in his work. For those individuals at the start of their career, the ownership game in the workplace can be akin to learning ownership rules as a toddler. If a child is given a toy by an older child, but then has it snatched back from them, they become wary and confused. If normative ownership requirements are not clear in a working situation, a certain element of wariness may develop, making it more challenging for some individuals to accept or take control in further instances, especially if they lack self-efficacy.

The PO route of control has been mainly considered from its technical elements, such as work environment structure, which includes technology and job design. What this study suggests is the social element between supervisors and team members who permission and/or show trust to the WPS in allowing them control of tasks and their job role. This perspective has not been considered in detail in previous PO research and whilst it may seem more pertinent to some employee groups, especially those at the start of their career, any situations where work demarcations are not clear, or where "giving and trusting" are not aligned with PO routes, individuals seem less likely to "take" ownership.

#### 4.6 Investing Self in Target

Investing self is the second PO route allowing an individual to cultivate, shape and invest psychic energy into the target, until it feels as if it is part of them (Pierce et al. 2001, 2003).

PDA1 discussed how three months into the start of the placement, when she undertook her first company visits, some students were already investing themselves into their tasks:

"the good ones, the ones that are already sort of like, identifying... "this process seems to be taking a lot longer, what if we change that we could do that a lot quicker"" PDA1

Once again, the indications were that "giving ownership" facilitated the process by giving permission to WPS that they could invest themselves into job-related targets such as tasks, projects, or the entire role. P11 talks about this permissioning and the freedom to invest herself in her job role:

*"I think it all goes to the management, so it's all about … how much freedom you're given to carry on your own projects. Not just your own projects but the way that you carry on a specific task and everything." P11* 

One way that some organisations helped facilitate this self-investment was by providing the WPS with a personal project, which was often open-ended in terms of scope. P4 discussed how that freedom can then build greater self-investment: *"the social impact stuff because I really invested myself into it and I really cared for it"* (P4)

This refers to behavioural commitment and the importance of ownership by choice (Brehm 1956; Harmon-Jones and Mills 1999; Ye and Gawronski 2016a). If workers and especially early career professionals can choose additional elements to work on, such as a project, this can help construct self-investment in the job role as it builds self-congruity (Morewedge 2021). Nevertheless, ownership by choice is also about control and it feels that it is the entwining of control and self-investment that is important in the "taking" of ownership in this group of participants:

"XXX basically took the PMB report off me and that was it. He said, is it alright if I just smarten it up a bit. I said, do what you want and within the first month I was like, wow where does this come from." E1A

One again however, it should be made clear that self-investment and other PO routes are facilitated by individuals, such as the supervisor or via an organisational culture, which provides these opportunities that allow individuals to feel trusted and safe. Ownership is being constructed between individuals giving and taking ownership, rather than a PO route magically opening by itself.

#### 4.7 Self-Congruity

Self-congruity links to the mere and endowment effects (Heider 1958; Beggan 1992) whereby individuals look favourably on objects they own, due to the

psychological attachment made. Morewedge (2021) suggests that this attachment is stronger when the object links to an individual's social identity. As a recently suggested PO route, there are no comparison studies available. Nevertheless, there were examples of self-congruity which suggest that is should be investigated further.

As mentioned in the "giving" ownership section, results from this study indicate the self-congruity route may start for some during the recruitment process, when the WPS build that psychological attachment to the organisation, by firstly making a choice to apply and by then receiving positive feedback from the organisation. This may come from being taken further through the selection process, but also via shared experiences at the interview and assessment centre. P7 demonstrates how self-congruity was developed during this period and how she started to look more favourably towards an organisation that showed interest in her application:

"I kind of thought...that it would be as a backup...he said if I wanted to come to the office and look around...and when I went in there...I was actually more interested in this role than I was in the XXX role and XXX role. Because he gave me a much more personalised experience like, whereas like XXX...obviously it's all via email and stuff but it is very much all on their terms...I felt a little bit more like kind of valued." P7

Most students were relieved to receive an offer, even if it wasn't from their initial preferred organisation. In the same way that (Sartre 1956) suggests that buying objects can be an act of creation, because energy has been invested, the psychic energy invested in an organisation's selection process, alongside the act of saying yes, may facilitate the early feelings of self-congruency, especially if the role or the organisation links with the student's self-worth (Ferraro et al. 2011; Ye and Gawronski 2016a). Research into PO routes has not focussed on this early stage, but further research into the selection process may provide clarity on the starting point of PO routes.

Once in a job role, the Mere effect (Beggan 1992) may influence how individuals consider their role and if the WPS believe it is congruent with their identity, they may form a positive opinion of it. P11 initially felt more interest in another intern's role at the start of her placement, but over time adjusted her opinion:

"And then for the role I think, initially when I first saw it, I really liked my colleague's role because she was doing UK specifics, so she got to do a lot of events and actually go to the events. Now having finished, my opinion of that has changed." P11

Investing in a role shapes our perception and the combination of selfinvestment and controlling tasks is likely to strengthen such feelings. But what if an individual is less interested in a role? P4 struggled to settle initially into the organisation, but involving himself in a community project, and speaking to his supervisor regarding his interests, *"my eyes kind of wavering over to, to, to being customer facing"* (P4) facilitated self-congruity, which then helped activate the sense of place / home and self-efficacy motives (Pierce et al. 2001, 2003; Pierce et al. 2004). Nonetheless, not all WPS experienced the self-congruity route and some, such as P3, whilst developing ownership feelings for her project, were unable to use them as a bridge to job-related or organisational PO.

P6 found his role via a friend and so did not have to invest as much effort into the recruitment process as others, and in fact was *"a bit sceptical but I thought it would be good to see what it would actually be like in an actual job"* (P6). A combination of a busy supervisor, a slight change to the job role and being left to fend for himself resulted in little self-congruence with the role or the company, nor the opportunity to invest himself. Only a project towards the end of the placement gave him the feeling that he had contributed anything of value:

"some of the jobs that I was doing...it seems like a waste of time and I, I don't think at times people took it seriously. But then other times when I like with that productivity data, umm you know, people were actually towards the end...were taking more, let's say, seriously and it was good to kind of get that side across." (P6)

By his own admission, this project was his main example of ownership and ultimately, this experience came too late to build sufficient attachment, and he seemed unsatisfied with the placement experience.

Confidence and self-efficacy also seem to play a role in self-congruity. If individuals perceived themselves as efficacious, they were more likely to be able to deal with challenges, such as a supported 'sink or swim' scenario, because they would allow themselves to deal with such situations as P2 did. Getting through hurdles built self-confidence, but also helped develop self-congruity:

"Umm a big thing for me as well is confidence...As I said I have never been in this scenario before and to be not chucked into the deep end, but after a couple of months I was in a bit uncomfortable situation, but it's all for the better." (P2)

#### 4.8 Intimate Knowledge of the Target

Intimate knowledge is the third route to PO suggested by Pierce and colleagues (2001, 2003; 2004) reflecting coming to know a target intimately. Given the limited time these WPS were in the workplace, combined with the implication that a longer time was required to undertake this route (Pierce et al. 2003), prior to the study, this felt like a less accessible route. Nevertheless, when combined with other routes (including self-congruity) and having been permissioned and trusted to take ownership, this route was utilised by some WPS to anchor themselves into the job role, although to a lesser degree than other routes.

P17 became the company's 'go-to' person for certain processes, due to his high level of Excel skills, which allowed him to modify some processes:

*"in a sense, responsibility fell on me in the workplace and I sort of owned that section of what we do…and that was my forte (laughs) So, yeah, after those two months, that was the point where I sort of said, I own this* 

#### process, I do this and that and that...So yeah, it feels like I've become part of the office's history almost." P17

Becoming the expert in a certain area allowed him intimate knowledge of his job role, but also helped his feelings of belonging towards the organisation. Brown et al. (2014b) suggest that intimate knowledge can help satisfy the "sense of home" motive and whilst this wasn't widely seen in this study, this WPS equated ownership with belonging and so it was obviously important to him. P2, 4, 8, 11, 12 and 15 all demonstrated intimate knowledge of the job role and generally this resulted in them being seen as the "expert" in a particular area. P2 for example, was bought into a small organisation as a project manager, but took over the company's social media. Whilst initially he had not perceived this area as one with particular skills, he was far more confident with it than other employees and so "took" ownership of this area. He found he really enjoyed this part of the role and was able to invest himself into this area to a greater degree than his project management role, where there were other project managers with far more expertise.

Nonetheless, there were perceived downsides to individuals whose level of "intimate knowledge" was conceivably greater than other staff members, including their supervisor. P12 joined a department that was going through a number of changes, in a role managing early talent recruitment. Joining the department a couple of months before her new supervisor, she demonstrated control, competence and intimate knowledge of the role, to the point that the client came solely to her, rather than her supervisor:

"she didn't even see herself as my manager for a couple of months. I kept saying to her, you know, you are my manager, you need to tell me what to do. And at the end, I was actually telling her what to do so the responsibility between us, it kind of actually fell more on my shoulders because the client just kept coming to me (emphasis) with issues. So I was actually taking more of her stuff than she was of mine." P12

For WPS who have little experience of "how work works" or managing upwards, having greater knowledge than their manager can prove to be a challenging experience. This instance will be discussed further in the pleasure and pain section. Pierce et al (2003) suggest that having responsibility for a particular task or job role helps develop feelings of ownership, because individuals are expected to take control and invest themselves in targets. This study concurs and suggests that it is the degree of responsibility, control, opportunity for self-investment and self-congruity which may result in strong ownership feelings, rather than the time period that has been suggested to relate to intimate knowledge of a job role. For example, P15's role included initially working on UK-based events with one very competent supervisor (his words) resulted in events being well established and running seamlessly within the organisation. A few months further into the placement there was an opportunity to work on a similar type of product in EMEA with a less competent manager (again his words), which then allowed him to feel more involved in the target. This opportunity for agency, control, and the investment of self resulted in the rapid development of job-related PO for the EMEA role, whilst feelings of PO never strongly manifested for the UK part of the role:

"I guess it's the kind of how much responsibility you feel like you've put into it. So, if something goes wrong in the UK, ultimately, I think maybe XXX could have done something different...But then when an event goes wrong that is me because that is my job and it's in my job description and I should be doing that right." P15

Consequently, the suggested notion that the longer the time period, the more chance of developing PO (Pierce et al. 2003) negates the importance of attractive, accessible attributes alongside strong control, self-congruity and self-investment routes. In this instance, as well as PO routes such as control and self-investment being available, the fact that the other target was also extremely accessible and perceived as attractive to cultivate, resulted in opportunities to job craft and to entwine self and tasks. That the UK part of the role did not allow him the opportunity to craft and invest self, nor fully control because of his competent supervisor (although this could also be partly contagion or principal possessor related) (Friedman and Neary 2008; Watkins et al. 2015), resulted in him developing less job-related PO for this part of the job role, even though he had a longer time period to develop intimate knowledge: *"I guess it's the kind of, how much responsibility you feel like you've put into it"* (P15).

Therefore, it could be surmised that intimate knowledge of a target may not be as related to a time period (Pierce et al. 2003), but rather relates to an assortment of routes, attributes, values and organisational factors joining together to facilitate the 'perfect mix'. That mix is likely to differ for everyone, suggesting there is not a perfect PO recipe, but rather a collection of ingredients which can be combined in different ways. This makes PO more challenging to pinpoint as PO surveys assess the strength of PO via motives (Avey et al. 2009) or statements, such as "this is MY organisation" (Van Dyne and Pierce 2004) neither of which may be sufficient to ascertain how PO was developed, nor identify the relational factors which encouraged PO development. Further research regarding PO measurements will therefore be beneficial.

#### 4.9 Psychological Safety

Safety is relatively recently posited PO route, which (Zhang 2020) suggests consists of four key elements: trust, organisational justice regarding perceptions of fairness, relational closeness, and perceived organisational support(Zhang 2020). Trust has already been considered in more detail as part of the "giving ownership" section. Nonetheless, there are elements of "taking" ownership which relate to safety.

Being "given" and then "taking ownership has interpersonal connotations bearing similarities to child development and possessions literature, in which the transfer of ownership and being "given ownership" are perceived as social actions (Friedman and Neary 2008; Kim and Kalish 2009; Blake and Harris 2011; Kalish and Anderson 2011; Tatone et al. 2015). Inevitably within social actions trust, fairness, relational closeness, and perceived support will play a part.

Zahid et al. (2019) suggest that organisational justice can influence the development of PO and there were some examples in this study. WPS such as P13 expressed a lack of organisational justice from his manager: *"She formed ideas very quickly and never changed her mind. So that was very difficult."* (P13). He only seemed to develop task-related ownership when he was working
on an individual project, that he had complete control of. Nonetheless, whilst job-related ownership seemed to never be fully formed, because of his supervisor and a lack of organisational justice, trust, and relational closeness, he felt slightly differently about the organisation as a PO target: *"A bit. But not as much as I would like to"* (P13) and still suggested that he felt he had a future in the organisation: *"XXX would be a really good company to work for, not on that team, but....a really good place to go back to"* (P13).

Whilst this was a slightly different case to other participants who also talked about organisational justice, interestingly they often still perceived the organisation positively, even if the perceived injustice related to organisational controls.

P10 was frustrated with the appraisal system, which she felt didn't consider perceived tough marking from some mentors:

"I remember crying twice in my grading...this isn't fair. I'm trying so hard and my store ops director is going to look at this sheet of paper having not seen me do any of this and go, Oh, she's got a three, but XXX got a five when we're being graded by two completely different people with two completely different specifications" (P10)

Nonetheless this perceived injustice did not affect her strong ownership feelings for the organisation: *"I definitely think personally, I feel ownership over it."* (P10), nor her strong desire to return to the organisation. P12, who initially had a challenging job undertaking what had been two positions in the previous year now rolled into one, also blamed her team rather than the organisation for this perceived organisational injustice *"So I was getting slightly annoyed with the team more than the company"* (P12) even though she was aware that they were also overworked. These types of examples were also seen with P3 and 7, seem to relate to distributive and procedural forms of organisational justice (Colquitt and Chertkoff 2002) However, the WPS seem to process them as more interpersonal forms of justice ('shooting the messenger'). This may be due to their position in the organisation, where they may feel remote from a more abstract target such as an organisation or it may be that other distributive elements, such as salary and graduate job opportunities, are more

important at this point and so outweigh any perceived organisational negativities. It does however demonstrate the relational aspect of PO and safety, and so further investigation of this area for early career professionals may be warranted.

Other elements concerning safety relating to perceived organisational support or relational closeness tended to be positive. P17's example is common of those whose desire for a sense of belonging was strong: *"talk to them....getting issues solved....new things set up....it's just a feeling of belonging between everyone"*(P17)

Feelings of safety were important for this group of participants and as a route, it was often aligned to other PO routes and motives being fulfilled. If neglected, it may act as a blocker to PO, whereby routes such as control are inaccessible. As this is a newly-suggested PO route (Zhang 2020) more research is required, especially at different levels of an organisation. It may be that safety is partly context-driven, with those at the start of their career having slightly different experiences, due to their lack of experience in the working world. However, a lack of trust, a perceived lack of justice or organisational support can occur at all levels. Research relating to safety's relationship with other routes would also be beneficial.

### 4.10 Job Crafting

Pierce et al. (2009b); Peng and Pierce (2015) highlighted the top-down process of job design facilitating job-related PO, but in this study job crafting was indicated by participants to be a more powerful influence on ownership, perhaps because it provides individuals with greater control and opportunities for self-congruity. For P19, missions gave him the opportunity to re-mould the task boundaries within his job role to fit his perceived skills, knowledge, and areas of interest (Wrzesniewski and Dutton 2001). Expansion-orientated job crafting (Naeem et al. 2021) was positively encouraged in this organisation, with the WPS final presentation requiring individuals to demonstrate how they had added value to the organisation, "do a 10-minute presentation to the Commercial Officer…how we impacted the business" (P19)

As mentioned previously, some individuals, such as P4, took their job role in a quite different direction to the one initially specified, job crafting both the task and relational boundaries (Wrzesniewski and Dutton 2001) to suit his interests and skills:

"Well, I kind of thrive off autonomy as well because I also don't like people telling me what to do so I like... I like saying you know; this is what we're looking to do...at first it was daunting... I even got to the point where I just messaged my manager and I'd say I've got really good idea...I want to make something like an impact monitor." P4

Whilst he was responsible for taking ownership boundaries in a different direction via job crafting, he was aided by his supervisor who allowed him a great deal of autonomy and control. Therefore, whilst he had the self-efficacy and confidence to take ownership, she also permissioned job crafting, as well as ownership. P4 did suggest that taking ownership and owning a process from start to finish could be daunting and cause anxiety at times, but the opportunities to invest self, take control, contribute to the business, and feel valued outweighed the negatives in this instance.

Of the suggested forms of job crafting, task crafting was most frequently observed, with all WPS engaging in this form because of a desire to control. Those who suggested the fewest examples of job crafting were individuals such as P3,5, 6, 13,16 and 18. They all provided some job crafting examples, but either the opportunities were less frequently available, such as P6, who asked around other departments to find work at times, "So that meant *I got to do some tasks, which I probably would not have got to do.*" (P6), or P18, who felt that the company dictated the way individuals were required to work: *"Umm, not really. We have a certain way of doing things. I've changed a lot of it to make it easier for me*" (P18). Job crafting in this instance played less of an identity-shaping role (Wrzesniewski and Dutton 2001) with fewer opportunities for self-congruity.

Some participants, such as P2, 7, 9, 14, 17 and 20, either had scope to engage in more task crafting or, in cases such as P14 and P20, were able to choose how they worked. This was the case with P17, who because of his strong Excel skills, simplified processes with macros, thus demonstrating mastery: *"I write up a load of macros and stuff that automate lots of processes."* (P17). These were individuals who would not naturally push the scope of their roles without permission, but once permissioned were confident to broaden their job role, *" so I've kind of taken charge of that side of things"* (P14)

Examples of cognitive crafting, whereby an individual alters their job role perspective (Wrzesniewski and Dutton 2001) in a form of meaning-making behaviour, did occur at times. PDA2 talked about a WPS who was struggling with their job role in a printer. The PDA suggested the WPS look at the print room as his own business and show pride in their work, as that would result in further business:

"And he saw from a totally different light as to why he was actually there. And I think it helped him a lot. In the end, at the end the meeting he said, you really changed my outlook on business and the way I look at doing business" PDA2

Others, such as P8, talked about maturing over the year and how their understanding of "how work works" altered their job role perspective. Whilst this may only relate to early career professionals, it did facilitate the taking ownership process in this WPS *"that understanding of… the world of work…I was more understanding….taking in other perspectives of views"* (P8). Cognitive crafting in this instance was related to a change in work identity which is crucial for successfully exiting a liminal phase.

Employer 1a, who in the culture section suggested ownership culture was embedded more with individuals than within the organisation, talked about how each of their three WPS had taken ownership of various tasks each year:

"XXX basically took the PMB report off me...He said, is it alright if I just smarten it up a bit. I said, do what you want...I was like, wow where does this come from. XXX took over all the health and safety stuff, he just did new reports, new things all around the rooms...The sustainability award. He just took that on" E1 This building of a job role biography resulted in each WPS in turn job crafting their placement experience to meet their strengths. P7 was the current WPS, but when comparing her experience to other WPS realised that there was no structure in place to bring the WPS together in a formal placement scheme. She took it upon herself to suggest this idea to her supervisor and once they signalled their acceptance, she met with senior staff in other departments including Human Resources. This resulted in a more formal placement scheme being invested in across the organisation for the following year's WPS: *"we want to change that into a proper scheme"* (P7). This "taking ownership" in an organisation where ownership fluctuated, required the WPS to personally invest psychic energy to create, control and bring enthusiasm to her idea, but resulted in great satisfaction, the feeling of adding value and providing a legacy for her placement:

" Getting that set up is definitely one of the achievements...we're starting a new element to XXX that could potentially carry on for the rest of XXX's lifetime. In a way I feel like I left my mark on XXX now when I leave, it's going to be kind of something that I've created still there." P7

Within this job role, there was the core role which might slightly differ each year depending on projects, but then each WPS took ownership for an area of interest and harnessed their skills via job crafting. This reframing of the job role was emphasised in the handover book that each added to during the divestment process. This ritual of the WPS being able to inform the next WPS allowed them to include information depending on what they believed was important and demonstrates a key form of cognitive job crafting (Wrzesniewski and Dutton 2001). P7 suggested that where the book had been factually based on the completion of tasks, she had focussed more on business etiquette. The liminal process moving from student to professional identity (from who she was, to who she wanted to be) was reinforced in this ritual:

"kind of having more of a positive, um, attitude towards things and the way you put yourself across and networking and things like that, I think I've put more of an element of that into my handover." P7

This notion of "me" shows how roles and individuals become entwinned and over a period, each individual adds a small piece of themselves. This handover book is both a moment of pride, a key memory maker, but also allows the individual to positively 'contaminate' the role. Contamination is suggested to be a means of integrating possessions into the extended self (Belk 1988; Watkins et al. 2015) and by "taking" ownership individuals did feel able to transfer the essence of the role to become "theirs." Whilst this need to transfer the soul of a job role was important when both "giving" and "taking" ownership, it seemed particularly important when individuals were replacing a previous WPS. This was not due to WPS feeling that there was negative contamination attached to the role, more of a desire to be part of the heritage of the role and to connect experiences (Kleine et al. 1995; Ahuvia 2005).

Gift giving is also suggested to be a means of extending self (Sartre 1956) and therefore for some, handing over the job role may have similar outcomes, as it cements an individual's place in the office biography. This may be particularly true for ECPs, as they have had limited opportunities to extend self in this way. Further research is required, but this could be a way in which organisations bind WPS to the company to benefit the graduate talent pipeline.

There were other examples of different forms of job crafting. P19 very strategically developed the relational boundaries within his role. He decided early on to form a strategy whereby he would meet someone more senior for coffee each week and then ask them to introduce him to two further contacts that he could meet. This built a network around him which he was using in both job role and future career crafting and would be considered a "gains" approach of promotion-orientated crafting (Bindl et al. 2019). This relational crafting, combined with task crafting via his missions and skills crafting for his future development, *"always thinking about how our work is going to contribute to…..getting back on the grad scheme"* (P19), demonstrates how much job crafting can be taken by some early career professionals. Given that P11, 12 and 15 also demonstrated many forms of expansion-orientated crafting Naeem et al. (2021) and demonstrated high levels of job-related PO, this does suggest some form of connection between job crafting and job-related PO:

"So, my regular day-to-day job; carry it however way I needed to, however there's always areas like, maybe we can look into this and then if the answer is yes, then can I then look into doing it myself, and exploring other areas" P11

Nevertheless, more work needs to be undertaken to ascertain the exact relationship between job crafting and PO. Bindl et al. (2019) suggest that job crafting fulfils a need for relatedness, autonomy and competence, which bear similarities to PO motives and therefore further work concerning how these constructs differ and work together would be beneficial.

Overall, however, once ownership had been given and permissioned to WPS, many used forms of job crafting to systematically take ownership of tasks, relations, and skills, facilitating the development of job-related PO and their emergent identities knowing that they were supported:

"the good ones....identifying maybe..."this process seems to be taking a lot longer, what if we change that we could do that a lot quicker um"...they've got the free range to suggest the ideas to maybe get on and make the processes and changes to- to be proactive as well" PDA1

Nevertheless, E3 also cautioned that decisions needed to be taken by the right individual and WPS needed to understand what ownership was within their realm:

*"transference of tasks in an appropriate manner is easy to do, but then people have to understand that. So if something naturally should come to me, it should come to me, no matter what it is" E2* 

These ownership boundaries are socially constructed and along with permissioning need to be clearly identified. P11 was always aware of these

boundaries and not overstepping them: "how far you can take it...take into account the other factors like how senior someone is" (P11).

Overall, the results from this study partly concur with previous studies regarding suggested PO routes, such as control, investing self, self-congruity, intimate knowledge and some elements of safety, such as trust and perceived organisational support (Pierce et al. 2001, 2003; Pierce et al. 2004; Van Dyne and Pierce 2004; Zhang 2020; Morewedge 2021). The entwinning of self and the role resulting in job-related PO occurred guicker if ownership has been permissioned and the WPS feels trusted to take control and invest self by changing processes, procedures and/or mindsets. If there is self-congruity with the job role, or it can be crafted to build self-congruity, this also built jobrelated PO. Those with high levels of self-efficacy feel motivated to take ownership because of their need to demonstrate competence, whilst for others, cognitive crafting linked to their identity and belonging helping build a feeling of "mine" or "ours". Fulfilling PO motives of belonging and identity may also inadvertently help to facilitate 'boundary-hopping' from "student" to "professional" and therefore move through a liminal space (Mele et al. 2021). Whilst further research needs to be undertaken, a tentative conclusion may be that the elements of "giving and taking PO" aids the liminal process, allowing the WPS to build confidence and speed up the transition through to a professional identity. The certainty that the permissioning of ownership brings to WPS takes away some of the boundary 'fuzziness' (Mele et al. 2021) characterised when ownership is not given or there is no demarcation of responsibilities.

Nonetheless, for this group of participants, intimate knowledge of the job role was viewed less frequently than other routes, partly because of the short time the participants spent in the workplace. However, as seen in examples, intimate knowledge can also be achieved by taking absolute control of a piece of work or project. In this instance intimate knowledge refers to the intensity of knowledge, self-congruity, and self-investment that the participants felt towards their job role or tasks, rather than the intimate knowledge that comes to us in the workplace from a long period of time. The attractiveness of the target often related to the chance to manage a certain task or project alone, rather than working with another individual such as the supervisor. This suggests that PO does not need to be time-bound and other elements, such as PO motives, the process of giving and taking ownership, as well as the attractiveness of the target, may be as important. Of the forementioned targets this study concurs with (Pierce et al. 2001, 2003; Pierce et al. 2004) that control and investing self are key routes, though self-congruity and trust were also significant. By being offered opportunities to manage a certain task or project alone, rather than working with another individual or through the various forms of job crafting, there was a greater feeling of control and sense of mine.

The following section will consider the many outcomes related to feelings of psychological ownership.

#### 4.11 Theme Three: Pleasure and Pain

Theme three identifies job-related PO outcomes for this group of early career professions. As mentioned previously there is a duality to PO which can on the one side provide very positive job-related PO outcomes for both the WPS and the employer, whilst also causing some negative outcomes which seemingly relate to the strength of job-related PO felt by individuals.

#### 4.11.1 The Pleasurable Side of Job-Related Ownership

Scholars have suggested that experiencing job-related PO can provide pleasure-producing feelings, such as job satisfaction, self-esteem, affective commitment, in-role performance, pro-active work behaviour, joy and job engagement (Van Dyne and Pierce 2004; Mayhew et al. 2007; Avey et al. 2009; Avey et al. 2012; Knapp et al. 2014; Peng and Pierce 2015; Wang et al. 2018; Zhang 2020; Henssen and Koiranen 2021), which this study concurs with and which will be discussed further below. Nonetheless, the feelings for ECP have probably been underestimated, as the intensity of delight and pleasure caused by successfully completing tasks far surpass those reported previously: "Then I would feel really happy....I was like, that was a really great day, what an achievement" (P12), demonstrating the satisfaction of a job well done (Bernhard and O'Driscoll 2011; Avey et al. 2012). As experienced workers, it is sometimes hard to remember that first time when you worked on a task or a project which took you out of your comfort zone into unknown territory. The exhilaration of taking ownership of something, entwining your own psychic energy and creativity into a piece of work, which more experienced professionals then praise, was well beyond the expectations of these WPS before they joined the workplace. They are still mentally tangling with their own sense of self, regularly moving betwixt and between student and professional identity, and then they are rewarded with the pleasure of successful ownership, "Yeah, I felt very like wow, this is this is ours. And actually like people are saying it's really good" (P10).

Henssen and Koiranen (2021) study of CEOs in a family business highlighted the joy of work felt from PO on an individual and collective basis. This study partly concurs with the notion of the 'joy of work', although the perspectives and influences were quite different, due to ECPs' relative positions in the workplace.

As a group of early career professionals, most had not experienced ownership at this level previously and it had a profound effect on their self-esteem (Liu et al. 2012) *"So I felt a sense of like, just shock as well....Look at me I'm doing this"* (P12). The three key suggested PO motives of efficacy and effectance, selfidentity and sense of place / belonging were often fulfilled (Pierce et al. 2001, 2003; Van Dyne and Pierce 2004; Pierce et al. 2009a; Brown et al. 2014b; Dawkins et al. 2017). Other previously posited motives include accountability and territoriality which, alongside the motives, form part of promotional and preventative PO (Avey et al. 2009). Whilst the inclusion of territoriality has been debated (as mentioned in the literature review), little discussion has focussed on accountability. This study finds evidence to suggest the inclusion of accountability, along with (Avey et al. 2009) proposal, that it can be developed via beliefs of holding oneself and others answerable. P8's response was typical of how the motive of accountability spurred her on:

"Success, because you're going to see something through, you're going to see it to the end, you're going to want it to go well, because if its associated with you, you don't, you don't want to be associated with something that's failed" (P8)

Being able to justify your actions to others is a key element of accountability (Tetlock 1999) and this separates it from responsibility, although at times the words were used interchangeably by the WPS and employers: *"Responsibility is always good" (E3),* and P15 noting that *"I think ownership means that I feel responsible for how I perform and how my region performs."* (P15).(Pierce et al. 2001; Pierce et al. 2004) have suggested that responsibility is an outcome of PO, which this study concurs with. As discussed previously, P15 felt less ownership for other regions such as the UK, believing that his supervisor led this area and was thus more accountable. In contrast, in the EMEA region, where he had invested himself to a greater degree, he perceived himself as leading and so was motivated to be accountable for its outcomes. Other WPS, such as P1 and 12, also highlighted working with others to be accountable for their actions.

The way in which individuals fulfil PO motives seemed socially constructed. The organisation, and individuals within the organisation who "gave ownership", role-modelled expected behaviours, thus allowing motives of self-efficacy, identity, sense of belonging and accountability to be satisfied. P17 stressed the sense of belonging motive, but this affective commitment (Mayhew et al. 2007; Liu et al. 2012) was also felt by P4, 11, 16 and P18:

*"it's just a feeling of belonging between everyone. Everyone understands what we are all here to do and everyone understand that we should help each other, and we all go out of our way to do that." P17* 

P17's desire 'to be part of something' resulted in a social construction of ownership, in which (he perceived) everyone understood their role. Whilst this may or may not be the shared perception of others in the organisation, from his perspective, he felt that this shared understanding was a fundamental part of PO. Individuals owned their workload to be part of the collective with what P17 described as a resulting positive relationship with people (Howell and Hill 2009).

Nevertheless, not all PO motives were identified in this study. Territoriality is posited as a form of 'preventative' PO by Avey et al. (2009), highlighting the cognitive elements of this construct, although (Brown et al. 2005b; Brown 2009) it is also a behavioural expression. This study found few examples of territoriality and so is unable to confirm or deny its inclusion. Stimulation has also been suggested to be both an 'arousal activation' and a 'territorial satisfaction' (Pierce and Jussila 2011; Rantanen and Jussila 2011; Jussila et al. 2012; Jussila et al. 2015), although as mentioned in the literature review, very few scholars have referred to it since its inclusion. This may be partly due to the challenges of measuring activation in the workplace and the individuality of the construct (Hackman and Oldham 1976), which also puts it outside of the scope of this study. Nevertheless, it does seem likely that individuals are motivated to find stimulation in the workplace (Pierce and Jussila 2011) and further research in this area may be beneficial.

For those individuals who felt high levels of ownership, PO motives were often enabled through the control, self-investment, self-congruity and safety routes, resulting in the development of WPS confidence and pride. Taking ownership, doing a job well and being praised for it gave most WPS immense pride (P1, 2, 4, 7, 8, 10, 12, 14, 16, 19, 20 in particular). P12 provides a common example of the sense of pride felt by the WPS:

"But also pride in your work as well. Like that's my task and that's my responsibility... I see myself as owning that day, because that's my day that I run" P12

Yet feelings of pride are not sufficiently covered in PO research, other than in the works of Di Muro and Noseworthy (2013); Kirk et al. (2015); Ahuvia et al. (2018), whose research focuses on PO from a consumer perspective. Nonetheless, similarities were found in this work-related study. Authentic pride formed via competence and self-investment (Tracy and Robins 2007) was observed in most WPS: *"wow, I've actually done all of that*  from starting with nothing" (P4), with only P3 struggling to provide examples. The PDAs and employer 1 concurred, all expressing how proud many WPS felt owning tasks, projects and their job role: *"they use the word proud quite a lot*" (PDA1); *"They feel such a sense of pride in what they'd done*" (PDA3) *"pride in their job"*" (*E1a*). There were fewer examples of 'hubristic pride' (Tracy and Robins 2007), which relates to the evaluation of one's traits or attributes as superior, although this may be because the WPS may not have wanted to show this sort of negative pride during the interview (Kirk et al. 2915).

Feedback from supervisors and others in the organisation was important for a continuous cycle of accountability, feedback and pride, with the WPS professional identity feeling validated by managers:

## *"If I owned projects that have positive feedback and, you know, I've had that pat on the back then I definitely feel proud" P7*

Li and Qian (2016) suggest that job-related PO results in individuals proactively seeking feedback, and whilst P15 suggested an early adoption of this during his first meeting with his Early Professional Manager, when he asked her *"what right now, would you change?"* (P15), in most instances, feedback was not initially sought. More frequently, ownership of small tasks resulted in feedback and praise which then gave the WPS confidence to take more ownership. The PDAs talked about this circle of validation, including this example from PDA2:

" I think ownership also comes from good feedback from managers as well, it sounds really bizarre, but I suppose that they get their validation from the feedback...and that makes them own it more I think, that makes them like the job more" PDA2

Whilst pride often fulfilled the efficacy and effective motive, pride is also a means of fulfilling the identity motive. Having experienced a liminal phase, many felt pride and confidence in their new selves: "*I feel like I am an equal to the other people in my team*" (P14) and this tended to be the only occasions when hubris pride was observed (Tracy and Robins 2007), although this statement is as much about P14 changing identity and self-efficacy. The PDAs concurred, suggesting that many WPS returned to university with a different identity. P12 talked about how the experience had changed her and how frustrated she was

with her old self for not always going onto campus for lectures and seminars during her 2<sup>nd</sup> year at university. She felt she had been lazy, but this experience of work, owning tasks and her job role brought her the satisfaction and pleasure of doing a job well, giving her a picture of her future self: *"I'm actually embarrassed...because it's like, what was I thinking, what was wrong with me"* (P12).

Confidence in presenting, talking in meetings and confidence in the quality of their work were all by-products of taking ownership for their job role: "They made that advert; they wrote that copy. So that obviously builds their confidence" (E2). P19 suggested that because he felt believed in and trusted to take ownership, this built his confidence in his own abilities. Nevertheless, this confidence can be fragile at times, as P1 discovered after making a mistake: "there was a bit of push back....and it sort of knocked my confidence a little bit" (P1). Nevertheless, with the help of his supervisor, he rebuilt his confidence and was then happy to offer to take on an additional role, "so I felt at that point I felt confident enough in myself and in the job role" (P1). All WPS identified an improvement in confidence, except for P5/P13 who did not mention it during the interview. Even P3, whose confidence was dented at times because of her supervisor, felt that her confidence had developed "to some extent" (P3). Although this was a WPS who joined her organisation with greater efficacy and effectance from her previous job role, which didn't fully transfer to new surroundings. (Edwards 2014b) study of WPS also found positive self-efficacy following a work placement, so these outcomes are not surprising. It is therefore difficult to ascertain if feelings of ownership satisfied self-efficacy or if this was the by-product of a work placement.

Other than the outcomes mentioned above, those participants who displayed job-related PO felt many positive emotions from feeling job-related ownership, such as the joy and happiness at doing a job well (P15), feeling valued and fulfilled (P4) and giving some of these WPS a sense of purpose: *"Sense of purpose. Sense of achievement when it goes well. Just feeling really responsible, really being part of it"* (P12, but also mentioned by P4, P10 and P20). Meaningful work and PO have been broadly mentioned together (Wang et al. 2018a), but there is no empirical evidence making it a fruitful area for future PO studies.

Nevertheless, confidence and pride were the most frequently stated positive outcomes of job-related PO for this group of participants, both of which link to suggested outcomes of job satisfaction and self-esteem (Bernhard and O'Driscoll 2011; Avey et al. 2012; Liu et al. 2012). For many of these WPS, this was the first time they had carefully crafted and created something that was theirs, but which also had an impact on the organisation. Legacy-building was noted frequently (you may remember P7's example of the placement programme that she and another WPS created within their organisation). P12 was delighted that her supervisor noted her strong contribution, including designing a new and significantly different placement recruitment process: "my manager, XXX....she said, I've left such a legacy in the team" (P12). The term 'legacy' was used by Employer 1, PDA 1 and 3 as well as the WPS. PDA3 suggested the WPS mindset was often mentioned in terms of their contribution: "I want to be able to leave something," or, "I want to be able to leave a lasting kind of impact" (PDA3). Not all WPS (P3, 13 and 16) felt they provided a tangible contribution and given that their ownership experiences were less frequent, or they were not "given" ownership, shows there may be patterns between being given ownership and positive organisational outcomes for both individuals, their supervisors, and organisations. E3 sums up the all-round positive outcomes, that job-related PO in WPS can bring:

"Significant and it's multi-faceted I would say...It's very easy in business I think to get used to what you do and get a bit bored by it, and then when you see it through the lens of somebody else who's coming at it fresh, it can motivate you...But we also like the ideas they come up with, like the fresh way that they do things" E3

### 4.11.2 The Pain of Job-Related Psychological Ownership

There is however a duality to PO which for this group of early career professionals demonstrated more negative PO outcomes.

Whilst early PO empirical research considered antecedents and the positive of feelings of ownership, studies have also suggested other outcomes which may

have a negative impact on individuals and their managers / colleagues. Most frequently posited is the notion of territoriality and how feelings of ownership may lead to claiming and defending behaviours of what individuals consider "theirs" (Brown et al, 2005; Brown et al 2014a; Brown and Zhu, 2016; Kirk et al, 2018), as well as protecting what is "mine" via knowledge-withholding behaviour (Han et al; 2010; Peng, 2013; Peng and Pierce, 2015; Anand et al 2020). This will be discussed at a later point in this section.

Organisational studies literature generally suggests that a lack of individual control may lead to stress (Greenberger et al., 1988, 1989; Lee et al., 1990; Spector, 1986; Terry and Jimmieson, 1999), which has led PO scholars to tentatively suggest this may be a negative PO outcome (Pierce et al. 2001; Van Dyne and Pierce 2004; Pierce et al. 2009b; Liu et al. 2012; Brown et al. 2014b; Zhang 2020). Dawkins et al. (2017) suggests that in the same way that the relationship between job demands and job stresses is curvilinear, this may also be true of PO. Adil and Kamal (2018) are the only theorists who have assessed burnout, suggesting in their study of teachers that burnout has a negative relationship with preventative PO.

In this study, those individuals who displayed signs of high levels of job-related PO, did also display evidence of stress-related behaviour. P12 and P15 were individuals who experienced the highest levels of stress, although for varied reasons. P12 worked in a team managing the recruitment of early career employees at a multinational organisation. This role had previously been undertaken by two people, but a company restructure resulted in it becoming one role. P12 initially managed the role, and was enjoying the responsibility provided, but as the recruitment season continued, she was running assessment centres three days per week, working 12-hour days and struggling to keep on top of the administration involved in recruitment on the other two days. Her story is best shown in her own words:

"But I don't think I actually really truly spoke up about it and expressed how much work I was doing, until the November, probably mid-November and I started to say, whoa, whoa, whoa, whoa, I need help with this" P12

P12, who showed high levels of ownership, often worked long hours, taking on increased responsibility which on the face of it she was coping with. However, she had internalised any anxiety about the workload for fear of disappointing her supervisors. For individuals at the start of their career, there is no prior frame of reference to help them understand fair and achievable workloads, nor how to ask for help without feeling as if they are letting people down:

*"I felt like I was disappointing people by not being able to do everything for everyone, and actually by saying, I need help, and getting help, it benefited me in the long run. But it was just learning how to say, listen that's too much, I'm only an intern here, I've only been here a couple of months". P12* 

Whilst this is partly about feeling a lack of control due to a high workload (Pierce et al. 2001; Van Dyne and Pierce 2004; Pierce et al. 2009b; Liu et al. 2012; Brown et al. 2014b; Zhang 2020), this is also partly about identity. If an individual sees themselves as a competent person and prides themselves on their ability to manage themselves, then in this type of situation there may be a disconnect between their personal identity and social identity (Hillenbrand and Money 2015), with the resulting lack of self-congruity (Morewedge 2021) resulting in feelings of stress. Given this was also likely to occur during a period of liminality (Beech 2011), in which P12 hadn't fully moved from her student mindset to that of a worker, in the extract above she labels herself as "only the intern" (P12), demonstrating the disconnect between what she perceived to be the role of an intern, versus her lived experience. For individuals with perceptions of high competence, the loss of face with supervisors and herself may have resulted in her justifying the challenges she faced by returning to the comfort of a simpler identity.

However, there is also another side of her that was excited by the responsibility and the opportunities presented to her:

"I felt responsible, and I loved the responsibility of it... and I felt like it was on my shoulders more than it should have been. I was impressed that the company would trust me....but yeah, it was more the team that I was frustrated....Why did it take me having to be upset or something happening for help to set in?" P12

This appeared to lead to an internal battle in her mind raging between frustration at the team for not helping her when she obviously couldn't cope with the workload, but pride in being trusted to do what she perceived to be a responsible role:

"it can be too much, especially at a young age when you don't quite know how ... when you're still learning about how to deal with it. Stress and pressure that you might not really expect or should be dealing with at that point in your career. P12

Thus, for those individuals at the start of their career, there can be a desire to people-please and therefore if they feel under too much pressure, they may not feel comfortable to let these feelings be known, either for fear of letting their supervisor or the team down or being disappointed in their own capabilities. The lack of frames of reference regarding how "work works" means that some may internalise this pressure with the resultant build-up of stress.

P15 also felt a great deal of stress, but for different reasons. He was a very competent individual (as suggested by employer 2) who took on a great deal of additional responsibility, but struggled in two ways. By his own admission he wanted to complete all pieces of work the moment they arrived on his desk, which resulted in him working late nights and weekends:

*"I think especially at the start, the kind of weight that I felt on me meant that XXX just ate into my whole life" P15* 

Learning the ropes of a job role is an expected learning curve, but sometimes with strong performers, because of a high level of competence, this seemed sometimes to be forgotten and it was assumed that they were managing well. P15 also had a desire to be involved in lots of different work and had elevated expectations of himself, which also caused anxiety. As mentioned previously, his job role literally possessed him, taking over his life in both a pleasurable way, as he enjoyed being consumed in work, but ultimately with some negative consequences, such as working long hours and constantly thinking about work. He talks about the negative sides of ownership again in terms of stress:

"I think it can have a negative effect, just in terms of stress and tiredness and not being able to focus, not being able to relax and not be able to really spend time with the people that are with you but sometimes worrying about something else." P15

P15 was also living at home, which was an hour away from his office. After one early company event that he was responsible for, he had a car accident on his way home: "I had a slight ... I had a car accident on the way home which kind of tarnishes how that event well" (P15). Whilst this accident is not being attributed to the effects of a high level of PO, it did occur during a period of high stress. By his own admission, living so far from his workplace caused additional stress in an individual who was already consumed by his work. He was however very appreciative of support from his Early Professional Manager straight afterwards: "Early Professional Manager helped me sort my life out...not all companies would have done" (P15). One of his managers, E2, also became aware of his high level of work and discussed this with him:

"And he did get to the point where me and our manager...had a chat where we just said, we don't want you to burn out. The work you're doing is great, but it's too much, no one person can do all of that. I think that's the negative side...it's taking on too much, trying to own a lot of things. E2

Other employers also realised that certain individuals could take too much ownership, which can have a detrimental effect. E3 talks about a graduate who had joined his team:

"She takes too much ownership. She could struggle to delegate, so in her most recent appraisal, where she feels a little bit overwhelmed by how much work she's had, we've had to encourage her to delegate up and also to delegate down. So she was holding it all in because she thought it was all her responsibility." E3

In addition to P12 and 15, P19 also emphasised the burden of trust and responsibility:

"trust puts a weight on your shoulders, which isn't necessarily a negative but it's always at the back of your mind, I don't want to let these people down, they've given me this much responsibility....so there's definitely a level of stress there" P19

However, P19 did suggest that the organisation tried to actively combat stress with resources including on-site mental health support. Nevertheless, as an outsider looking in, I was surprised at how heavy the weight of responsibility seemed to weigh on some ECPs and especially those who exhibited the highest levels of job-related PO. They were both delighted and surprised to be given so much responsibility, but equally, so strong was their desire to do well, they at times struggled with the heavy mental load.

All three of these individuals worked in organisations where ownership was at the core of their organisational culture and therefore it shows that, whilst there can be many positives for both individuals and organisations who have a culture of ownership, there may also be negative aspects, such as stress, which could lead to burnout. Whilst this may not occur in all early career professionals who demonstrate early signs of strong ownership, it may be that this is internalised and eventually leads to burnout (Kaur et al, 2013; Adnan and Kamal, 2015; Su and Ng, 2019). P4, for example, was an individual that had strong feelings of ownership and could see how it could be all-consuming:

"it's a really tricky one, because it's critical to have ownership, but its critical not to own everything. Because you know, if you do own everything, you will just stress out too much and you, you won't understand the true value of what you're doing because you won't be able to invest yourself deeply into everything" P4

Nevertheless, whilst he seemed to work very hard, ownership never possessed him in the same way that others suggested. Why this might be was not entirely clear, but a tentative suggestion would be due to a mixture of personality traits, the organisational culture in his workplace, being slightly less ownership-driven, and also the fact that he may have not been as honest about stress levels. Therefore, it would be wrong to make assumptions, but this study provides us with a first glimpse of how stress may impact the working life of early career professionals and in particular those individuals in organisations where ownership is part of the organisational culture. Most participants were aware of the potential hazard of taking on too much work putting pressure on themselves resulting in stress, however simultaneously they were worried about letting people down, especially their supervisors, and that can add weight to their shoulders. The burden of PO has been mentioned briefly, but mainly in terms of protecting one's self identity, but this may be the first study to demonstrate how heavy a weight PO can be for some individuals. The weight of responsibility better captures the effect that PO may have on some individuals. Further studies in this area would be beneficial, not only using early career professionals as

participants, but also with other groups of employees. Studies might seek to ascertain if there are certain groups of individuals more prone to stress and at what point on a PO curve these feelings might occur.

Previous research suggests that individuals' territorial behaviour may communicate a target as "theirs" by 'marking' or 'claiming'. This may be via anticipatory defensive behaviour, such as putting a coat over a chair or knowledge-withholding, such as not sharing a report with a colleague (Brown et al, 2005; Brown et al 2014a; Brown and Zhu, 2016; Kirk et al, 2018). However, few instances of this type of behaviour were seen in this study.

P10, who worked in a trainee area manager role, did not want to delegate, because she preferred to keep control herself:

"I think, it wasn't so much that I didn't have the confidence to delegate in that I wouldn't feel comfortable asking people it's that I wouldn't feel comfortable asking them because I'd rather do it myself." P10

This control feels less about territorial connotations, but came from prior experiences of student projects where she was afraid of trusting others with her "grade", as she had had bad experiences previously. She believed that she learnt to trust others during that year, allowing her to relinquish some control.

Some participants (P2,10, 11, 14) mentioned their role / projects in terms of their "babies", suggesting territorial undercurrents. P11 struggled to hand back projects towards the end of her placement:

"Very proudly. And also, sad because they're kind of my babies now. It was difficult. It was more difficult than I thought to hand over those projects. I like sharing them but also fully giving them and then ... not even giving them and moving onto different projects, but it was tough for me," P11

However, this feels more like sadness at knowing that a good journey is about to end. It did not seem to lead to territorial behaviour for P11 or the other WPS, and the PDAs and employers suggested that they rarely saw this type of behaviour at this level. This may be partly due to the brief time period that individuals spent with the organisation and their lowly status. Further studies may be able to explore this area.

Towards the end of the placement, it could be concluded that the desire to craft the handover book for the incoming student was a form of identity-orientated marking (Brown et al. 2005a; Brown 2009) for some individuals. In these instances, however, it felt more about pride and showing their distinctiveness, rather than a form of marking their territory.

To conclude, ownership does have this duality, which means that it can be hard at times to balance the outcomes. P15 nicely sums up the positives of feeling ownership and the joy and job satisfaction it may bring, but also how ownership can weigh heavily on individuals if it is not effectively controlled:

"I think ownership means that you work hard...that you are driven and that you care and perform and I think it helps you to get up in the morning, but at the same time, you might not be able to sleep at night. I think especially at the start, the kind of weight that I felt on me meant that XXX just ate into my whole life" P15

### 4.12 Theme Four: Longevity or liquidity?

The final theme highlights the implications of job-related PO on these early career professionals' futures. Does the strength of feelings for the job role influence how they feel about the organisation and how might that impact their thoughts and plans on future careers, given their return to university for a final year of study?

### 4.12.1 Decoupling / Divestment

Decoupling or divestment rituals provide an opportunity to move a target from "mine" to "not me". According to Pierce et al. (2003), decoupling usually occurs due to a withdrawal of one of the motives or catalysts, resulting in

targets becoming less accessible or routes no longer available. However, beyond this initial consideration, scholars have not considered workplace divestment from a PO perspective. Nevertheless, due to the fixed-term nature of these job roles, these participants' experiences may be atypical to more established work participants, although there are comparisons between this sample and other samples demonstrating the divestment of possessions, especially for those on fixed-term contracts.

Whilst many rituals are aimed at attaching the target to the individual's extended self, there are also divestment rituals designed to disengage with targets when an individual leaves a role or organisation (Pierce et al. 2003). Lastovicka and Fernandez (2005) suggested three ways of disposition, depending on the valence of the item: divesting of "never me", separating "me to not me" and disposing to someone similar, and leaving a common identity.

There were very few examples of WPS divesting a role that was "never me". As you are aware by now, P3 did not show examples of job-related ownership whilst on her work placement and so it could be said that her experience is not relevant to this study. Nevertheless, this experience is comparable to someone disposing of an unwanted possession and for those individuals who feel they never fit in a job-role, so this may be a comparable experience.

P3 never felt suited to her job role, was moved between different departments, and therefore felt that the routes of self-congruity, intimate knowledge and investing self (Pierce et al. 2001; Morewedge 2021) were closed to her. At the six-month point of the placement, her line manager told her she was not on track to come back as a graduate and therefore she entered a liminal phase, decoupling from "never me" to considering what was "more like me". This process was undertaken in two ways, firstly by ascertaining potential future career options via a continual analysis of what was or wasn't "her" in terms of future career options, and secondly by distancing herself from the role and her "past undesired self" (Lastovicka and Fernandez 2005).:

## *"I've said this to a few people now, but if I ever spend another week in front of the computer where all I have to do is type, I think I might go crazy"* P3

This desire to move away from "never me" was so strong that she had to disassociate herself with key elements of the role, both publicly and

privately. During the interview, she later stated that if she was interested in the role, it would be fine to be in front of a computer, but the need to publicly disassociate with elements of the role in the first instance was required for her to be able to move through this liminal phase. Her desire to move as quickly from "never me" to "more like me" left her feeling uncertain and *"like a 10-year-old again"* (P3) revelatory of key elements of being "betwixt and between" identities (Turner 2011). Her identity confusion was all the more powerful to her, because of her previous strong ownership experience in a part-time role. However, her reflection on this experience from being efficacious in one environment (p/t role) to a complete lack of confidence (work placement) helped her look forward and move on from this liminal space: *"I'd say one of my downfalls has definitely been not being able to just accept what I know"* (P3)

There were examples of positive valence when WPS were leaving a job role that they enjoyed. Divestment rituals were frequently employed to change "mine" to "not mine", especially if individuals were handing over to a new WPS. Rituals allow owners to cleanse and transition meaningful elements of the job role from their life, whilst still maintaining private meanings. The transference of a 'handover book' from one work placement student to another was a symbolic form of facilitating the new WPS transition into the workplace, but also providing the outgoing WPS with an opportunity to parade their "intimate knowledge of the job role" and demonstrate their legacy, as typified by P7 when discussing her input to the handover book:

"You can teach anyone how to use a system but you can't always teach people to smile to people when introducing themselves...that's a new element I kind of brought to the role" P7

In addition, the WPS also frequently sat on the interview panel for the new WPS, providing them with a form of control in the decoupling process by expressing who "their job" should pass onto: *"so I'll be interviewing my replacement, which will be fun"* (P14). In the same way that the Mere effect (Heider 1958; Beggan 1992) suggests we favour items that we own, passing ownership to someone we favour may help facilitate the divestment process.

Those individuals not handing over to a new WPS used ionic transfer rituals (Lastovicka and Fernandez 2005; Peck and Shu 2018) to facilitate the divestment process, such as photographs taken at the organisation, last day messages on LinkedIn, and by completing a self-reflective portfolio as part of their coursework, which PDA one suggested allowed them to be proud of their achievements and professional development. P8 completed a final presentation and was delighted that her friend took a picture as a reminder: *"I have a picture of my slide, my friends was really good and she got photo"* (P8).

There were also examples of shared meaning beyond the team and organisation which includes prior WPS. As mentioned previously, employer one discussed how their WPS had started an "alumni board", including nicknames for current and former WPS. There was a big sense of 'belonging' in this department, beyond its current staff members, reminding us of the narratives of ownership. Biographies can go beyond the current and new incumbents and the ghosts of previous owners can also play a role in the divestment process, by making individuals still feel part of something even if they are not there.

There were also occasions when individuals could disinvest to someone "like them". This could also include the divestment rituals mentioned above. Whilst having some control in the recruitment process and investing self in the handover book made this process easier, the PDAs were aware that this transfer of ownership was still not entirely easy due to a duality of feelings. PDA1 expressed the tensions often arising:

" they also- they want to be the ones to produce the very best handover notes to train up the students as well because...if they've got to hand it over to someone, then they want it to be a good job, and yet they don't necessarily want them to be slightly- quite as good as they were...it's something like I want to be the best person handing over, but I don't want them to be quite as good as me." PDA1

These feelings remind us that divestment is never truly easy and whilst the cognitive element of PO will suggest the pragmatic, practical response, there

may also be a spectrum of emotions felt by individuals regarding the loss of ownership and once again finding themselves betwixt and between identities (Turner 2011). High levels of PO may cause divestment challenges, with employer one suggesting that both former WPS frequently visited their department post their work placement:

"We can't actually get rid of our previous placement, they keep on coming back....it's only recently that XXX has dropped off the radar really, since he's graduated, otherwise they were here all the time, all of them, we just couldn't get rid of them. E1

Therefore, a keen sense of belonging and identity with the team can have implications for a successful divestment. Other WPS at different organisations, such as P11, found it hard to hand over her projects, as she thought about them as "hers". Whilst she was happy to share, full divestment was a challenge. Given this was likely to be the first time that she had divested her creations, there is a small sense of personal loss, especially in this instance, as the organisation's discussions about a graduate scheme at that point had not made any decisions:

## *"very proudly and also, sad because they're kind of my babies now. It was difficult. It was more difficult than I thought to hand over those projects" P11*

Other outcomes for those individuals who demonstrated strong ownership feelings were more mixed. P12 had been offered a place on the graduate scheme, which made the disinvestment process different, as she was mourning the job role, not the organisation. P4, 15 and 19 were still considering options, which will be discussed further in the following sections.

Nevertheless, some WPS (P2, P10, P12) knew that they were returning to the organisation post-graduation, were continuing to work part-time for the organisation (P9) or (P1, 14, 15 and P20) being fast-tracked to the final stages of the Graduate recruitment process. Therefore, their divestment experiences may be different to the norm, as they had both the security of returning to university and the strong possibility of returning to the organisation.

PO divestment should be of interest to scholars and organisations, due to the opportunities to consider out-duction processes and opportunities to retain and

the possible re-recruitment of alumni. The transfer of ownership from one individual to another may be full of emotions, even when the outgoing employee has chosen to move roles, and further research with both incoming and outgoing employees could be beneficial.

Nevertheless, there is potentially a new PO target which may both strengthen PO feelings and result in continuous career changes.

### 4.12.2 Career-related Psychological Ownership

One unexpected ownership target mentioned by WPS during their interviews was career ownership. Whilst two previous PO studies mention PO as moderating the relationship between career ownership and work-related outcomes (Olckers and du Plessis 2014; Olckers and Koekemoer 2017), this study takes a different approach, suggesting that for some their career is an ownership target, rather than an antecedent of career success. Career PO allows individuals to fulfil traditional PO motives of self-efficacy, identity, sense of belonging, stimulating them to make individualised choices such as P8 talking about ownership:

# *"massively, especially like terms of career. Because at the end of the day, the only person you can really rely on is yourself. So I kind of own my future" P8*

This desire for career ownership appears to be partly constructed via school, university and parental influences regarding the rhetoric suggesting career flexibility rather than "a job for life", but was raised independently by some participants (P1, 5, 8, 15 and PDA1) when discussing their beliefs on workplace ownership. P1 suggested a duality to ownership, which P5 concurred with:

One...being on a day to day role perspective of owning the requests...and owning the projects to make sure they are delivered...at the same time it's the ownership of myself...if there are opportunities out there...it's down to me to make sure I do that" P1

PDA1 also agreed with these WPS, suggesting she often heard surprising outcomes from WPS when they discussed this duality. Below she mimics how

the WPS often spoke about the tension between forms of ownership during their final meetings:

"Well you know, sort of like, yeah, I've had a great time, but I want to do this, and this"...so- so they have- they have two different forms of ownership, they feel very much integrated while they're there, but there's also...that kind of like, um, I own my own career, and it's not going to be just fixed in one place for a long time" PDA1

This perception has connotations of a boundaryless career (DeFillippi and Arthur 1994), yet in some instances where job-related or organisational PO is also high, some individuals seemed happy to remain in their organisation, such as P2 and P9: *"hopefully I can be a XXX for years to come"* (P2). Therefore, career PO may or may not be boundary-less depending on the individual and other PO targets. Consequently, whilst PO targets may align simultaneously with career PO, if there is tension between targets, the strongest attachment may take precedence, which will be referred to later in this section.

Donald et al. (2019) suggest that individuals who take career ownership forge a protean career (Hall and Moss 1998), whereas those individuals who do not own their career are more likely to establish a more traditional work life. This study partly concurs, suggesting those who seemed to demonstrate strong feelings of career PO were more likely to look at a variety of different career opportunities: *"this year has made me realise that I'm probably going to start my own business in the summer"* (P15).

However, whilst a protean career does have similarities to career PO regarding self-direction, there are significant differences, with protean careers answering the question, "how can I be effective in managing my career", whilst PO asks, "what do I feel is mine" (Brown et al. 2014b). When P15 was asked if he owned his career, he suggested "Yes, 100%. 100%....I think a career is like, *it's almost your life" (P15)*. Donald et al (2019) and Hall and Moss (1998) suggest protean career orientations are a mindset driven by the individual rather than the organisation towards advancing their careers (Hall and Moss 1998; Briscoe et al. 2006; Hall et al. 2013). PO is also driven by individuals who have a desire for self-direction "*owning my year.... to plan it so....I can get the best possible year for myself*" *(P1)*. Nonetheless, because

PO has both cognitive and affective elements, it goes beyond a rational desire to shape one's own work destiny, to include this emotional connection of "mine". James (1890) suggests our psychic powers are as much a part of us as tangible possessions, and it is the sum of everything that equates to the "self". The example with P15 above shows how for some individuals with high levels of career PO, the target can become entwined into self, which goes far beyond being in control.

Furthermore, protean careers also highlight an individual's values and desire for meaningful work, and whilst rituals may make ownership meaningful to the individual, it does not necessarily follow that this is value-laden work. In fact, some WPS, such as P1, showed a strategic ruthlessness in ensuring that the work he took on could be beneficial for his career above all else:

"as it's on the radar of our UK CEO....opportunity to stand out ...Which means I have taken the decision to stand down from the XXX role... But even when I think about my part time job or the football media... if the opportunity doesn't impress me enough....I don't get involved. But if I feel... I can learn something here or I can really make an impression, I would take up an opportunity." P1

This strategic ownership of their career was also observed in other participants, such as P19 and P20. P19 ritually built his network via weekly coffee meetings, asking the staff member for two names of other staff members that he should also meet. The organisation systematically encouraged this strategic thinking in the form of a final ritual of presenting to the Commercial Officer, demonstrating the business impact they had made over the year. Whilst the organisation used this as part of the criteria to consider WPS for their graduate programme, P19 also utilised this as a means to gain experience, which could be promoted both internally and externally in a pursuit of their chosen career:

### "me and two other interns, we are always thinking about how our work is going to contribute to us umm getting back on the grad scheme, and I think that ties into my career ownership" P19

Nevertheless, the strength of career ownership did differ from those forementioned examples in a similar way to job-related PO. An example of

limited career PO was P17: "Ownership of the career? (pause) Not particularly, to be honest" (P17). P16 had also not considered career ownership previously and whilst the strength of feelings was on the low side, she did suggest taking ownership of whether she would like a career in the future: "I will take that responsibility to find my own career and if I don't want a career, I won't have one" (P16). However, whilst Donald et al. (2019) suggest those without career ownership are more likely to take a traditional future path, this was not always the case with these WPS, with P16 suggesting her plans involved a "move to London and who knows, I don't particularly want...a grad scheme" (P16). P17 concurred, suggesting he would travel and then look at a range of opportunities, which might include his placement organisation but might not.

In other WPS, the career target was not as concrete in structure as other PO targets initially, but because it is more malleable and easily available, there were greater opportunities for the emotional connection to develop. P12 was a typical example of one of these individuals:

I've never known really what I want to do with my career since ever...I've never known...I've really learnt what I like, learnt what I don't like, learnt what I'm actually really good at and I now think, right go, you've got this opportunity and figuring out what it is. So I definitely want to take ownership of my career" P12

Career PO, like other forms of PO, involves psychic energy to cultivate, care and control (Sartre 1956; Csikszentmihalyi and Rochberg-Halton 1981; Ahuvia 2005; Watkins et al 2015). Like the opportunity to job craft to make the job role "mine" (Wrzesniewski and Dutton 2001; Demerouti 2014), individuals reframed the boundaries of their careers to adapt to new opportunities and possibilities. By taking control and investing themselves into their career-related goals, some of the WPS were forthright in making decisions. As mentioned previously, P4, early in his placement, spoke to his manager about a desire to move his role in a different direction, and was lucky to be able to job craft the role with his manager's agreement:

"in about the first couple months, I spoke to my manager like that quickly, and I said, I love I love what I'm doing now about really, my eyes kind of wavering over to, to, to being customer facing. So we started looking at

### different kind of opportunity ....started talking...to build up the networks saying I really want to do this" P4

As mentioned in the literature review, individuals can craft their career, as well as their job role, using both proactive career reflections and construction (Tims 2020). Examples of both forms have been found in this study (in particular P1,3,4,7,8 11,12, 13,15 and 19 for proactive career reflection) with P7 providing a typical example of her values, passions, expectations, and future possibilities:

"a company that care about their staff...work-life balance is really important but also...responsibility over things and being passionate about where I'm working...I'd love to be like area manager...I like that fast-paced environment, but at the same time you don't have that work-life balance....that would made me really unhappy" P7

There were many examples demonstrating proactive career construction with the strength of career ownership seemingly dictating how much psychic energy was invested. P1,11,15 and 19 seemed to invest the most time, with P11 being able to invest some time in work-shadowing three different areas: *"I couldn't make my mind up...she...gave me 3 weeks to do shadowing"* (P11). In the same way that job crafting seems to be an antecedent of job-related PO, it feels that career crafting could be an antecedent of career-related PO. Nevertheless, far more research into these areas need to be undertaken to establish any connection.

What also requires far more research is what role career PO may play in an individual's life and how organisations can facilitate individuals to gain job role and career PO self-congruity (Morewedge 2021) or provide future internal opportunities. Baruch (2006) suggests that scholars are quick to talk about the demise of the traditional career vs new careers, whereas in reality it is more nuanced than this, with some people and organisations still demonstrating traditional career options, whilst others veer towards boundary-less options. Therefore, there will be forms of careers that work for all. What may be interesting about career PO could be the relationship with job and organisational PO. At many points in an individual's career, you would hope

for alignment between these targets, in that individuals are in the right role and right company for their career aspirations. However what occurs for example if the desire to take ownership of their career is extraordinarily strong such as P12: "*I definitely want to take ownership of my career, where I go & I definitely have an idea of where that is*" (P12). Even if there is an attachment to the role and the organisation, may there also be a desire to continue on with the career master plan. More research in this area is required, but this does provide another approach to conceptualising careers. How does owning their future career then link to their identity and liquid ownership? The following section will combine these two areas.

### 4.12.3 Future Selves and Liquid Ownership

Ibarra (1999); Ibarra and Barbulescu (2010); Ibarra and Petriglieri (2010) suggest that professional identities are malleable at the start of a career, as individual's provisional selves help them to move from one identity to another. Towards the end of these individuals' work placements many of these participants felt clearer in their identity, ascertaining who they were and who / where they wished to be post-graduation.

Garsten (1999) suggested that temporary workers could find themselves in a constant liminal state in the workplace, but this was not demonstrated with this group of participants. This seems to be partly due to the nature of WPS, in that they have a fixed date to return to their previous life, but also partly due to the efforts of organisations to facilitate the process of making them feel part of the team or organisation. As we have seen, P3 and 13 did struggle, but this was more related to supervisory problems, rather than temporal concerns.

The digitalisation of possessions does not seem to have altered how these WPS develop feelings for workplace targets, in that they seem to cultivate, shape and control items in a comparable manner to that described in other studies. Therefore, I concur with the assertions of Belk (2013); Watkins et al. (2015); Denegri-Knott et al. (2020) that the process of ownership for digital possessions is like other possession ownership including intangible ones.

As mentioned at the start of this research, work placements are often utilised as a 'talent pipeline', and many of the organisations had strategies in place to extend this talent pipeline via opportunities to work shadow or have careerrelated meetings with more senior staff. Organisation 2 managed this process particularly well, with P12,14,15 and 20 all suggesting that they had met with individuals within the organisation to help evaluate future opportunities and/or discuss their career aspirations. Many were keen to return to the organisation, but the graduate roles were targeted into two key areas which differed from their placement role. Using an iterative process of identity matching, renegotiation, pragmatism (that this was a good company in a convenient location – P20) and the desire to be authentic, individuals danced between current provisional self and future selves (Ibarra and Petrigileri 2010), reconstructing their identities partly via a social construction, involving both an outside-in as well as inside-out orientation work (Beech 2011).

P14 experience was indicative of the experience of this organisation's participants:

"I've always thought, oh it's not for me, but I've decided to actually look into it now. So I spoke to my HR mentor...I had a call with that person... she knows someone that's more specific...I've got a call with her on Thursday.... I really want to go back into XXX not just because it was be easy for me...but...because I really like the company." P14

There did seem to be a genuine attachment to organisation 2, hence the desire to stay with this organisation, especially once discovering there may be a role that fits with their perception of their future self.

Nevertheless, within this group there was also some pragmatism to look at other organisation's graduate schemes, in case they were unsuccessful in their primary graduate applications or due to career PO, which took their plans in a slightly different direction. P15 was an example of this, as he planned to set up his own business over the summer months at the end of his placement. If he made more money from this venture than his "day job", then his future path would be running his own business, "So if I can make more than I was making in my day job at XXX, I know that it's a route I could go down" (P15). His self-efficacy (Bandura 1977; Banudra 2012) and proactive career construction (Tims

2020) was such that out of all the WPS at this organisation, he was most likely to proactively explore other opportunities, due to the strength of his career PO.

Those WPS who identified most strongly with the organisation were not necessarily the individuals who had the strongest desire to return. As mentioned in the previous section, P1, whilst suggesting a strong desire to return to the organisation, also mentioned a strong desire to move into football, where he currently worked part time. This was a prior area, where he had demonstrated PO from a part-time job role and thus, if opportunities became available in both sectors, this would result in a very tough choice for him:

"if you had to choose between xxx and a premiership football club, I would say it's too tough to answer (laughs all round) because I would absolutely love to work at xxx...so working for a digital...company makes sense...that's why for me it's a case of now if xxx and football came up, that would be a very tough decision (P1).

As mentioned in the career ownership section, occasionally these WPS did have strong attachment to two different targets, which resulted in a tension. Whilst no current PO research suggests the mental battle that may take place between two equally attractive targets, it seems likely that the more open, accessible and 'manipulatable' target will win. Nevertheless, further research in this area would be beneficial.

P11, whom you may recall talking about her organisation and job role as her "first love", did have a strong desire to return and whilst the organisation was trying to introduce a graduate programme, progress was slow. Therefore, as the organisation could not guarantee a job role (even though they were very keen to hire her), she used her network and opportunities to work shadow, to explore future areas of interest:

"especially doing the shadowing; I love working for a big company. I think that has opened a lot of doors. I already get some people messaging, oh you're interested in this, you're interested in that...However...sometimes on my shadowing, I actually didn't shadow the specific role,.. it was a person that I was interested in" P11

P11 used "future selves" modelling (Markus and Nurius 1986b) as a means of career decision-making. She was interested in hearing people's career stories and from this started to build a list of future prospects.

Nonetheless, there are also some indicators of liquidity within these words and the words of other WPS. PO in this instance was always going to have a temporal element because these individuals were undertaking a work placement, but for some individuals such as P11, whilst she discussed "organisational first love", when talking about the organisation towards the end of the interview, she quickly moved to expressing interest in other roles and other organisations. This "first love" felt fleeting, rather than lingering, and in the same way that global nomads valued possessions at a certain time and place (Bardhi et al. 2012), there were elements of ephemerality in their conversations:

"But I couldn't make my mind up on three different areas of the business...she actually gave me three weeks to do shadowing in all the different areas. I loved all of them and I do think that in a different area I can see myself ... I can see myself going in to." P11

Whilst this and other similar comments (P1, P3, P15) could be considered as a sign of liquidity, it could also be considered as career PO and then tension between attractive targets, or, as mentioned previously, pragmatism of their situation which influenced these comments. Nonetheless, for some participants this temporality did allow them to justify different decision making.

Immediacy (Bauman 2007; Bardhi et al 2012; Bauman 2013) was also demonstrated by the proximity of graduate roles in the large organisations. Whilst some individuals (P14) suggested that this proximity wasn't the reason that they were accepting a graduate role, others ruefully accepted this proximity was part of this: "it's such a great place to be .... such a great name to have behind you, it would be stupid not to" (P20). However, it could be suggested that this is true for graduates across the last few decades, with easy access and prior knowledge, helpful and comforting reasons to accept graduate roles.

There were however many indicators of stability with most WPS (P2,4,8,9,10,11,12,14, 18, 19, 20) still preferring the solidity of a graduate
scheme or working for their current organisation, rather than a protean/boundaryless career or liquid nomadism (DeFillippi and Arthur 1994; Bardhi et al. 2012; Hall et al. 2013). In fact, both P2 and P9 discussed working for their organisations for over 5 years post-graduation!

Furthermore, organisations clearly have liquid elements given their need for agility, flexibility which can result in continuous change (Bauman 2013). Organisation 2 recruited a large number of WPS ever year and so they knew that there wouldn't be graduate opportunities for all. Bauman (2013) suggests that progress is perpetual and therefore whilst these individuals want stability, their occasional liquid comments or their desire of career PO may be socially constructed, as society and organisations are developing these individuals for this liquidity. Therefore it could be suggested that this current generation of students and graduates are a product of what we want them to be. Society, universities and organisations are suggesting to individuals to become life-long learners, to understand their career will not be linear, so we are preparing these future generations for perpetual liquidity and therefore should we not be surprised when sometimes they want to take control of their own careers, as summed up by PDA1:

"maybe that's why it works that they chop and change so often, because actually when they're there, they have this real sense of ownership and everything as well, uh, but they still got there, their, their ownership of their own career goals" PDA1

Perhaps this is the new work-career reality for this group of individuals? They build job-related ownership quickly and whilst they are in the role, they have strong personal attachments, but just as we grow out of toys and move onto something else, they will do the same in the workplace. There may be some stronger-lasting attachments, like a favourite teddy bear that may be returned to, but these WPS have grown up in a time of an abundance of possessions and their job role is just their latest 'accessory'. Their one stable attachment may be to their career, because for those career-minded individuals it's the purest form of ownership. Further studies to ascertain the role of career ownership, alongside the role of other organisational targets, would be beneficial to build on this study's results.

#### 4.12.4 Organisational Psychological Ownership

This study has mainly considered job-related PO, because of the accessible nature of the job role as a PO target. Nevertheless, there is an additional key target often considered in the workplace by scholars, which relates to research question three, that of the organisation. Pierce et al. (2004); Mayhew et al. (2007) suggest that whilst the organisation is an important target to consider, because it's perceived as more abstract and ambiguous, individuals are less likely to feel able to control and manipulate it, thus lowering the chances of developing an attachment.

This study concurs with this viewpoint, with fewer WPS feeling organisational PO, in comparison to job-related PO. Yet, there were differences in degrees of strength of feelings with organisational PO occasionally nested in job-related PO. There were also some outliers which suggest that like job-related PO, the reasons for organisational PO are multifaceted and socially constructed.

If participants did feel organisational PO, those in smaller organisations generally developed these feelings more quickly. Bernhard and O'Driscoll (2011) suggest that PO emergence is more likely in small organisations, because the roles and the environment are less formalised, and individuals have closer proximity to leaders. P2's experience demonstrated this. He worked closely with the MD, was informally offered a graduate role exceedingly early on in his placement (within the first two months) and so a strong attachment developed quickly. He talked about working for the organisation for "years to come" (P2). In this instance, the organisation is seen in a less abstract form and is more accessible as a target, seemingly resulting in greater opportunities to invest self in the organisation. Both transactional and transformational leadership, along with ethical leadership, are positively related to organisational PO (Avey et al.2009; Bernhard and O'Driscoll 2011: Avey et al. 2012: Park et al.2013), and whilst this study did not cover leadership, in this instance, the role of the leader was crucial in the WPS developing an attachment to the organisation.

In another example, P17 felt a keen sense of belonging with his SME from around four months, partly from feeling that he had similar values to the

organisation, but also from learning more about the other individuals who worked in the organisation: "just learning where everyone came from, what the office was like before I was here" (P17). Supervisors or other team members can act as 'referents', influencing newcomers' relationships to organisational PO to be similar to their own (Sluss and Ashforth 2008: Sluss et al. 2012; Peng and Pierce 2015). In this instance, the sense of belonging motive was able to be fulfilled by the relationship he built with colleagues, allowing P17 to feel as if he was part of the history of the organisation. This study did not look at collective PO, so there is no way to ascertain if these feelings of attachment were felt by everyone, but it did suggest some form of social identity motive was present. Pierce and Jussila (2010) suggest that the social identity motive only underpins collective PO, but there may be instances, whereby those who have a strong need for social identity or self-congruity perceive this within an organisation, even if others do not. Dawkins et al. (2017) recommend further research relating to identity and this research would suggest it includes research relating to social identity.

The process of "giving ownership" also highlighted the role of organisational culture in expediating the development of organisational PO. P11's preferred placement was cancelled at the last minute, forcing her to take a role with a company that she was not attached to. Having been given significant job-related ownership, this also seemed to affect her attachment to the organisation over time. Previous research (O'Driscoll et al. 2006; Peng and Pierce 2015) suggests an individual's feelings about their job will influence how they feel about the organisation and therefore by enhancing job-related PO, individuals are more likely to feel organisational PO. This was partly demonstrated with some WPS exhibiting high levels of PO for both targets in SMEs (P2 and P17) and global organisation: *"I love the company....do you think its first job love?" (P11)*.

P10 however felt stronger ownership for the organisation, rather than for the job role, as she felt that whilst her team and supervisors might change over time, the company, which held similar values to her, would remain the same. She was the only WPS to show territorial feelings towards the organisation, discussing how she had forced her mother to shop at her

organisation, rather than shopping with a competitor: *"my mom shops at XXX and we just had an XXX open down the road and I'm…there's no excuse now"* (P11)

Nevertheless, strong job-related PO did not lead to organisational PO in all WPS. P4, who worked at a global tech company, suggested that *"it's hard to feel "we" at XXX, because it's a massive company"*. However, he did suggest feelings of attachment for parts of the organisation when working on smaller projects with a common goal. P15 gave similar feedback and his stance is particularly interesting, as family members, including his father, had previously worked for the organisation. He had turned down other placement opportunities to work in this company and seemed to demonstrate high levels of job PO. Yet his feelings about the organisation were more complex:

"(pause) yes and no...I feel ownership towards recruitment ...towards the people that I recruit, how many...if we hit our goals and deadlines. I don't feel ownership to how XXX perform, because ultimately, it's way too big a machine... Whereas if I think I worked for a smaller organisation that would be different because you're more...it doesn't matter about recruitment; it matters about the whole process." P15

These organisations both used onboarding rituals and gave their WPS opportunities to meet senior role models, yet organisational PO seemed to elude these WPS, because of the scale of the organisations, making this target feel distant and unmalleable. Both of these individuals exhibited high levels of job-related ownership, and were committed team members, but demonstrated far less ownership for the organisation, so it was interesting that transference did not occur, especially given the high levels of control in their job roles. Peng and Pierce (2015) suggested that job PO can mediate the relationship between organisational PO and experienced job control, but these examples suggest that this may not apply to all employees, and perhaps for organisational PO, the time spent within an organisation matters more than it does in relation to job-related PO.

One outlier was participant P9, who felt a strong attachment to his organisation prior to joining. He worked for a medium-sized local organisation which had an incredibly positive image amongst his age group and was an established brand in his home region: *"being Somerset born and bred, XXX always been sort of a big part. big part of growing up"* (P10). This resulted in him feeling

organisational PO before job-related PO. He was the only participant who indicated organisational PO helped facilitate job-related PO. As posited previously, it may indicate that certain brands may induce prior attachments (such as owning or using their products), thus providing an early building block to organisational PO. However, further work specifically relating to individuals working for well-known brands needs to be conducted.

Scholars have suggested that organisational PO can influence an individual's affective commitment (Mayhew et al. 2007; Liu et al 2012), organisational commitment (Van Dyne and Pierce 2004; Han et al. 2010), work engagement (Ramos et al 2014; Zhang 2020), organisational based self-esteem (Van Dyne and Pierce 2004; Liu et al. 2012; Zhang 2020), and job satisfaction (Avey et al. 2012; Liu et al. 2012; Peng and Pierce 2015; Zhang 2020), so there are many suggested positives outcomes to organisational PO for both the individual and the organisation. However, P4 and P15 also showed signs of most of these behaviours, even if they exhibited lower levels of organisational *attachment*. Therefore, whether the organisation should concern themselves with facilitating organisational attachment, if individuals are only in short-term roles, could be questioned, with the most likely answer being both 'yes' and 'no', depending on how they manage talent management.

Some of these organisations hire far more WPS than they hire graduates, and so in terms of supply and demand, an organisational attachment is not necessary, if it doesn't affect the quality of their work. Nonetheless, Employer 2 suggested that P15 was one of the strongest WPS that she had ever worked with, and therefore the organisation probably wants to keep their high performers. This example is also particularly interesting, as P15 worked for an organisation where ownership is embedded in the organisational culture and there are obvious signs of "giving ownership" to employees. The other WPS within the organisation, who were interviewed (P12, 14, 20), exhibited organisational PO and all were hoping to return to the organisation, but P15 did not feel as much of an attachment and was looking at other graduate options. Further research could consider how to take a job role attachment and build on it to ascertain a connection to larger targets, such as the organisation. For example, is it possible for the organisational culture and

supervisors to "give ownership" of the organisation in a comparable way to giving job-related ownership?"

One way may be via the building block process between task and organisational ownership mentioned below by employer 3:

"if you can identify the task to start with, because that...makes you feel needed, valued, which I think is a natural human reaction.. in any organisation... thereafter... we've certainly seen that they start to identify with the business and...understand their position within it over time...that's generally how it is until they leave." E3

This is a similar building block approach to that of job-related PO. However, E3 worked within an SME and reverts back to Bernhard and O'Driscoll (2011) suggestion about individuals having easier access to these organisational targets. If multi-national organisations wish their staff to feel organisational PO, additional building blocks linking job ownership to departmental attachment through to organisational attachment may be required. However, as stated above, an organisation may decide that job-related PO is ownership enough for their staff.

It was however possible for WPS to gain organisational PO in large companies, with P10, 12, 14, 19 and 20 all demonstrating this attachment. P19 neatly sums up the potential impact if organisations embed a culture where individuals feel like owners:

" the way we are treated as employees, it makes it feel like it's my business as well. So when I'm thinking of the outcomes, I'm thinking of the business...as well. So it's not just about me keeping my position, it's about me benefitting the company." P19

These feelings of attachment to the organisation, led to the desire, not only to do the best for themselves, but also do their best for the organisation, to essentially think like an owner. Many described themselves in terms of 'ambassadors' for the organisation, including P5, who understood he wasn't in the right role for him, but still felt responsibility towards the organisation:

"cause obviously, image and reputation you're, you're the face of the company at the end of the day. And although there are people behind you, you're the ones interacting with the customers. So yeah, I did feel like it was my job to represent XXX umm and it was our job to make sure that the business was continuing to grow." P5

Organisations can build a sense of belonging with individuals at the start of their career and show them how to act as ambassadors for the organisation and to take responsibility. Most of these individuals were keen to do this (P1,2,3,7,8,9,10,11,12,13,14, 16, 17,18, 19, 20) and even those such as P4 and P15, who showed less attachment to the organisation, were still acting as owners.

As mentioned in the "giving ownership" section, employers 2 and 3 highlighted how they tried to expand job role ownership to consider other organisational targets, such as the team or organisation. This was partly about helping WPS understand the bigger picture and the impact of their work on the business, but also helping them feel that sense of belonging towards the business and their team:

"Showing how those people and what they do and the actions that they have, how they aid the progress of, not just tasks, but also the progress of the business...helping them feel important and feel part of it" E3

As well as ownership being informally embedded via supervisors and teams, there were also some formal activities which cemented the company ownership ethos. For example, P19 was given or was permissioned to find missions or projects which created benefits to the organisation. His final placement presentation about his achievements was linked to opportunities for progression onto the graduate programme, and thus benefitted the WPS to take ownership of missions.

Both E2's and E3's organisation used the appraisal process for formalising ownership. E3's organisation bought in external HR people to manage the process, using it as a means of showing they valued the individual's contribution, as well as providing constructive feedback. E2's organisation asked WPS to demonstrate quarterly documentary evidence of their achievements, which E2 believed encouraged ownership in these inexperienced employees.

To conclude, some WPS were able to feel organisational PO in addition to jobrelated PO, which, given the brief time period that they worked in the organisation, demonstrates that early attachment is possible and this was particularly true in SME's. Nonetheless, some WPS with strong job-related PO found it harder to bind themselves to larger organisations to develop organisational PO, although it wasn't entirely clear what the reasons were. Once again, PO is more nuanced than previously advocated and, whilst studies have shown that job-related PO is not always a predictor of organisational PO (Pierce et al. 2004; Mayhew et al. 2007), especially in multinational organisations, further qualitative and quantitative studies would be beneficial. Nevertheless, there may also be a new form of PO which could impact on these attachments, which will be covered in the coming section.

#### 4.13 Bringing the Themes Together: Discussion

This study sought to explore the development of PO in individuals at the start of their career using a sample of work placement students from a post 1992 university. This section joins together the four key themes identified in this chapter.

Early research by Pierce and colleagues (2001, 2003; 2004) suggested that individuals can become attached to certain workplace targets via a process involving PO motives, routes, values and target attributes which this study concurs with. There have been a significant number of studies since this point, yet most have tended to overlook this complexity in order to address other PO elements. This qualitative study has returned to the initial conceptualisation to further our understanding of the intricacies of PO development using early career professionals on a year-long work placement as the basis for this study. PO studies have traditionally used experienced professionals for studies and the researcher believes this study is unusual in highlighting the experiences of participants who are both new to the workplace and in a contract role.

This study has highlighted how early participants may develop ownership of tasks with the majority of WPS providing examples of task-related ownership within their first three months. At this point, some WPS also demonstrated job-related PO, although both employers and PDAs interviewed concurred that full job-related PO often took between three and six months.

A few individuals showed ownership of their tasks from day one (even before ownership had been permissioned) with employer 3, highlighting how a new placement student suggested the inclusion of a new slide on her first day:

"On her first day, she took an ownership of that. She didn't just do what I told her to do, she took ownership of it. She received it and she said, okay, this is mine now, I've got a framework but if I've got an idea, I'm going to put it in there". Employer 3

It is therefore plausible, that in the same way that we attach ourselves quickly to favourite possessions even if we know they will not last a lifetime (a new car that is changed every few years for example), so ECPs may have strong attachments to intangible targets if there is self-congruity and/or if they have prior ownership experiences (in education, a part time job etc). The attachment might be fleeting or long lasting, but there may be a strength of feelings.

The strong theme of "giving" ownership emerged demonstrating a dynamic iterative, relational process initially permissioned for small tasks, then utilising a building block approach to larger, more intricate tasks. In this study, this was the most successful form of creating early ownership. When supervisors and the organisational culture permissioned a WPS with ownership over tasks this facilitated the development of competence, confidence, trust and a professional identity resulting in eventual "taking" ownership of their job role. This is far earlier than originally suggested (Pierce et al. 2003) and demonstrates to employers that it is possible for those at the start of their career or in contract roles to form early attachments to their job role if provided with immediate ownership opportunities. Also unexpectedly, given these individuals were working in a "contract" type role, that this was a short-term role seemed to make no difference to the WPS ability to attach themselves to workplace targets.

Self and material targets are closely entwinned (Belk 1988; Ferraro et al. 2011) and this is also true of immaterial targets such as work-related targets. Moreover, it was the social construction of "giving" ownership which most frequently facilitated a speedier development of job-related PO and the ECPs emerging professional identities. The interweaving of ingredients such as previous ownership experience, opportunities to fulfil motives, access to PO routes, the permissioning and trusting of ownership via the supervisor, organisational culture, target attributes, job crafting, rituals, values/personality traits (Pierce et al. 2003) as well as context all seem to play a part in PO development highlighting both its complexity and interplay between the self and other individuals.

Expected ownership was signalled by some organisations during the recruitment process allowing WPS to understand the level of responsibility and accountability that they might expect whilst in a work placement. Induction and handover periods were often full of rituals that underpin this process of "giving" ownership. Handover books were identity leaden and some organisations skilfully bonded the WPS and the organisation through team activities and rituals relating to the organisation.

It was most frequently the supervisor who facilitated an organisational culture of "giving" ownership. E3 suggested how he managed the "giving and taking" ownership process with WPS:

> "it's accepting that it's on their desk now, they've got to look after it, they have to make it better. Yes, they have to stay within the lines, but it's theirs, they don't necessarily have to refer back to somebody for everything they do." E3

E2 concurred, suggesting she clearly denotes a handover of tasks *"it's very much, off you go XXX"* (E2) thus transferring ownership. P19's organisation facilitated ownership transfer via "missions" which were either given to the WPS or they could suggest their own missions. Check-ins with supervisors were available if requested, but essentially the WPS were required to *"drive your own success."* (P19).

Those organisations signalling a strong ownership culture seemed to embed a controlled "sink or swim" mentality whereby individuals were supported to take control of their job role and opportunities within the organisation. This facilitated early opportunities to invest psychic energy into their job role/projects by crafting, controlling, caring and mastering (Sartre 1956; Csikszentmihalyi and Rochberg-Halton 1981; Belk 1988; Ahuvia 2005; Mittal 2006; Watkins et al. 2015) building block targets entwinning self and job role until the target became "theirs". WPS who felt trusted by their supervisor's suggested greater confidence in their ability and skills. This study concurs with Olckers and Enslin (2016a) regarding the important role that trust plays within PO, although whether this is as part of psychological safety as Zhang (2020) suggests or independent of other safety components requires further research. For early career professionals, trust seems to play a role in supervisor validation.

If ownership was permissioned by the supervisor, the WPS then had the opportunity to accept or reject ownership. Whilst it was rare for individuals to outright reject ownership, there was a strength continuum whereby some individuals accepted, took and developed job- related ownership to a greater degree than others. P1,2, 4, 10, 11, 12, 15 and 19 were in positions where ownership was permissioned, and they grabbed it. P5, 14, 17 and 20 were given ownership, but did not demonstrate the same strength of ownership as the first group. P6, 7, 8, 9, 16 and 18 worked in organisations where ownership did not seem to be embedded in the organisational culture and permissioning was sporadic as was the strength of their ownership. As discussed in the trust section, P3 and P13 were often not fully permissioned ownership and the "giving" of ownership seemingly became contaminated. When ownership is given reluctantly or with caveats, this studies results would suggest that it hampers the ownership process with the receiver then demonstrating less ownership and less trust in the giver. Again this is likely to mirror our relationship with possessions, whereby sometimes, offered ownership comes with caveats which holds back the receiver from that crucial self investment.

Pierce and colleagues (2001, 2003; 2004) suggest that personality and values are likely to also play a role in PO development and whilst this was not the focus of the study, this may be particularly true of those individuals who take

ownership. In this study, self-efficacy seemed to be an important key factor to initiate taking ownership as stressed by both employers and PDAs:

"people that are confidence enough will take ownership...You get your people that don't have the confidence...they won't volunteer, because they're scared, they're gonna mess it up...won't touch it with a barge pole...that's important, isn't it? Yeah, confidence that ownership is taken, sometimes. And it's also given, it works both ways." PDA2

Previous work placement studies also suggest the importance of self efficacy and how a placement may facilitate its development ((Edwards 2014b; Inceoglu et al. 2019; Mele et al. 2021). Employer two suggested that recognition from supervisors and other team members developed confidence and from that point individuals were more open to suggesting new ideas, job crafting as part of taking ownership *"So that obviously builds their confidence"* (E2). This again points towards the social construction of ownership with relational aspects building WPS self-efficacy. Previous studies (Van Dyne and Pierce 2004; Liu et al. 2012; Brown et al. 2014b; Morewedge 2021) have ascertained the importance of some PO routes to develop ownership feelings and this study did see examples of all previously posited PO routes (control, self investment, self congruity and psychological safety, although fewer examples of intimate knowledge perhaps due to the time scales).

"Taking ownership" often involved WPS crafting their job role and in some instances moving the relational boundaries of the tasks or networks. Job crafting therefore seems to play a role facilitating PO development providing a "bottom up" approach of worker control and self investment rather than just the top-down process of job design highlighted by Pierce et al. (2009a); Peng and Pierce (2015). Good job design will contribute to opportunities for ownership, provide job-related rituals and allows organisations to facilitate PO through the suggested routes, nonetheless opportunities for forms of job crafting may be more important.

When ownership was consistently given to WPS, "taking ownership" became embedded in the mindset in some WPS and especially if the individual felt they could improve on a task or process or had a greater level of expertise than other members of the team/ department. P17 felt that his strong excel knowledge provided him with an opportunity to take on tasks to automate the organisations systems. He suggested *"it feels like I've become part of the office's history almost"* indicating that roles and not just possessions have biographies or social lives (Lastovicka and Fernandez 2005). P17 was the WPS who most frequently mentioned the importance of belonging and demonstrates how "taking" ownership even when not given can have a significant impact on an individual's feelings of being part of something. Ownership may play out as a narrative through someone's life and if they believe this relates to their self-identity, this may be maintained via the selfcongruity route (Morewedge 2021).

Some individuals had previous ownership experience, and this was another form of transference that occurred in the development of PO. Whilst this was found in several WPS it may partly explain why those at the start of their career were able to "take" ownership if it wasn't given. P7 was a team leader in a part time job role and suggested that she was used to controlling workloads and taking responsibility:

"I was team leader and I definitely felt a lot of ownership. I had a lot to look after, um, I had task to—actually I had targets to complete, and I was basically in-charge of what I had to do, um, so I felt a lot of ownership and I had a lot of responsibility and that's a lot more than I have at XXX" P7

Nevertheless, P7 felt that she wasn't given as much responsibility in her role compared to some other WPS which may have been why she sought opportunities such as her organisation providing a formal placement scheme. When asked what she would change about her work placement she suggested *"probably the amount of responsibility that I'm given" (P7).* 

However historical examples of ownership in a part time roles did not always lead to job-related PO on placement such as P3. Given the poor relationship with her supervisor where ownership wasn't permissioned, it does demonstrate that not only can PO be contextual, if ownership is not "given" then ownership opportunities may be blocked. If the supervisor doesn't permission ownership, it could be anxiety inducing for the WPS . P3 who struggled with her supervisor, felt she showed ownership by reminding the team of key deadlines and email requests *"it's not me being bossy, if I tell you that you've missed something or forgotten something"* (P3). Whilst this may seem like a small form of ownership, this quote is an example of someone who has lost confidence trying to justify this act of control and self investment. Whilst more experienced professionals might be able to navigate around this, this may be challenging for others, particularly those inexperienced regarding "how work works".

As suggested by Van Dyne and Pierce (2004); Mayhew et al. (2007) feelings of job-related PO do not always build feelings of organisational PO and this study concurs. However, that simplifies a complex relationship with a number of factors playing a part. Whilst organisational PO was developed by WPS working in SME's in line with the research of Bernhard and O'Driscoll (2011), the organisation as a PO target in large organisations was less easy to manipulate and felt more distant. Nevertheless lack of organisational PO did not necessarily mean that WPS didn't feel a sense of affective or organisational commitment, rather, those feelings were just not as strong as they were for such a malleable target such as the job role.

Nevertheless in some instances, WPS were still very positive about their organisation with three of the four WPS working for international employer 2, feeling some level of organisational PO:

# *"I definitely identify with the company... it's a place that, not call home, but a good work environment that I'm happy to be part of."* P20

Whilst the job was the key target for most WPS, P9 and P10 felt comparatively strong organisational PO. P9 was particularly unusual as organisational PO developed first. Whilst he felt totally committed to the organisation and saw himself working there for many years post-graduation, he did not seem to show any job crafting behaviours, nor a desire to invest himself in the role:

"For me, personally, I haven't...So I'm just absorbing as much as I can. I haven't gotten feedback on how I feel the systems can do any better, because I haven't done anything like this before" P9 This experience was an anomaly. A tentative suggestion may be that when an individual identifies so strongly with an organisation, the job role becomes secondary, but further research in this area would be interesting.

Feeling PO to the job role can provide great pleasure as well as pain. There is a duality of outcomes relating to PO which organisations and individuals need to be aware of. There are the many posited positive PO outcomes which were also observed in this study including job satisfaction, self-esteem, affective commitment, in-role performance, pro-active work behaviour and job engagement (Van Dyne and Pierce 2004; Mayhew et al 2007; Avey et al. 2009: Avey et al.2012; Knapp et al.2014; Peng and Pierce 2015; Wang et al. 2018; Zhang 2020) all of which makes PO so beneficial for both individuals and their organisations.

Pierce et al. (2001) suggests that PO is pleasure producing and the intensity of feelings suggested by the WPS demonstrated the many highs that are achieved when ownership is taken by an ECP. It is worth remembering those highs are partly a result of socially sanctioned praise and it was the circle of validation that helped WPS develop greater confidence and pride in their work.

Taking responsibility for their job role or a project often resulted in feelings of excitement and exhilaration linking to both self efficacy and self-identity. By investing self and entwining control, care or other forms of psychic energy, they were able to become "one" with the target and so the pride felt when successfully managing something complex was so much the greater. These are new, inexperienced employers who experienced great personal satisfaction and joy when success first came.

This pleasure also built confidence which resulted in the WPS sharing more ideas and taking on more responsibility. This is a win-win for individuals and organisations who both gain something when an employee is able to add value. By feeling responsible for targets some individuals also felt a sense of achievement and purpose.

Nonetheless, this sense of responsibility can have negative implications with the "weight of PO" also mentioned and it did seem to result in high stress levels in some instances. There seemed a genuine concern of letting people down and this combined with a fear of failure meant that some of these WPS

internalised stress rather than talking to their line managers. For those WPS who seemed to show high levels of ownership, their lives were consumed by this ownership demonstrating that for some whilst PO can have many positive implications, for others, this desire to possess can be all consuming and take over an individual's life if not properly managed. Given the move to working from home, this may be less observable to line managers and so organisations may need to identify early career professionals who demonstrate high levels of ownership to ensure that this shadow side does not become all consuming.

P15 sums up how job PO possessed him:

"This is something that I've really had to battle with...I think I always want to do the best...it means that I sometimes did work too hard and I was over-tired and it got to the point where my parents even got involved and said, you really need to just manage your workload" P15

Whilst P15 was an extreme example, he was not alone in feeling the weight of PO (P4, 10, 12 and 20) and often those who felt the most pleasure, also felt the most pain. The weight of responsibility can be initially positive because it can feel empowering, but there is a tipping point when this responsibility became too heavy. Whilst the pain side of PO impacted on individuals more than organisations, it feels likely that this could eventually become stress related if not managed and so it is in everyone's best interest to manage the dichotomy of PO.

There was also examples of ephemerality or liquid ownership (Fleura Bardhi et al. 2012) which has been shown to relate to tangible possessions. Work placements are temporal which suggest that WPS attachments would be weaker, although the results from this study suggest otherwise. Counterintuitively whilst attachments could be strong, they could also be fleeting, with some participants strategically managing their career desires ahead of their cherished work placement. It is unclear if this is liquidity in action or the strength of a PO target which is close to the WPS heart (career PO), but the PDAs all recognised this contradiction as one that they now see when meeting WPS. The disinvestment of job roles and organisations to allow these participants to return to their studies highlights the biographical nature of PO being both part of the participant's memoirs, but also becoming part of the life history of a particular job role, team, department or organisation. Individuals do not just disappear from these life accounts, but live on through their actions, other people's memories and their shared activities. Disinvestment could be painful, but in many instances, those WPS who were given ownership, now gave ownership to others. Where ownership is embedded, ownership lives on through its rituals, it's essence and individuals desire to find something that is part of their soul. Moving from being permissioned ownership to "giving ownership" to others seemed to help the disinvestment process, although this is only a tentative observation.

This group of participants have grown up in a world where they are told to take control of their career as there are "no jobs for life" and this socially constructed rhetoric plays a role in how they view work related targets. Whilst they had strong attachments to their job role and for some (especially in SME's) to the organisation, these ECP feelings could also be fleeting with an element of liquidity towards job role and organisational targets. Employers in the future may expect to see forms of ownership, which are strong, but ephemeral, characterised by affective emotions that seem solid, but can quickly become fluid. This may also partly be due to a new proposed work place target, career ownership. For some this was ultimate work-related target because it is part of the essence of themselves and therefore the most accessible and controllable, allowing stability and permanence in an everchanging world. Career ownership is both rational, career planning, as well as an affective feeling of "mine" which means that it can be less about meaningful work than a Protean career and for some, the desire for stability suggests differences to a Boundaryless career.

This desire for stability amongst the participants was demonstrated when considering their next career steps, with the WPS still generally preferring access to a graduate scheme (often with their placement provider) as their first step post-graduation. Narratives around graduate schemes play strong to this group of participants with societal influences such as family, universities and

organisational role models extolling the many benefits of this route. Liquid ownership and career PO are socially constructed for a generation who have been told there are no jobs for life. This creates tension between notions of stability within an organisation against the strategy of driving their own career success. It is no wonder that to the outside world they show deep attachments yet a desire to quickly move on, they are a product of societies' mixed messages. This along with the pandemic's impact on the way we work, may move career PO and liquidity to the forefront of these young professional's minds. Nonetheless career PO needs further research to ascertain if it can be called a PO target although it is potentially a really exciting area for future research.

Previous PO research has not considered how ownership is transferred to individuals and this study is the first to consider that PO may be "given" to individuals and/or "taken" by individuals. As these participants are early career professionals, giving and taking ownership may be more relevant at this particular work life stage, however given how much time and money are invested in talent pipelines it may be fruitful for organisations and supervisors to understand how job-related PO may develop at other levels. It should not be assumed that more experienced workers would naturally take ownership nor that inexperienced employees may not. Whilst organisations may suggest that developing ownership in individuals who are in temporary roles is not a priority, this process of "giving ownership" is transferable to all individuals at the start of their career. As this research indicates, some organisations already weave ownership into their organisational culture and whilst there are also negative implications (discussed in section 4.11.2) there can also be many benefits for both organisation and individual.

By exploring the findings from this study, a new relational model of job-related PO development in Early Career Professionals (figure 5) has been produced demonstrating the key elements involved in the process. But first a reminder of the original PO model created by the author to signify current PO research which was shown in section 2.15 (figure 5):

Psychological Ownership Overview



Figure 5: Psychological Ownership Overview

The new relational model of job-related PO development in Early Career Professionals demonstrates the key elements involved in the giving and taking ownership process. This study concurs with Pierce et al (2003) who suggested not all motives and routes were required to activate PO, nonetheless what is important is the permissioning of ownership (usually by the supervisor) and an ownership culture which allows the taking of ownership via the most pertinent routes. Whilst this new ownership model is based on early career professionals there are many aspects of this model such as rituals, opportunities for job crafting and trust that organisations may wish to consider as part of "giving and taking" organisational culture. In fact, the "giving" of ownership becomes a ritual in itself if it is continuously embedded in an organisational culture. In a similar form to other models, this doesn't truly demonstrate the complexity, dynamic and iterative nature of this process that we find in all human interaction, rather an indication of core ingredients and a basic recipe.



#### A Relational Model of Job-Related Psychological Ownership Development in Early Career Professionals

Figure 6: Job-related PO Development in Early Career Professionals

The broken lines indicate where there may be a relationship, although the strength and direction of this relationship are likely to vary. Figure 7 below, then shows the key differences between the two models in red denoting either PO development additions or where there was little reference to them from participants in this study. That is not to say that they are not part of PO, rather that they were not demonstrated, or were less significant in this study.



A Relational Model of Job-Related Psychological Ownership Development in Early Career Professionals

Figure 7. Comparison of PO Development in Early Career Professionals

This chapter has presented the findings from this study indicating a relational aspect of PO which has been missing from previous studies. By concentrating on a group of early career professionals there has been the opportunity to ascertain PO development at the start of a career and has resulted in a new relational PO model demonstrating job-related PO development in early career professionals.

### **Chapter 5 Conclusion**

This chapter draws together the final conclusions and recommendations relating to PO development in early career professionals. This study was undertaken due to the researcher's previous experiences working with both early career professionals and students in the classroom, leading her to believe that PO research to date had neglected a key time period when PO can be shaped and developed.

Firstly, this chapter will address the completion of the study objectives, before demonstrating the study's contribution to knowledge and key recommendations to stakeholders. This chapter will conclude with future research directions and a final reflection from the researcher.

This study aimed to explore the development of work-related PO in individuals at the start of their career, using a sample of work placement students from a post-1992 university via four key objectives:

# 5.1 Objective one: To explore the formation of work-related psychological ownership in individuals at the start of their career.

To address the first objective, qualitative research was undertaken.

That individuals can develop PO for job-related targets has already been established. However, previous organisational PO research had not considered *how* ownership is transferred between individuals. This study is the first to consider the manner in which PO is socially constructed in the workplace, allowing individuals to develop job-related PO.

Through semi-structured interviews with WPS, employers and PDAs, the author confirmed that job-related PO was the most accessible form of workrelated attachment and thus most of the research in this study is concentrated in this area.

The study found that the majority of WPS in this study were able to demonstrate **job-related PO** during their work placement. Both the employers

and PDAs interviewed concurred, with most participants providing examples of **task-related PO** within three months. This is earlier than previous research had suggested (Pierce et al. 2001,2003,2004), demonstrating that early attachments can be developed and nurtured. There was however a myriad of differences in the WPS experiences, relating to timings, how ownership was developed and perceived strength of feelings, which had not previously been demonstrated.

The conceptual model developed from this research demonstrates how jobrelated ownership is "given" and "taken" between WPS and their supervisors or team members. Section 4.13 provides the relational model of the development of job-related PO in early career professionals. Supervisors permissioned ownership through utilising a building block approach, initially via small tasks, building up to the entire job role. This sounds a relatively simple activity, nevertheless this is a socially constructed process often requiring an organisational culture where ownership is ritually embedded and a supervisor who sanctions "giving" ownership demonstrating trust in their incumbent. Organisational culture had not been considered in the context of previous PO research, nor the role of the immediate manager, although there is some research relating to transformational, authentic and ethical leadership (Avey et al. 2012; Alok 2014; Park et al. 2015).

WPS were then able to "take" job-related ownership via the posited routes of control, psychological safety, self-congruity and self-investment (Pierce et al. 2001, 2003; Pierce et al. 2004; Zhang 2020; Morewedge et al. 2021) and through a suggested new route of 'job crafting'. Previous studies have highlighted the top-down influence of job design (Pierce et al. 2009b; He and Pierce 2015), rather than considering the bottom-up approach of job crafting.

Attachments to work targets could be instantaneous if the target was congruent with the individual but was strengthened by the constant reenactment of crafting and controlling entire processes or job roles aided by a circle of validation from supervisors and team members facilitating professional identities. There were differences in how much ownership the WPS "took" reminding us that there will be different degrees of workplace ownership taken.

Some WPS "took" ownership without prior permissioning, having experienced ownership previously in part-time job roles or education. Nevertheless, it was more likely that WPS required initial agreement before feeling confident to suggest new ideas or job craft to a significant level.

This study was taken with a backdrop of liminality as the student participants moved from student to professional identities. Attachments to the job role seem to hasten border-hoping to become a professional and a tentative conclusion would be that job-related PO can aid liminal processes in early career professionals. Liminality is also dynamic and intricate, requiring an iterative process of identity development. "Taking" PO may help these ECPs "try on" different identities as they job craft to find the elements of the role that fit.

Nevertheless, conclusions were reached that not all individuals developed jobrelated PO and it was found to be weak and sporadic in WPS where the culture of the organisation did not encourage ownership, the supervisor blocked ownership development (consciously or unconsciously), or if the WPS felt a lack of trust, perceived organisational support or organisational justice. WPS confidence was also important.

Whilst most of the study relates to the job role, this study concurs with previous studies (Pierce et al. 2004; Mayhew et al. 2007) which suggest that attachment to the organisation is not necessarily nested within the job role. Those working at SMEs were more likely to feel a stronger organisational attachment than those working in larger organisations where the organisation is more abstract. Finally, a new workplace target was identified ('career ownership') which will be discussed further in objective three.

This study fulfils objective one by demonstrating that job-related PO develops via a dynamic, relational, iterative, complex process, in which employees both "give" and "take" ownership. The ownership process may be emotion-laden with previous experiences, personal feelings and values from both giver and taker eliciting a significant level of complexity. This is a social exchange which can be fraught with sentiment that can allow the transfer of ownership to become confused. Yet, when it is managed successfully and expectations are clear, there are benefits to both the individual and organisation.

#### 5.2 Objective two: To uncover the positive and negative outcomes suggested from the participant's experience of job-related psychological ownership

Objective two was also completed via semi-structured interviews with WPS, employers and PDAs.

This study saw examples of the pleasurable side of PO highlighted in section 4.11.1. The strength of feelings that ownership could provide was akin to first love at times, providing great happiness, pride and confidence instilled by external validation. As suggested by other scholars (Van Dyne and Pierce 2004; Mayhew et al.2007; Avey et al.2009; Avey et al. 2012; Knapp et al. 2014; Peng and Pierce 2015; Wang et al. 2018; Zhang 2020; Henssen and Koiranen 2021), PO has positive individual and organisational benefits which this study concurs with.

Nonetheless, this study demonstrates the dichotomy of PO for individuals, with job-related attachments providing great pleasure, but also pain. Whilst individuals who felt high levels of ownership gained great job satisfaction, sometimes a sense of purpose with their work achievements stretching beyond expectations of WPS, they also experienced the greatest "weight of ownership". These individuals often had high expectations of themselves as well as being aware of the esteem felt by the team. This creates pressure and because these WPSs had little experience of the workplace and seemed less able to raise fears of being overwhelmed with supervisors, it resulted in them carrying the weight of PO on their shoulders. This was particularly true for those individuals who developed strong feelings whereby the intensity of PO caused them simultaneously to feel great pleasure and pain.

Whilst topics related to stress and mental health have gained prominence in the workplace, those new to the workplace or in contract roles / short-term assignments may feel obliged to hide the weight of ownership due to the precarious nature of their position or fear of letting others down.

In conclusion, objective two has been fulfilled by demonstrating the duality of PO in early career professionals, where the pleasure and pain of PO can be

interrelated. The completion of objectives one and two are the basis for the relational model of job-related PO development in ECPs.

### 5.3 Objective three: To identify areas where qualitative research can add value to Psychological Ownership studies

As previously alluded to, PO research has mainly been conducted from a positivist perspective using the survey method to ascertain if individuals have PO for organisational targets such as their job role. This has helped develop our understanding of PO and in particular PO antecedents.

Nonetheless, this **qualitative study** has provided a very different perspective of PO highlighting the relational experience whereby PO can be developed by people and *between* people. A strong social element was found in this study, demonstrating the influence of line managers, senior managers and some team members. This could be in many forms from role modelling, organisational rituals, but most frequently via the permissioning of ownership ("giving ownership"). Whilst this study's population was individuals at the start of their career and thus is situated in a particular context, the process of giving and taking ownership may be demonstrated at most work levels, highlighting the importance of an organisational culture that permissions ownership. Whilst some individuals may have the confidence, experience and desire to take ownership, this study does explain the challenges faced by individuals in organisations where ownership is not the norm. If line managers or the organisation "lend" ownership, do not permission the taking of ownership, contaminate the ownership relationship, or actively hold on to ownership targets themselves (consciously or unconsciously), it appears difficult for an incumbent to take ownership, especially at the start of their career.

This qualitative study provides some indication of the socially constructed nature of PO and the intricacies involved in such a dynamic process. PO motives, PO conditions and the giving and taking of ownership are interwoven, with each experience being different because of the context and human relations. Qualitative research paints a picture of this complexity and the interactive relationships, which is not possible with quantitative research. Silverman (2011) suggests that research cannot mirror the social world, but we can observe people bringing meaning to their world. Whilst previous PO studies have determined some of the positive and negative outcomes of PO, statistics cannot tell the lived experiences of PO and in particular the weight of PO. Qualitative research captures these experiences and emotions, making it easier to communicate the pain sometimes felt by individuals.

Whilst this study could never be truly inductive because of the researcher's prior knowledge of PO, the open-ended questions used in this study provided new areas to investigate. Job crafting has had limited recognition in previous PO studies, yet seems to be an important component when "taking" ownership. The more opportunities for the different forms of job crafting, the greater the number of opportunities for control and self-investment.

Open ended questions also demonstrated the benefits of this qualitative study by highlighting a possible new work-related target, that of **career PO**. An open-ended question such as, "My study is about psychological ownership. What does ownership in the workplace mean to you?" elicited unexpected responses. If we consider James (1890) quote, *"A man's Self is the sum total of all that he can call his"*, aside from the gender inequity of the statement, PO and our possessions is about personal choice. For those who feel strongly about their career, this attachment became stronger than the attachment to the job role. Some participants loved their job, but already had a career plan and this became their priority. Whilst further research is required to consider this potential PO target, section 4.12.2 highlights that whilst it may be closely aligned to boundaryless and protean careers, career PO is conceptually, a different construct. Importantly, this construct may not have been considered without an inductive mindset and demonstrates the lovely surprises that using a different method may bring.

Nevertheless, one of the challenges of this study was the inability to compare "apples with apples" at times in the literature review or results section, because PO for this study was viewed from a different lens. This made the results section challenging at times, akin to reading directions when your road map only documents certain roads. On the other hand, qualitative research in this instance did reveal many interesting new facets, allowing future researchers

the opportunity to consider the more relational aspects of PO, which feels essential to bring diversity to our understanding of PO.

#### 5.4 Objective four: To provide a framework for how PO may be utilised and developed by organisations and their key stakeholders (e.g., line managers, HR).

Objective four is answered in the "giving ownership" workplace framework below and links to the recommendations for employers, ECPs and education providers which can be found in section 5.5. The COLTOS "Giving Ownership" framework acts as a reminder of the key elements relating to the "giving ownership" process:

**COLTOS "Giving Ownership" Framework** 

Control, self investment and job-crafting opportunities

Organisational culture facilitating ownership

Line manager "giving" ownership

Trust and other forms of psychological safety

**Opportunities for self-congruity** 

#### Stress management systems in place

Organisations may need to ascertain to what extent they have an ownership work culture and/or if it is something that they wish to develop. Ownership cultures may be significantly different in some organisational cultures, but if there is a desire for staff to feel job-related PO, organisations need to create a "giving ownership" culture at all levels within the organisation.

Whilst the new framework does not include organisational PO, by building jobrelated PO in an organisation, fulfilling the PO motives and providing social referents, there are opportunities to simultaneously strengthen feelings of organisational PO. This is particularly true in an ownership organisational culture which encourages individuals to take ownership in a psychologically safe environment.

#### 5.5 Recommendations

The following recommendations link to the COLTOS "Giving ownership" framework, but also provide some broader suggestions for education providers, early career professionals, employers and researchers.

#### 5.5.1 Educational Providers

Recommendations relating to early career professionals and PO from an education perspective are as follows:

Educate students about psychological ownership, including its positive and negative implications, to help them gain a sense of how ownership may affect their lives.

The COLTOS model may also be applicable to Programme Leaders and Unit Leaders to provide a framework or starting point for those starting university. Given that many Business Schools encourage group work, it may be helpful for students to understand how to "give" and "take" ownership in a similar manner to the workplace, helping them understand the relational nuances of group work. PO could then be reintroduced with second-year students to consider ownership in the workplace ahead of any work placements or internships.

Education providers may wish to help students understand the terminology within job descriptions to ascertain expected ownership levels. Decoding an organisational culture can be challenging for the inexperienced and so practical advice may help students to read between the lines.

Whilst universities have directed more attention to the transition onto placement and back to university than in the past, more attention could be emphasised regarding liminality and being "betwixt and between" identities (Turner 2011) before and after the placement. In particular, the ownership many of the WPS developed on placementcould then be taken back to the final year of studies. Given that Universities emphasise the professional persona, do they do enough to encourage this new identity post placement or are students immediately plunged back into a student identity? Emphasising

how students can take control and investment in their Research Project may help returning students make the connections between different forms (work or study) of "taking" ownership.

Finally, those staff who visit students on placement should be aware of the weight of PO and recognise those WPS who may be susceptible. High ownership feelings can be managed and some early awareness may be beneficial to all.

### 5.5.2 Students/Early Career Professionals

Recommendations relating to students and other early career professionals and PO are as follows:

Students and ECPs may wish to reflect on their levels of self-efficacy and need for psychological safety in their first job role. Not all individuals are well suited to a "sink or swim mentality" when they first enter the workplace, however well supported this practice may be in an organisation. Therefore, ECPs may wish to consider "organisational fit" when job hunting. Individuals could consider how important ownership is to them when looking for a placement / internship / apprenticeship / graduate job role. Those individuals who feel they may develop high levels of ownership should look for organisations signalling responsibility within job roles.

ECPs are encouraged to take ownership to understand signals relating to their own well-being and take positive stress management actions such as talking to their line manager or placement tutor. If students understand the burden of ownership, they are more likely to signal the challenges faced.

ECPs may wish to use the building block approach from tasks to the job role to build confidence. They should then consider forms of job crafting to offer them means of investing self, remembering that there may be boundaries that they need to stay within. Building PO early in education or part-time work will also help the transition to the workplace.

#### 5.5.3 Employers

Recommendations to employers mainly relate to the COLTOS framework. Nevertheless, there are some further recommendations relating to early career professionals (ECP) and PO which may be particularly pertinent for those in Senior Leadership positions and HR:

Organisations should follow the COLTOS "Giving Ownership" framework as a means of embedding ownership. A building block process within departments is likely to be the most successful initial form of developing ownership, starting with Departmental Heads "giving ownership" to staff they line manage and working downwards to other levels of management. Whilst this is a study about early career professionals, it may be relatively challenging for an individual line manager to give ownership in isolation and therefore embedding departmental ownership is most likely to develop the required organisational culture. This may take time depending on the organisation. If "giving ownership" becomes part of organisational rituals, such as the recruitment strategy, onboarding process, annual appraisals, there is a greater chance of being embedded within the organisation more quickly. Ownership is likely to be entrenched in an organisational culture if it is an organisational value with a 'golden thread' from recruitment process through to disinvestment processes. Rituals relating to ECPs, such as "handover books" between placement students, interns and apprentices, may provide peer to peer signalling which can also be important, but other forms of ritual signalling relating to organisational referents (Sluss et al. 2012) would also be beneficial.

Line managers are key to the "giving ownership" framework and the reality is that not all line managers naturally "give ownership". Support is suggested via workshops to help line managers understand the "giving ownership" process to ensure ownership is clearly offered and permissioned. Trust and a supported "sink or swim" attitude allow the individual to learn from mistakes. As jobrelated PO is a socially constructed process, managers should not underestimate the intricacies it involves and should consider if they openly and trustingly give ownership or if historical experiences, such as borrowing ownership or contaminating ownership, could inadvertently play a part. Secondly, role model exemplars are important to signpost how they permission ownership and trust. This can help other managers understand how to give ownership clearly and unconditionally in the workplace.

As previously mentioned, Line managers should provide building blocks for developing job-related PO for ECPs, starting with small tasks and building up to the whole job role. There should be opportunities for controlling the job role in particular and using the many forms of job crafting. Job crafting also helps to develop self-congruity by providing opportunities to alter the role or participate in projects that link to their identity. Self-congruity may also be important for moving between liminal boundaries and line managers play an important role in praising and building confidence by validating the ECPs' professional identity. In addition, if line managers signpost how building task ownership leads to individuals gaining future job-related PO, it may also help new employees see a mental "ownership road map".

Ensuring psychological safety of staff members and in particular developing stress management processes for early career professionals is particularly important. Fear of failure or letting down line managers may mean that ECPs do not indicate feelings of stress whilst still experiencing them. Nevertheless, the heavy weight of ownership can be felt at any level. Regular face to face opportunities for informal conversations will be helpful and also managing the workload given to ECPs in particular. Whilst further research is required, in this study, those who had a high work ethic and seemed to take ownership of a high workload, were more likely to feel the weight of PO on their shoulders and may require stress management plans.

Recommendations related to ECPs in the workplace in particular:

Given that ECPs are likely to experience liminality, rituals and role models to aid this identity change are even more crucial. Senior managers and HR should consider how rituals are embedded into a work placement, internship and/or apprenticeships to expediate the transition from student to professional. Role models at all levels, including peer groups, provide possible selves.

When considering suitable roles for ECPs, HR should advise teams to provide positions with opportunities to control and job craft. Flexible boundaries to job roles provide greater opportunities to invest self and gain role-self congruity.

Given hybrid working is likely to be the norm in many organisations in the future, consider line managers who are accessible for both face to face meetings, but also accessible in attitude. This is a key socialisation period and early career professionals need to be able to see role models demonstrating "how work works", but also feel supported.

ECPs need to feel supported to learn from mistakes and to not carry too heavy an ownership burden. HR need to ensure that there are check-in points away from the line manager which provide psychological safety.

#### 5.5.4 Researchers

Most points relating to PO researcher can be found in the "future research directions" section. However, the point below refers more generally to researchers and thus feels more appropriate in the recommendations section.

A recommendation to researchers is that it can be beneficial to see constructs from a different perspective and we should actively encourage diversity of methods as well as other perspectives. Whilst in PO research there is a lack of qualitative research, in other disciplines it could equally be a lack of quantitative research.

#### 5.6 Contributions to Knowledge

#### This study has made the following contributions to knowledge:

Firstly, by exploring the development of PO at the start of an individual's career using qualitative research, this thesis has provided much needed methodological diversity to an area almost solely focussed on quantitative research. By utilising a different method, it has highlighted how ownership is constructed through individuals and their organisation thus emphasising the strong relational and contextual elements of PO. This study indicates how PO can be socially constructed, which I believe is an important practical contribution for organisations and individuals. Nonetheless, this study also contributes knowledge about PO methods which is also important for the academic community.

Secondly, by focussing on an under-explored group of participants, this research contributes to our understanding of PO development. Through demonstrating the complex, iterative dynamism of PO, this study has demonstrated that job-related ownership can be both "given" and "taken" in organisations and has provided a "giving ownership" model, demonstrating this process in early career professionals. The use of a building blocks demonstrates how organisations can encourage ownership development of the job role, whilst job crafting ascertains how individuals may be able to take control. These are all areas where other researchers may wish to consider feasibility with more experienced staff members.

Thirdly, this study contributes to PO research by suggesting that job-related PO may be developed far quicker than previously suggested (Pierce et al. 2001, 2003) if the right conditions are in place, such as those previously mentioned. Some individuals may bring an ownership mindset from previous experiences (such as part-time work and education) and take ownership even when not always actively permissioned. In addition, by using an atypical group of participants, this study suggests that PO can be developed by those in short-term roles or transitory roles and so can still be effectively utilised with

contract or temporary staff. This builds on our knowledge of how PO can be developed by different groups of individuals.

By highlighting the role of job crafting, this study links PO to ways in which individuals adapt their role to build self-congruity and self-efficacy. Whilst job design is suggested to be an important, the bottom-up approach of job crafting allows opportunities for self-investment which seemed equally important to this group of ECPs.

This study highlights the duality of PO and how it can result in both pleasure and pain. In particular the "weight of PO" has not been explored in such detail and should remind both academic and organisational communities of PO's possible negative connotations. Given this was a group of ECPs, there may be some reluctance to discuss the weight of ownership and thus bringing it to the forefront may allow supervisors and HR departments to put in place strategies to manage individuals who demonstrate high levels of ownership.

Finally, a **contribution to professional practice** is that by providing a "giving ownership" framework, supervisors, HR teams and educators are able to put in place "giving ownership" techniques, which provide a structure for developing job related PO in the workplace. This is a practical contribution to knowledge but does provide a concrete structure to manage PO development.

#### 5.7 Future Research Directions

There are so many fascinating areas for future research relating to PO, but further research into PO relating to selection and onboarding processes may demonstrate activities and rituals for early embedding of PO.

An obvious suggestion relates to the widening of PO research methods to ensure diversity of perspectives. Further qualitative studies would be welcomed, but the author believes there are also opportunities for mixed methodologies and the use of different technologies, such as digital data capture on social media. It would be beneficial for PO research to invoke the same sort of methodological diversity as the possession and ownership literature which, whilst comparably diverse, still believes that further diversity is needed. PO is complex and therefore methodological diversity will provide a broader picture of its intricacies.

Secondly, a greater understanding of how PO develops at the early career stage is important. This is especially true post-pandemic, as hybrid working becomes more normalised. Pre-pandemic, interviews with WPS suggested that some participants really missed having a line manager who was frequently in the same building, through whom they could role model "how work works", who could help with "silly" questions, and with whom they could build a relationship. These elements seem important and further research in this area could ascertain if this still holds true in a hybrid workplace.

More ECP research into internships, graduate roles, apprenticeships as well as work placements, would be beneficial. Internships tend to last for 12 weeks, which may be insufficient to build any long-standing PO attachments, though this is just surmising. Further research in this area would help us understand how quickly attachments can be built, and long-term research would also ascertain if the attachments can last post-graduation and beyond. This study also suggests that PO can be developed quickly and therefore those organisations who employ contract staff could still facilitate the "giving" ownership process. Research into contract roles and the gig economy would also help build knowledge in this area. Nevertheless, further research would clarify a potential tipping point, at which the length of assignment (when all other factors suit PO development) impacts on an individual's ability to develop PO.

From a different perspective, qualitative studies between line managers and their staff at different levels would also allow us to see if "giving ownership" is important with other staff members. Does "giving ownership" stop at a certain level and is it replaced with another form of ownership permissioning? What happens if individuals "take ownership" that hasn't been permissioned? One of the line managers in this study talked about ownership boundaries, which
partly relates to job crafting. The author has already suggested further research relating to the role of job crafting in PO development.

One of the key quotes from this study was around the duality of ownership, "getting you up in the morning, but keeping you awake at night", and again it would be interesting to consider the impact of being "possessed" by ownership of work-related targets.

This research also questions whether job crafting, rather than job design, is more important for job-related PO. Whilst job design might provide opportunities for autonomy and scope for PO, it may be the opportunity to job craft which ultimately feeds psychic energy and provides opportunities to make tasks, projects, or jobs "mine". Whilst both elements may be important, the opportunity to personalise may supersede the work job design can do.

Outcomes of pride and confidence were also suggested in this study, which had not been discussed previously. These may be particularly pertinent to these new employees. Nevertheless, these are emotions that can be reflected across the working population, thus warranting further development. The strength of job-related PO also highlights the force of feelings that some of these ECPs seem to experience. In particular, some individuals felt the "weight" of ownership and whilst the burden of ownership has briefly been mentioned previously, this research suggests how much it can be felt in some ECPs, even in organisations that feel they support a culture of ownership.

Further research relating to the relational aspects of PO transfer would be beneficial, including the mechanisms which may stop job-related PO being developed. Additionally, whilst it is possible to take ownership of the job role, how does this fit with organisational PO?

Finally, this study has also introduced a new potential PO target, careerrelated PO, in which some individuals felt a strong attachment to their career, which could supersede other organisational targets because of the ease in which it can be accessed. Whilst some theorists may suggest that everyone feels ownership for their career, the author would suggest this is not true, as we see many who are unhappy with their career, yet make minimal efforts to

206

change it. In this study, different strengths of career-related PO were suggested, which sometimes might have a negligible impact on other PO targets, but in others seemed to suggest an intensity which would result in it becoming their main PO attachment. Further research in this area would be interesting. The inclusion of liquid PO also suggests how new generations with less access to full ownership of possessions, may start to reflect that thinking in their work life and this, combined with the notion of "no job for life", may strengthen the career target as their one true target to be called "mine".

#### **5.8 Final Researcher Reflections**

Ownership is a word that is frequently referred to, but its conceptualisation is rarely considered in any detail. As a word, its meaning is clear and so why overthink something which has reasonably obvious connotations?

Yet, because we are dealing with humans, the actual process of "giving and taking ownership" is both simple and complex. The ownership process may be emotion-laden with previous experiences, personal feelings and values from both giver and taker, eliciting a significant level of complexity. This is a social exchange which can be fraught with sentiment that can allow the transfer of ownership to become confused. Yet when it is managed successfully and expectations are clear, there are benefits to both the individual and organisation. I believe that helping people understand the relational aspects of PO can have a positive impact in the workplace, as long as the burden of ownership is understood.

On reflection, the research questions selected were not utilised effectively and were too similar to the research objectives. Anderson (2013) suggest scholarly disagreements relating to their use given that some researchers use one or the other rather than both. Nevertheless, the value of good research questions are that they can shape a studies direction (Agee 2009), frame particular curiosities and influence the studies methods (Andrews 2003). Sandberg and Alvesson suggest curiosity, imagination and reflexivity are required to craft original research questions from literature reviews, a notion which is similar to the FINER criteria of feasibility, interesting, novel, ethical and relevant (Hulley et al. 2013). Sandberg and Alvesson go on to suggest that research

207

questions are most likely to be derived from empirical research, literature reviews. personal experience and society. I used personal interest initially to consider the PO topic, however it was via the literature review, that directed this study towards the "novel" aspect of relational PO, although the research questions could have been more considered. Nevertheless, in the future the results from this study might give direction for future research questions such as:

- Can individuals feel psychological ownership for their career?
  - What factors influence the development of career PO?
  - What impact does career PO have on other work related PO targets if any?
- What influences the demise of psychological ownership towards work place targets?
  - How do individuals feel about the demise of psychological ownership towards work place targets?
  - How does the demise of psychological ownership for work place targets impact on work place relationships?

Overall, I have loved having the opportunity to question PO! At the start of this PHD, I wished that I had the opportunity to be a full-time student and immerse myself. However, now I feel lucky to complete this doctorate on a part-time basis. The analysis took a long time, but having to complete other tasks gave me the headspace to look at things in a different way. I feel that ultimately the study is stronger for this, although that of course is for others to judge.

#### References

- Adil, A. and Kamal, A., 2018. Impact of Perceived Authentic Leadership and Psychological Capital on Burnout: Mediating Role of Psychological Ownership. *Psychological Studies*, 63 (3), 243-252.
- Adler, P. A. and Adler, P., 1987. *Membership roles in field research*. [online]. London: SAGE.
- Agee, J., 2009. Developing qualitative research questions: a reflective process. *International Journal of Qualitative Studies in Education*, 22 (4), 431-447.
- Aggett, M. and Busby, G., 2011. Opting out of internship: Perceptions of hospitality, tourism and events management undergraduates at a British University. *Journal of Hospitality, Leisure, Sport & Tourism Education (Oxford Brookes University)*, 10 (1), 106-113.
- Ahuvia, 2005. Beyond the Extended Self: Loved Objects and Consumers' Identity Narratives. Journal of Consumer Research, 32 (1), 171-184.
- Ahuvia, A., Garg, N., Batra, R., McFerran, B. and De Diesbach, P. B. L., 2018. Pride of Ownership: An Identity-Based Model. *Journal of the Association for Consumer Research*, 3 (2), 216-228.
- Alok, K., 2014. Authentic leadership and psychological ownership: investigation of interrelations. *Leadership & Organization Development Journal*, (4), 266-285.
- Anderson, V., 2013. *Research methods in human resource management*. 3rd ed. edition. London: Chartered Institute of Personnel and Development.
- Andrews, R., 2003. *Research Questions*. London, UNITED KINGDOM: Bloomsbury Publishing Plc.
- Argo, J. J., Dahl, D. W. and Morales, A. C., 2006. Consumer Contamination: How Consumers React to Products Touched by Others, *70*, 81-94.
- Asatryan, V. S., Slevitch, L., Larzelere, R., Morosan, C. and Kwun, D. J., 2013. Effects of Psychological Ownership on Students' Commitment and Satisfaction. *Journal of Hospitality & Tourism Education*, 25 (4), 169-179.
- Avey, J. B., Avolio, B. J., Crossley, C. D. and Luthans, F., 2009. Psychological ownership: Theoretical extensions, measurement and relation to work outcomes. *Journal of Organizational Behavior*, 30 (2), 173-191.
- Avey, J. B., Wernsing, T. S. and Palanski, M. E., 2012. Exploring the process of ethical leadership: The mediating role of employee voice and psychological ownership. *Journal of Business Ethics*, 107 (1), 21-34.
- Bardhi, F. and Eckhardt, G. M., 2012. Access-Based Consumption: The Case of Car Sharing, 881.
- Bardhi, F. and Eckhardt, G. M., 2017. Liquid Consumption. *Journal of Consumer Research*, 44 (3), 582-597.
- Bardhi, F., Eckhardt, G. M. and Arnould, E. J., 2012. Liquid Relationship to Possessions, 510.
- Baruch, Y., 2006. Career development in organizations and beyond: Balancing traditional and contemporary viewpoints. *Human Resource Management Review*, 16 (2), 125-138.
- Bauman, Z., 2000. Liquid modernity. Oxford: Polity.
- Bauman, Z., 2007. Liquid times : living in an age of uncertainty. Oxford: Polity.
- Bauman, Z., 2013. *Liquid Modernity* [Book]. Hoboken: Polity.
- Beech, N., 2011. Liminality and the practices of identity reconstruction. *Human Relations*, 64 (2), 285-302.
- Beggan, J. K., 1992. On the social nature of nonsocial perception: The mere ownership effect. Journal of Personality and Social Psychology, 62 (2), 229-237.
- Belk, R., 2010. Sharing. Journal of Consumer Research, 36 (5), 715-734.
- Belk, R., W., 2013. Extended Self in a Digital World. *Journal of Consumer Research*, 40 (3), 477-500.

- Belk, R. W., 1988. Possessions and the extended self. *Journal of Consumer Research*, 15 (2), 139-168.
- Belk, R. W., 2014. The Extended Self Unbound. *Journal of Marketing Theory & Practice*, 22 (2), 133-134.
- Bellezza, S., Ackerman, J. M. and Gino, F., 2017. "Be Careless with That!" Availability of Product Upgrades Increases Cavalier Behavior Toward Possessions. *Journal of Marketing Research (JMR)*, 54 (5), 768-784.
- Berger, R., 2015. Now I see it, now I don't: researcher's position and reflexivity in qualitative research. *Qualitative Research*, 15 (2), 219-234.
- Bermúdez, J. L., 2015. Bodily ownership, bodily awareness and knowledge without observation. *Analysis*, 75 (1), 37-45.
- Bernhard, F. and O'Driscoll, M. P., 2011. Psychological Ownership in Small Family-Owned
   Businesses: Leadership Style and Nonfamily-Employees' Work Attitudes and
   Behaviors. Group & Organization Management, 36 (3), 345-384.
- Bindl, U. K., Unsworth, K. L., Gibson, C. B. and Stride, C. B., 2019. Job Crafting Revisited: Implications of an Extended Framework for Active Changes at Work, *104*, 605-628.
- Blaikie, N., 2018. Confounding issues related to determining sample size in qualitative research (5. pp. 635). Great Britain: Taylor & Francis.
- Blake, P. R. and Harris, P. L., 2011. Early representations of ownership. *New directions for child and adolescent development*, 2011 (132), 39-51.
- Botsman, R. and Rogers, R., 2010. Beyond Zipcar: Collaborative Consumption. *Harvard Business Review*, 88 (10), 30-30.
- Brasel, S. A. and Gips, J., 2014. Tablets, touchscreens, and touchpads: How varying touch interfaces trigger psychological ownership and endowment. *Journal of Consumer Psychology*, 24 (2), 226-233.
- Braun, V. and Clarke, V., 2006. Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3 (2), 77-101.
- Braun, V. and Clarke, V., 2013. *Successful qualitative research : a practical guide for beginners*. Los Angeles: SAGE.
- Braun, V. and Clarke, V., 2014. What can "thematic analysis" offer health and wellbeing researchers? *International Journal of Qualitative Studies on Health and Well-being*, 9.
- Braun, V. and Clarke, V., 2016. (Mis)conceptualising themes, thematic analysis, and other problems with Fugard and Potts' (2015) sample-size tool for thematic analysis. *International Journal of Social Research Methodology*, 19 (6), 739-743.
- Braun, V. and Clarke, V., 2019. Reflecting on reflexive thematic analysis (4. pp. 589): Taylor & Francis.
- Braun, V. and Clarke, V., 2021a. Can I use TA? Should I use TA? Should I not use TA? Comparing reflexive thematic analysis and other pattern-based qualitative analytic approaches. *Counselling & Psychotherapy Research*, 21 (1), 37-47.
- Braun, V. and Clarke, V., 2021b. One size fits all? What counts as quality practice in (reflexive) thematic analysis? (3. pp. 328). Great Britain: Taylor and Francis.
- Braun, V. and Clarke, V., 2021c. One size fits all? What counts as quality practice in (reflexive) thematic analysis? *Qualitative Research in Psychology*, 18 (3), 328-352.
- Brehm, J. W., 1956. Postdecision changes in the desirability of alternatives. *The Journal of Abnormal and Social Psychology*, 52 (3), 384-389.
- Brinkmann, S. and Kvale, S., 2015. *Interviews : learning the craft of qualitative research interviewing* [Non-fiction]. Third edition / Svend Brinkmann, Steinar Kvale. edition.: SAGE.
- Brooks, J. and Lewis, M., 1974. Attachment behavior in thirteen-month-old, opposite-sex twins. *Child Development*, 45 (1), 243-247.
- Brooks, J., McCluskey, S., Turley, E. and King, N., 2015. The utility of Template Analysis in qualitative psychology research. *Qualitative Research in Psychology*, 12 (2), 202-222.

- Brooks, R. and Youngson, P. L., 2016. Undergraduate work placements: an analysis of the effects on career progression (9. Vol. 41, pp. 1563-1578). Great Britain: Taylor & Francis.
- Brown, B., Perkins, D. D. and Brown, G., 2003. Place attachment in a revitalizing neighborhood: Individual and block levels of analysis. *Journal of Environmental Psychology*, 23, 259-271.
- Brown, G., 2009. Claiming a corner at work: Measuring employee territoriality in their workspaces. *Journal of Environmental Psychology*, 29 (1), 44-52.
- Brown, G., Brown, Thomas, B. L. and Sandra, L. R., 2005a. Territoriality in Organizations. *The Academy of Management Review*, (3), 577.
- Brown, G., Crossley, C. and Robinson, S. L., 2014a. Psychological Ownership, Territorial Behavior, and Being Perceived as a Team Contributor: The Critical Role of Trust in the Work Environment. *Personnel Psychology*, 67 (2), 463-485.
- Brown, G., Lawrence, T. B. and Robinson, S. L., 2005b. TERRITORIALITY IN ORGANIZATIONS. *Academy of Management Review*, 30 (3), 577-594.
- Brown, G., Pierce, J. L. and Crossley, C., 2014b. Toward an understanding of the development of ownership feelings. *Journal of Organizational Behavior*, 35 (3), 318-338.
- Brown, G. and Zhu, H., 2016. 'My workspace, not yours': The impact of psychological ownership and territoriality in organizations. *Journal of Environmental Psychology*, 48, 54-64.

Burk, C. F., 1900. The collecting instinct. *Pedag. Sem.*, VII, 179-207.

- Carminati, L., 2018. Generalizability in Qualitative Research: A Tale of Two Traditions. *Qualitative health research*, 28 (13), 2094-2101.
- Carter, T. J. and Gilovich, T., 2012. I Am What I Do, Not What I Have: The Differential Centrality of Experiential and Material Purchases to the Self, 1304.
- Carver, C. S., Sinclair, S. and Johnson, S. L., 2010. Authentic and hubristic pride: Differential relations to aspects of goal regulation, affect, and self-control. *Journal of Research in Personality*, 44 (6), 698-703.
- Chi, N.-W. and Han, T.-S., 2008. Exploring the linkages between formal ownership and psychological ownership for the organization: The mediating role of organizational justice. *Journal of Occupational and Organizational Psychology*, 81 (4), 691-711.
- Clarke, V. and Braun, V., 2017. Thematic analysis (Vol. 12, pp. 297-298). Oxfordshire, : Routledge.
- Clarke, V. and Braun, V., 2018. Using thematic analysis in counselling and psychotherapy research: A critical reflection. *Counselling & Psychotherapy Research*, 18 (2), 107-110.
- Cocieru, O. C., Delia, E. B. and Katz, M., 2019a. It's our club! From supporter psychological ownership to supporter formal ownership. *Sport Management Review*, 22 (3), 322-334.
- Cocieru, O. C., Lyle, M. C. B., Hindman, L. C. and McDonald, M. A., 2019b. The 'Dark Side' of Psychological Ownership during Times of Change. *Journal of Change Management*, 19 (4), 266.
- Cocieru, O. C., Lyle, M. C. B., Hindman, L. C. and McDonald, M. A., 2019c. The 'Dark Side' of Psychological Ownership during Times of Change. *Journal of Change Management*, 19 (4), 266-282.
- Colquitt, J. A. and Chertkoff, J. M., 2002. Explaining Injustice: The Interactive Effect of Explanation and Outcome on Fairness Perceptions and Task Motivation. *Journal of Management*, 28 (5), 591-610.
- Constantinou, C. S., Georgiou, M. and Perdikogianni, M., 2017. A comparative method for themes saturation (CoMeTS) in qualitative interviews. *Qualitative Research*, 17 (5), 571-588.
- Crawford, I. and Wang, Z., 2015. The impact of individual factors on the academic attainment of Chinese and UK students in higher education (5. Vol. 40, pp. 902-920). Great Britain: Taylor & Francis.

Creswell, J. W., 2015. Educational Research: Planning, Conducting, and Evaluating Quantitative and Qualitative Research, Enhanced Pearson eText with Loose-Leaf Version--Access Card Package. Fifth Edition. New Jersey: Pearson Education, Inc.

- Creswell, J. W. and Poth, C. N., 2018. *Qualitative inquiry & research design : choosing among five approaches*. Fourth edition. Los Angeles: SAGE.
- Cron, W. L., 1984. Industrial salesperson development: A career stages perspective. *Journal of Marketing*, 48 (4), 41-52.
- Cron, W. L. and Slocum, J. W., 1986. The influence of career stages on salespeople's job attitudes, work perceptions, and performance. *Journal of Marketing Research*, 23 (2), 119-129.
- Crotty, M., 1998. *The Foundations of Social Research : Meaning and Perspective in the Research Process*. London: SAGE Publications Ltd.
- Csikszentmihalyi, M. and Rochberg-Halton, E., 1981. *The meaning of things : domestic symbols and the self* [Non-fiction]. Cambridge University Press.
- Dai, Y.-D., Altinay, L., Zhuang, W.-L. and Chen, K.-T., 2021. Work engagement and job burnout? Roles of regulatory foci, supervisors' organizational embodiment and psychological ownership. *Journal of Hospitality and Tourism Management*, 46, 114-122.
- Dalton, G. W., Thompson, P. H. and Price, R. L., 1977. Department Research: The four stages of professional careers— A new look at performance by professionals. *Organizational Dynamics*, 6, 19-42.
- Dawkins, S., Tian, A. W., Newman, A. and Martin, A., 2017. Psychological ownership: A review and research agenda. *Journal of Organizational Behavior*, 38 (2), 163-183.
- De Vos, A., Jacobs, S. and Verbruggen, M., 2021. Career transitions and employability. *Journal* of Vocational Behavior, 126.
- Dearing, R. S., 1997. *Higher education in the learning society*. London: Her Majesty's Stationery Office.
- DeFillippi, R. J. and Arthur, M. B., 1994. The Boundaryless Career: A Competency-Based Perspective (4. Vol. 15, pp. 307-307). Great Britain: JOHN WILEY & SONS LTD.
- Demerouti, E., 2014. Design your own job through job crafting. *European Psychologist*, 19 (4), 237-247.
- Denegri-Knott, J., Jenkins, R. and Lindley, S., 2020. What is digital possession and how to study it: a conversation with Russell Belk, Rebecca Mardon, Giana M. Eckhardt, Varala Maraj, Will Odom, Massimo Airoldi, Alessandro Caliandro, Mike Molesworth and Alessandro Gandini.
- Denegri-Knott, J. and Molesworth, M., 2009. 'I'll sell this and I'll buy them that': eBay and the management of possessions as stock. *Journal of Consumer Behaviour*, 8 (6), 305-315.
- Denegri-Knott, J., Watkins, R. and Molesworth, M., 2016. The relationship between ownership and possession: observations from the context of digital virtual goods.
- Denzin, N. K., 1978. *The research act : a theoretical introduction to sociological methods* [Non-fiction]. 2nd ed. edition.: McGraw-Hill.
- Denzin, N. K. and Lincoln, Y. S., 2013. *Collecting and interpreting qualitative materials* [Non-fiction]. 4 edition. edition.: SAGE.
- Denzin, N. K. and Lincoln, Y. S., 2018. *The SAGE handbook of qualitative research*. Fifth edition. edition. Los Angeles: SAGE.
- Di Muro, F. and Noseworthy, T. J., 2013. Money Isn't Everything, but It Helps If It Doesn't Look Used: How the Physical Appearance of Money Influences Spending. *Journal of Consumer Research*, 39 (6), 1330-1342.
- Dik, B. J., Duffy, R. D. and Eldridge, B. M., 2009. Calling and vocation in career counseling: Recommendations for promoting meaningful work. *Professional Psychology: Research and Practice*, 40 (6), 625-632.

- Direnzo, M. S. and Greenhaus, J. H., 2011. Job search and voluntary turnover in a boundaryless world: A control theory perspective. *The Academy of Management Review*, 36 (3), 567-589.
- Dittmar, H., 1992. *The social psychology of material possessions: to have is to be*. Hemel Hempstead: Harvester Wheatsheaf.
- Dodgson, J. E., 2019. Reflexivity in Qualitative Research. *Journal of human lactation : official journal of International Lactation Consultant Association*, 35 (2), 220-222.
- Donald, W. E., Baruch, Y. and Ashleigh, M., 2019. The undergraduate self-perception of employability: human capital, careers advice, and career ownership (4. pp. 599). Great Britain: Taylor & Francis.
- Duffy, R. D., Dik, B. J., Douglass, R. P., England, J. W. and Velez, B. L., 2018. Work as a calling: A theoretical model. *Journal of counseling psychology*, 65 (4), 423-439.
- Dwyer, S. C. and Buckle, J. L., 2009. The Space Between: On Being an Insider-Outsider in Qualitative Research. *International Journal of Qualitative Methods*, 8 (1), 54-63.
- Edwards, M., 2014a. The impact of placements on students' self-efficacy. *Higher Education, Skills & Work-based Learning*, 4 (3), 228-241.
- Edwards, M., 2014b. The impact of placements on students' self-efficacy. *Higher Education, Skills and Work-based Learning*, 4 (3), 228-241.
- Eldh, A. C., Årestedt, L. and Bertero, C., 2020. Quotations in Qualitative Studies: Reflections on Constituents, Custom, and Purpose. *International Journal of Qualitative Methods*, 19, 1-6.
- Emmel, N., 2013. *Sampling and choosing cases in qualitative research : a realist approach*. Los Angeles: SAGE.
- Eynon, M. J., O'Donnell, C. and Williams, L., 2018. Gaining qualitative insight into the subjective experiences of adherers to an exercise referral scheme: A thematic analysis. *Journal of health psychology*, 23 (11), 1476-1487.
- Fasig, L. G., 2000. Toddlers' understanding of ownership: Implications for self-concept development. *Social Development*, 9 (3), 370-382.
- Ferraro, R., Escalas, J. E. and Bettman, J. R., 2011. Our possessions, our selves: Domains of self-worth and the possession-self link. *Journal of Consumer Psychology*, 21 (2), 169-177.
- Fleura Bardhi, a., Giana M. Eckhardt, a. and Eric J. Arnould, a., 2012. Liquid Relationship to Possessions. *Journal of Consumer Research*, (3), 510.
- Forrier, A., Verbruggen, M. and De Cuyper, N., 2015. Integrating different notions of employability in a dynamic chain: The relationship between job transitions, movement capital and perceived employability. *Journal of Vocational Behavior*, 89, 56-64.
- Friedman, O. and Neary, K. R., 2008. Determining who owns what: Do children infer ownership from first possession? *Cognition*, 107 (3), 829-849.
- Fritze, M. P., Marchand, A., Eisingerich, A. B. and Benkenstein, M., 2020. Access-Based Services as Substitutes for Material Possessions: The Role of Psychological Ownership. *Journal of Service Research*, 23 (3), 368-385.
- Fugard, A. J. B. and Potts, H. W. W., 2015. Supporting thinking on sample sizes for thematic analyses: a quantitative tool (6. pp. 669). Great Britain: Taylor & Francis.
- Furby, L., 1978a. Possession in humans: An exploratory study of its meaning and motivation. *Social Behavior and Personality*, 6 (1), 49-65.
- Furby, L., 1978b. Sharing: Decisons and moral judgments about letting others use one's possessions. *Psychological Reports*, 43 (2), 595-609.
- Galvin, B. M., Lange, D. and Ashforth, B. E., 2015. Narcissistic organization identification: Seeing oneself as central to the organization's identity. *Academy of Management Review*, 40 (2), 163.
- Garton, S. and Copland, F., 2010. 'I like this interview; I get cakes and cats!': The effect of prior relationships on interview talk. *Qualitative Research*, 10 (5), 533-551.

Gennep, A. v., 1960. The rites of passage [Non-fiction]. University of Chicago Press.

- Gennep, A. v., 2013. *The Rites of Passage* [Book]. London: Routledge.
- Graham, B. and Sandra, L. R., 2011. Reactions to Territorial Infringement. *Organization Science*, (1), 210.
- Gray, S. M., Knight, A. P. and Baer, M., 2020. On the Emergence of Collective Psychological Ownership in New Creative Teams, 141.
- Greenhaus, J. H., Callanan, G. A. and Godshalk, V. M., 2010. *Career management. 4th edition*. Thousand Oaks, CA: SAGE, 2010
- Greenhaus, J. H. and Parasuraman, S., 1986. Vocational and organizational behavior, 1985: A review. *Journal of Vocational Behavior*, 29, 115-176.
- Guest, G., Bunce, A. and Johnson, L., 2006. How Many Interviews Are Enough?: An Experiment with Data Saturation and Variability, 59.
- Guetterman, T. C., 2015. Descriptions of sampling practices within five approaches to qualitative research in education and the health sciences: DEU.
- Hagens, V., Dobrow, M. J. and Chafe, R., 2009. Interviewee Transcript Review: assessing the impact on qualitative research. *BMC medical research methodology*, 9, 47.
- Hall, D. T., Feldman, E. and Kim, N., 2013. Meaningful work and the protean career. *In:* Dik, B.
  J., Byrne, Z. S. and Steger, M. F., eds. *Purpose and meaning in the workplace*.
  Washington, DC: American Psychological Association, 57-78.
- Hall, D. T. and Moss, J. E., 1998. The new protean career contract: Helping organizations and employees adapt. *Organizational Dynamics*, 26 (3), 22-37.
- Han, T.-S., Chiang, H.-H. and Chang, A., 2010. Employee participation in decision making, psychological ownership and knowledge sharing: Mediating role of organizational commitment in Taiwanese high-tech organizations. *The International Journal of Human Resource Management*, 21 (12), 2218-2233.
- Harmon-Jones, E. and Mills, J., 1999. An introduction to cognitive dissonance theory and an overview of current perspectives on the theory. *In:* Harmon-Jones, E., Mills, J.,
   Harmon-Jones, E. and Mills, J., eds. *Cognitive dissonance: Progress on a pivotal theory in social psychology.* Washington, DC, US: American Psychological Association, 3-21.
- Hart, A. and Montague, J., 2015. Negotiating Identity and Alterity: Cultural Competence, Colonization and Cultural Voyeurism in Students' Work-Based Learning. *Journal of Psychological Issues in Organizational Culture*, 6 (3), 7-23.
- He, P. and Pierce, J., 2015. Job- and organization-based psychological ownership: relationship and outcomes. *Journal of Managerial Psychology*, 30 (2), 151.
- Heider, F., 1958. The psychology of interpersonal relations. . Hoboken: Wiley.
- Henri, H. B. B., Guy, G. P. P. and Claude, C. S. S., 2008. Linking IT implementation and acceptance via the construct of psychological ownership of information technology. *Journal of Information Technology (Palgrave Macmillan)*, 23 (4), 269-280.
- Henssen, B. and Koiranen, M., 2021. CEOs' Joy of Working for the Family Firm: The Role of Psychological Ownership and Stewardship Behavior. *Entrepreneurship Research Journal*, 11 (3), 191-218.
- Henssen, B., Voordeckers, W., Lambrechts, F. and Koiranen, M., 2014. The CEO autonomy– stewardship behavior relationship in family firms: The mediating role of psychological ownership. *Journal of Family Business Strategy*, 5 (3), 312.
- Hershcovis, M. S., Ogunfowora, B., Reich, T. C. and Christie, A. M., 2017. Targeted workplace incivility: The roles of belongingness, embarrassment, and power. *Journal of Organizational Behavior*, 38 (7), 1057-1075.
- Higgins, E. T., 1997. Beyond pleasure and pain. American Psychologist, 52 (12), 1280-1300.
- Hillenbrand, C. and Money, K. G., 2015. Unpacking the Mechanism by Which Psychological Ownership Manifests at the Level of the Individual: A Dynamic Model of Identity and Self. *Journal of Marketing Theory & Practice*, 23 (2), 148-165.

- Holmes, L. M., 2015. Becoming a graduate: the warranting of an emergent identity. *Education* + *Training*, 57 (2), 219-238.
- Howell, R. T. and Hill, G., 2009. The mediators of experiential purchases: Determining the impact of psychological needs satisfaction and social comparison. *The Journal of Positive Psychology*, 4 (6), 511-522.
- Hulland, J., Thompson, S. A. and Smith, K. M., 2015. Exploring Uncharted Waters: Use of Psychological Ownership Theory in Marketing. *Journal of Marketing Theory & Practice*, 23 (2), 140.
- Hulley, S. B., Cummings, S. R., Browner, W. S., Grady, D. and Newman, T. B., 2013. Designing clinical research [online]. Fourth edition. edition. Philadelphia: Wolters Kluwer/Lippincott Williams & Wilkins.
- Humphreys, A. and Grayson, K., 2008. The Intersecting Roles of Consumer and Producer: A Critical Perspective on Co-production, Co-creation and Prosumption, 963.
- Ibarra, H., 1999. Provisional selves: Experimenting with image and identity in professional adaptation. *Administrative Science Quarterly*, 44 (4), 764-791.
- Ibarra, H. and Barbulescu, R., 2010. Identity as narrative: Prevalence, effectiveness, and consequences of narrative identity work in macro work role transitions. *The Academy of Management Review*, 35 (1), 135-154.
- Ibarra, H. and Petriglieri, J. L., 2010. Identity work and play. *Journal of Organizational Change Management*, 23 (1), 10-25.
- Inceoglu, I., Selenko, E., McDowall, A. and Schlachter, S., 2019. (How) Do work placements work? Scrutinizing the quantitative evidence for a theory-driven future research agenda. *Journal of Vocational Behavior*, 110 (Part B), 317-337.
- Irit, M.-J., 2011. 'Is that what I Said?' Interview Transcript Approval by Participants: An Aspect of Ethics in Qualitative Research. *International Journal of Qualitative Methods*, 10.
- Isaacs, S., 1933. Social Development in Young Children. 3 (3), 291-294.
- Islam, G. and Zyphur, M. J., 2009. Rituals in Organizations: A Review and Expansion of Current Theory, 114.
- Jaaffar, A. H., Ibrahim, H. I., Rajadurai, J. and Sohail, M. S., 2019. Psychological Impact of Work-Integrated Learning Programmes in Malaysia: The Moderating Role of Self-Esteem on Relation between Self-Efficacy and Self-Confidence. *International Journal* of Educational Psychology, 8 (2), 188-213.
- Jackson, D., 2015. Employability Skill Development in Work-Integrated Learning: Barriers and Best Practice. *Studies in Higher Education*, 40 (2), 350-367.
- Jackson, D., 2017. Developing Pre-Professional Identity in Undergraduates through Work-Integrated Learning. *Higher Education: The International Journal of Higher Education Research*, 74 (5), 833-853.
- Jackson, D. and Collings, D., 2018. The influence of Work-Integrated Learning and paid work during studies on graduate employment and underemployment. *Higher Education*, 76 (3), 403-425.
- James, W., 1890. The principles of psychology. New York: H. Holt.
- Jones, C. M., Green, J. P. and Higson, H. E., 2017. Do work placements improve final year academic performance or do high-calibre students choose to do work placements?
   (6. Vol. 42, pp. 976-992). Great Britain: Taylor & Francis.
- Jussila, I., Goel, S. and Tuominen, H., 2012. Member Commitment in Co-operatives: The Utilitarian Approach. *Business & Management Research*, 1 (3), 9-16.
- Jussila, I., Tarkiainen, A., Sarstedt, M. and Hair, J. F., 2015. Individual Psychological Ownership: Concepts, Evidence, and Implications for Research in Marketing. *Journal* of Marketing Theory & Practice, 23 (2), 121-139.
- Kalish, C. W. and Anderson, C. D., 2011. Ownership as a social status. (132), 65-77.
- Kanungo, R. N., 1982. Measurement of job and work involvement. *Journal of Applied Psychology*, 67 (3), 341-349.

Karahanna, E., Xu, S. X. and Zhang, N., 2015. Psychological Ownership Motivation and Use of Social Media. *Journal of Marketing Theory & Practice*, 23 (2), 185-207.

- Karanika, K. and Hogg, M. K., 2013. Trajectories across the lifespan of possession-self relationships. *Journal of Business Research*, 66 (7), 910-916.
- Kaur, D., Sambasivan, M. and Kumar, N., 2013. Effect of spiritual intelligence, emotional intelligence, psychological ownership and burnout on caring behaviour of nurses: A cross-sectional study. *Journal of Clinical Nursing*, 22 (21-22), 3192-3202.
- Kidder, L. H. and Fine, M., 1987. Qualitative and quantitative methods: when stories converge. *New Directions for Program Evaluation*, (35), 57-75.
- Kim, M. and Beehr, T. A., 2017. Self-Efficacy and Psychological Ownership Mediate the Effects of Empowering Leadership on Both Good and Bad Employee Behaviors. *Journal of Leadership & Organizational Studies*, 24 (4), 466-478.
- Kim, S. and Kalish, C. W., 2009. Children's ascriptions of property rights with changes of ownership. *Cognitive Development*, 24 (3), 322-336.
- Kirk, C. P., Swain, S. D. and Gaskin, J. E., 2015. I'm Proud of It: Consumer Technology Appropriation and Psychological Ownership. *Journal of Marketing Theory & Practice*, 23 (2), 166.
- Kleine, S. S., Kleine, R. E., III and Allen, C. T., 1995. How is a possession 'me' or 'not me'? Characterizing types and an antecedent of material possession attachment. *Journal of Consumer Research*, 22 (3), 327-343.
- Knapp, J. R., Smith, B. R. and Sprinkle, T. A., 2014. Clarifying the Relational Ties of Organizational Belonging: Understanding the Roles of Perceived Insider Status, Psychological Ownership, and Organizational Identification. *Journal of Leadership & Organizational Studies*, 21 (3), 273-285.
- Knapp, J. R., Sprinkle, T. A., Urick, M. J. and Delaney-Klinger, K. A., 2019. The belonging model of trust. *Nonprofit Management & Leadership*, 30 (1), 133-153.
- Kramer, T. and Block, L. G., 2014. Like Mike: Ability contagion through touched objects increases confidence and improves performance, 215.
- Kumar, A., 2007. *Personal, academic and career development in higher education : SOARing to success*. London: Routledge.
- Kvale, S., 2007. Doing interviews. . London: SAGE.
- Lastovicka, J. L. and Fernandez, K. V., 2005. Three Paths to Disposition: The Movement of Meaningful Possessions to Strangers, 813.
- Law, J., 2004. *After method: mess in social science research*. London; New York: Routledge.
- Lawler, E. E. d. and Hall, D. T., 1970. Relationship of job characteristics to job involvement, satisfaction, and intrinsic motivation. *Journal of Applied Psychology*, 54, 305-312.
- Lawson, C., Latsis, J. S. and Martins, N., 2007. *Contributions to Social Ontology* [Book]. London: Routledge.
- Lazazzara, A., Tims, M. and de Gennaro, D., 2020. The process of reinventing a job: A meta– synthesis of qualitative job crafting research. *Journal of Vocational Behavior*, 116 (Part B).
- Levitt, H. M., Motulsky, S. L., Wertz, F. J., Morrow, S. L. and Ponterotto, J. G., 2017. Recommendations for designing and reviewing qualitative research in psychology: Promoting methodological integrity. *Qualitative Psychology*, 4 (1), 2-22.
- Li, R., Ling, W.-Q. and Liu, S.-S., 2012. The antecedents and outcomes of psychological ownership for the organization: An analysis from the perspective of person-situation interactions. *Acta Psychologica Sinica*, 44 (9), 1202-1216.
- Li, X. and Qian, J., 2016. Stimulating employees' feedback-seeking behavior: The role of participative decision making, *44*, 1-8.
- Lincoln, Y. S., Guba, E. G. and Pilotta, J. J., 1985. Naturalistic inquiry: Beverly Hills, CA: Sage Publications, 1985, 416 pp., \$25.00 (Cloth). *International Journal of Intercultural Relations*, 9 (4), 438-439.

- Liu, J., Wang, H., Hui, C. and Lee, C., 2012. Psychological Ownership: How Having Control Matters. *Journal of Management Studies*, 49 (5), 869-895.
- Locke, 1690. Two treatises of government. London: Awnsham Churchill.
- Locke, A., 2004. Accounting for Success and Failure: A Discursive Psychological Approach to Sport Talk, 302.
- Lodahl, T. M. and Kejner, M., 1965. Definition and measurement of job involvement. *Journal* of Applied Psychology, 49, 24-33.
- Lu, L., Jun, L. and Na, Z., 2017. Why employees stay: the roles of psychological ownership, territoriality and work relationship closeness in affecting employee turnover intention. *Frontiers of Business Research in China*, 11 (2), 186-120.
- Mahto, R. V., Ahluwalia, S. and Khanin, D., 2014. Psychological Ownership of Family Firm Successors: A Conceptual Approach. *Small Business Institute® Journal (SBIJ)*, 10 (2), 65.
- Markus, H. and Nurius, P. S., 1986. Possible selves. American Psychologist, 41, 954-969.
- Mason, G., Williams, G. and Cranmer, S., 2009. Employability skills initiatives in higher education: what effects do they have on graduate labour market outcomes?, *17*, 1-30.
- Masset, J. and Decrop, A., 2016. 'God, I have so many ashtrays!' Dependences and dependencies in consumer–possession relationships. *Journal of Business Research*, 69 (1), 94-109.
- Matilainen, A., Pohja-Mykrä, M., Lähdesmäki, M. and Kurki, S., 2017. "I feel it is mine!" Psychological ownership in relation to natural resources. *Journal of Environmental Psychology*, 51, 31-45.
- Mayhew, M. G., Ashkanasy, N. M., Bramble, T. and Gardner, J., 2007. A Study of the Antecedents and Consequences of Psychological Ownership in Organizational Settings. *Journal of Social Psychology*, 147 (5), 477-500.
- Mayring, P., 2007. On Generalization in Qualitatively Oriented Research. *Forum: Qualitative Social Research*, 8 (3), 1.
- McClelland, A. M. and McManus, R. L., 1941. Appraising personality. *AJN American Journal of Nursing*, 41, 579-584.
- McCracken, G., 1986. Culture and consumption: A theoretical account of the structure and movement of the cultural meaning of consumer goods. *Journal of Consumer Research*, 13 (1), 71-84.
- McLntyre, N., Srivastava, A. and Fuller, J. A., 2009. The relationship of locus of control and motives with psychological ownership in organization. *Journal of Managerial Issues*, 21 (3), 383-401.
- Md-Sidin, S., Sambasivan, M. and Muniandy, N., 2009. Impact of Psychological Ownership on the Performance of Business School Lecturers. *Journal of Education for Business*, 85 (1), 50-56.
- Mele, E., Español, A., Carvalho, B. and Marsico, G., 2021. Beyond technical learning: Internship as a liminal zone on the way to become a psychologist. *Learning, Culture and Social Interaction*, 28.
- Mifsud, M., Molines, M., Cases, A.-S. and N'Goala, G., 2019. It's MY health care program: Enhancing patient adherence through psychological ownership. *Social Science & Medicine*, 232, 307-315.
- Milkovich, G. and Hall, D. T., 1977. Careers in Organizations. *Administrative Science Quarterly* ; volume 22, issue 3, page 535 ; ISSN 0001-8392.
- Mittal, B., 2006. I, me, and mine-how products become consumers extended selves, 550.
- Molesworth, M., Watkins, R. and Denegri-Knott, J., 2016. Possession Work on Hosted Digital Consumption Objects as Consumer Ensnarement. *Journal of the Association for Consumer Research*, 1 (2), 246.
- Moores, E. and Reddy, P., 2012. No regrets? Measuring the career benefits of a psychology placement year, 535.

- Morewedge, C. K., 2021. Psychological ownership: implicit and explicit. *Current Opinion in Psychology*, 39, 125-132.
- Morewedge, C. K., Monga, A., Palmatier, R. W., Shu, S. B. and Small, D. A., 2021. Evolution of Consumption: A Psychological Ownership Framework. *Journal of Marketing*, 85 (1), 196-218.
- Morrison, K. R. and Johnson, C. S., 2011. When what you have is who you are: selfuncertainty leads individualists to see themselves in their possessions. *Personality & social psychology bulletin*, 37 (5), 639-651.
- Morse, J. M., 2015. Critical analysis of strategies for determining rigor in qualitative inquiry. *Qualitative Health Research*, 25 (9), 1212-1222.
- Naeem, R. M., Channa, K. A., Hameed, Z., Ali Arain, G. and Islam, Z. U., 2021. The future of your job represents your future: a moderated mediation model of transformational leadership and job crafting. *Personnel Review*, 50 (1), 207-224.
- Newheiser, A. K., Barreto, M. and Tiemersma, J., 2017. People like me don't belong here: Identity concealment is associated with negative workplace experiences. *Journal of Social Issues*, 73 (2), 341-358.
- O' Driscoll, M. P., Pierce, J. L. and Coghlan, A.-M., 2006. The Psychology of Ownership: Work Environment Structure, Organizational Commitment, and Citizenship Behaviors. *Group & Organization Management*, 31 (3), 388-416.
- Odio, M. A., McLeod, C. M. J. E. L. and Education, T. i. H., 2021. Using Liminality to Understand How Identity and Temporary Status Influence Interns' Vulnerability. 4 (2), 92-98.
- Olckers, C., 2013. Psychological ownership: Development of an instrument. SAJIP: South African Journal of Industrial Psychology, 39 (2), 1-13.
- Olckers, C. and du Plessis, Y., 2014. Exploring the Theoretical Relationship Between Psychological Ownership and Career Anchors. *In:* Coetzee, M., ed. *Psycho-social Career Meta-capacities: Dynamics of contemporary career development.* Cham: Springer International Publishing, 155-173.
- Olckers, C. and Enslin, C., 2016a. Psychological ownership in relation to workplace trust and turnover intent (2. pp. 119): Taylor & Francis.
- Olckers, C. and Enslin, C., 2016b. Psychological ownership in relation to workplace trust and turnover intent. *Journal of Psychology in Africa*, 26 (2), 119-126.
- Olckers, C. and Koekemoer, E., 2017. Linking Psychological Ownership with Subjective Career Success and Positive Work-Related Outcomes. *In:* Olckers, C., van Zyl, L. and van der Vaart, L., eds. *Theoretical Orientations and Practical Applications of Psychological Ownership.* Cham: Springer International Publishing, 3-20.
- Olckers, C. and Schaap, P., 2013. Measurement Equivalence of the South African Psychological Ownership Questionnaire for Diverse South African Cultural Groups. *Journal of Psychology in Africa*, 23 (4), 569.
- Olckers, C. and Zyl, L., 2016. The Relationship Between Employment Equity Perceptions and Psychological Ownership in a South African Mining House: The Role of Ethnicity. *Social Indicators Research*, 127 (2), 887.
- Ori, F., Julia, W. V. d. V., Margaret, A. D. and Karen, R. N., 2013. First Possession, History, and Young Children's Ownership Judgments. *Child Development*, 84 (5), 1519-1525.
- Orth, D., Thurgood, C. and Van Den Hoven, E., 2019. Designing meaningful products in the digital age: How users value their technological possessions. *ACM Transactions on Computer-Human Interaction*, 26 (5), 1-28.
- Palinkas, L. A., Horwitz, S. M., Green, C. A., Wisdom, J. P., Duan, N. and Hoagwood, K., 2013. Purposeful Sampling for Qualitative Data Collection and Analysis in Mixed Method Implementation Research. *Administration and Policy in Mental Health and Mental Health Services Research ; volume 42, issue 5, page 533-544*
- Park, C. H., Kim, W. and Song, J. H., 2015. The impact of ethical leadership on employees' inrole performance: The mediating effect of employees' psychological ownership. *Human Resource Development Quarterly*, 26 (4), 385-408.

- Park, C. H., Song, J. H., Yoon, S. W. and Kim, J., 2013. A missing link: psychological ownership as a mediator between transformational leadership and organizational citizenship behaviour. *Human Resource Development International*, 16 (5), 558.
- Parker, S. K., Wall, T. D. and Jackson, P. R., 1997. 'That's not my job': Developing flexible employee work orientations. *Academy of Management Journal*, 40 (4), 899-929.
- Patton, M. Q., 1999. Enhancing the quality and credibility of qualitative analysis. *Health* services research, 34 (5 Pt 2), 1189-1208.
- Patton, M. Q., 2002. Two decades of developments in qualitative inquiry: a personal, experiential perspective. *Qualitative Social Work*, 1 (3), 261-283.
- Peck, J. and Shu, S. B., 2018. *Psychological ownership and consumer behavior*. Cham: Springer International Publishing.
- Peeters, E., Nelissen, J., De Cuyper, N., Forrier, A., Verbruggen, M. and De Witte, H., 2019. Employability Capital: A Conceptual Framework Tested Through Expert Analysis, *46*, 79-93.
- Peng, H., 2013. Why and when do people hide knowledge? *Journal of Knowledge Management*, 17 (3), 398-415.
- Peng, H. and Pierce, J., 2015. Job- and organization-based psychological ownership: Relationship and outcomes. *Journal of Managerial Psychology*, 30 (2), 151-168.
- Petrelli, D. and Whittaker, S., 2010. Family memories in the home: contrasting physical and digital mementos, 153.
- Pick, M., 2021. Psychological ownership in social media influencer marketing. *European Business Review*, 33 (1), 1-22.
- Pierce, Stephen, A. R. and Susan, M., 1991. Employee Ownership: A Conceptual Model of Process and Effects. *The Academy of Management Review*, (1), 121.
- Pierce, J. L. and Furo, C. A., 1990. Employee ownership: Implications for management. Organizational Dynamics, 18 (3), 32-43.
- Pierce, J. L. and Jussila, I., 2010. Collective psychological ownership within the work and organizational context: Construct introduction and elaboration. *Journal of Organizational Behavior*, 31 (6), 810-834.
- Pierce, J. L. and Jussila, I., 2011. *Psychological ownership and the organizational context: Theory, research evidence, and application*. Northampton, MA, US: Edward Elgar Publishing.
- Pierce, J. L., Jussila, I. and Cummings, A., 2009a. Psychological ownership within the job design context: Revision of the job characteristics model. *Journal of Organizational Behavior*, 30 (4), 477-496.
- Pierce, J. L., Jussila, I. and Cummings, A., 2009b. Psychological ownership within the job design context: revision of the job characteristics model, 477-496.
- Pierce, J. L., Jussila, I. and Li, D., 2018. Development and validation of an instrument for assessing collective psychological ownership in organizational field settings. *Journal of Management & Organization*, 24 (6), 776-792.
- Pierce, J. L., Kostova, T. and Dirks, K. T., 2001. Toward a Theory of Psychological Ownership in Organizations, 298.
- Pierce, J. L., Kostova, T. and Dirks, K. T., 2003. The state of psychological ownership: Integrating and extending a century of research. *Review of General Psychology*, 7 (1), 84-107.
- Pierce, J. L., O'Driscoll, M. P. and Coghlan, A.-M., 2004. Work Environment Structure and Psychological Ownership: The Mediating Effects of Control. *Journal of Social Psychology*, 144 (5), 507-534.
- Porteous, J. D., 1976. Home: The Territorial Core. *Geographical Review*, (4), 383-390.
- Prelinger, E., 1959. Extension and structure of the self. *The Journal of Psychology: Interdisciplinary and Applied*, 47, 13-23.

- Rantanen, N. and Jussila, I., 2011. F-CPO: A collective psychological ownership approach to capturing realized family influence on business. *Journal of Family Business Strategy*, 2, 139-150.
- Reto, F. and Jacob, A., 2019. Nourish what you own: psychological ownership, materialism and pro-environmental behavioral intentions. *Journal of Consumer Marketing*, 36 (1), 82-91.
- Rhew, N. D., Jones, D. R., Sama, L. M., Robinson, S., Friedman, V. J. and Egan, M., 2021. Shedding Light on Restorative Spaces and Faculty Well-Being. *Journal of Management Education*, 45 (1), 43-64.
- Ripamonti, S., Galuppo, L., Bruno, A., Ivaldi, S. and Scaratti, G., 2018. Reconstructing the internship program as a critical reflexive practice: the role of tutorship (6. Vol. 23, pp. 751-768). Great Britain: Taylor & Francis.
- Römgens, I., Scoupe, R. and Beausaert, S., 2020. Unraveling the concept of employability, bringing together research on employability in higher education and the workplace (12. Vol. 45, pp. 2588-2603). Great Britain: Taylor & Francis.
- Rossano, F., Rakoczy, H. and Tomasello, M., 2011. Young children's understanding of violations of property rights. *Cognition*, 121 (2), 219-227.
- Rousseau, D. M., Sitkin, S. B., Burt, R. S. and Camerer, C., 1998. Not so different after all: A cross-discipline view of trust. *Academy of Management Review*, 23 (3), 393-404.
- Rozin, P., Nemeroff, C., Wane, M. and Sherrod, A., 1989. Operation of the sympathetic magical law of contagion in interpersonal attitudes among Americans. *Bulletin of the Psychonomic Society*, 27 (4), 367-370.
- Rubin, H. J. and Rubin, I., 2004. *Qualitative interviewing : the art of hearing data*. 2nd ed. edition. London: SAGE.
- Russell, W. B., 1988. Possessions and the Extended Self. *Journal of Consumer Research*, (2), 139.
- Saila, R., Iiro, J. and Pasi, T., 2012. Leading Together for Mutual Benefit: Shared Leadership in the Context of Co-operative Banking. *Business & Management Research*, 1 (4), 99.
- Saleh, S. D. and Hosek, J., 1976. Job Involvement: Concepts and Measurements. Academy of Management Journal, 19 (2), 213-224.
- Sandberg, J. and Alvesson, M., *Constructing Research Questions: Doing Interesting Research* [online]. Sage Publications.
- Sandelowski, M., 1994. The use of quotes in qualitative research. *Research in nursing & health*, 17 (6), 479-482.
- Sartre, J.-P., 1956. *Being and nothingness*. Oxford: Philosophical Library.
- Savickas, M. L., 2011. New Questions for Vocational Psychology: Premises, Paradigms, and Practices, *19*, 251-258.
- Shaw, R. L., Dyson, P. O. and Peel, E., 2008. Qualitative Psychology at M Level: A Dialogue Between Learner and Teacher, 179.
- Shouse, R. L., 2017. Examining the Influence of Perceptions of a Supervisor's Leadership Style on Levels of Psychological Ownership Among Entry Level Professionals: Bowling Green State University / OhioLINK.
- Siddiqui, S. and Turley, D., 2006. Extending the Self in a Virtual World, 647.
- Silverman, D., 2011. *Qualitative research : issues of theory, method and practice* [Non-fiction]. 3rd ed. edition.: SAGE.
- Sim, J., Saunders, B., Waterfield, J. and Kingstone, T., 2018. Can sample size in qualitative research be determined a priori? (5. pp. 619). Great Britain: Taylor & Francis.
- Sluss, D. M., Ployhart, R. E., Cobb, M. G. and Ashforth, B. E., 2012. Generalizing newcomers' relational and organizational identifications:processes and prototypicality. *Academy* of Management Journal, 55 (4), 949-975.
- Smith, J. A., Harré, R. and Langenhove, L. v., 1995. *Rethinking methods in psychology*. London: Sage.

- Smith, K. M., Hulland, J. and Thompson, S. A., 2018. Cheaters, trolls, and ninja looters: The dark side of psychological ownership. *In:* Scheinbaum, A. C., ed. *The dark side of social media: A consumer psychology perspective.* New York, NY: Routledge/Taylor & Francis Group, 71-88.
- Solana, S. and Alicia, O., 2019. Job involvement in current research: Update and state of the art. *Papeles del Psicólogo*, 40 (1), 46-56.
- Song, L., Professor David Lamond, P., Pan, X.-F., Qin, Q. and Gao, F., 2014. Psychological ownership, organization-based self-esteem and positive organizational behaviors. *Chinese Management Studies*, (1), 127-148.
- Staller, K. M., 2021. Big enough? Sampling in qualitative inquiry. *Qualitative Social Work*, 20 (4), 897-904.
- Steketee, G. and Frost, R., 2007. *Compulsive Hoarding and Acquiring* [Book]. New York: Oxford University Press.
- Su, X., Liang, K. and Wong, V., 2020. The impact of psychosocial resources incorporated with collective psychological ownership on work burnout of social workers in china. *Journal of Social Service Research*.
- Tatone, D., Geraci, A. and Csibra, G., 2015. Giving and taking: Representational building blocks of active resource-transfer events in human infants. *Cognition*, 137, 47-62.
- Tetlock, P. E., 1999. Theory-Driven Reasoning About Plausible Pasts and Probable Futures in World Politics: Are We Prisoners of Our Preconceptions? *American Journal of Political Science*, 43 (2), 335-366.
- Tian, K. and Belk, R. W., 2005. Extended Self and Possessions in the Workplace. *Journal of Consumer Research*, 32 (2), 297-310.
- Tims, M. a. A., J, 2020. Job and career crafting to fulfill individual career pathways. Career pathways: From school to retirement. 165-189.
- Tomlinson, M., 2013. *Education, work and identity.* [electronic resource] : themes and *perspectives.* London: Bloomsbury Academic.
- Tracy, J. L. and Robins, R. W., 2007. The Psychological Structure of Pride: A Tale of Two Facets, 92, 506-525.
- Trice, H. M. and Beyer, J. M., 1984. Studying Organizational Cultures Through Rites and Ceremonials. *Academy of Management Review*, 9 (4), 653-669.
- Tsai, C.-T., Hsu, H. A. n. e. t. and Yang, C.-C., 2017. Career decision self-efficacy plays a crucial role in hospitality undergraduates' internship efficacy and career preparation. *Journal of Hospitality, Leisure, Sport & Tourism Education (Elsevier Science)*, 21, 61-68.
- Turner, V. W., 2011. *The ritual process : structure and anti-structure*. New York: AldineTransaction.
- Van Dyne, L. and Pierce, J. L., 2004. Psychological Ownership and Feelings of Possession: Three Field Studies Predicting Employee Attitudes and Organizational Citizenship Behavior, 439-460.
- Wang, L., Law, K. S., Zhang, M. J., Li, Y. N. and Liang, Y., 2018. It's mine! Psychological ownership of one's job explains positive and negative workplace outcomes of job engagement. *Journal of Applied Psychology*.
- Watkins, R. and Molesworth, M., 2012. *Attachment to Digital Virtual Possessions in Videogames*. Great Britain: Emerald Group.
- Watkins, R. D., Denegri-Knott, J. and Molesworth, M., 2015. The relationship between ownership and possession: observations from the context of digital virtual goods.
- Weil, S., 1952. *The need for roots: prelude to a declaration of duties toward mankind*. New York: Putnam.
- White, R. W., 1959. Motivation reconsidered: the concept of competence. *Psychological review*, 66, 297-333.
- Williams, E. N. and Morrow, S. L., 2009. Achieving trustworthiness in qualitative research: a pan-paradigmatic perspective. *Psychotherapy research : journal of the Society for Psychotherapy Research*, 19 (4-5), 576-582.

Willig, C. and Stainton-Rogers, W., 2017. *The Sage handbook of qualitative research in psychology*. Second edition. edition. Thousand Oaks: SAGE Inc.

Wilson, 2012. Review of business–university collaboration.

- Wilton, N., 2012. The impact of work placements on skills development and career outcomes for business and management graduates, *37*, 603-620.
- Wood, C. M., 2003. The Effects of Creating Psychological Ownership among Students in Group Projects, 241.
- Wrzesniewski, A. and Dutton, J. E., 2001. Crafting a Job: Revisioning Employees As Active Crafters of Their Work. *Academy of management review*, 26 (2), 179-201.
- Xiao-Fu, P., Qiwen, Q. and Fei, G., 2014. Psychological ownership, organization-based selfesteem and positive organizational behaviors. *Chinese Management Studies*, 8 (1), 127-148.
- Ye, Y. and Gawronski, B., 2016a. When possessions become part of the self: Ownership and implicit self-object linking, 72-87.
- Ye, Y. and Gawronski, B., 2016b. When possessions become part of the self: Ownership and implicit self-object linking. *Journal of Experimental Social Psychology*, 64, 72-87.
- Yuan, C., Wang, S., Yu, X., Kim, K. H. and Moon, H., 2021. The influence of flow experience in the augmented reality context on psychological ownership. *International Journal of Advertising*, 40 (6), 922-944.
- Zahid, H., Ikram Ullah, K., Zaryab, S., Tahir, I., Muhammad Imran, R. and Rana Muhammad, N., 2019. Organizational justice and knowledge sharing behavior : The role of psychological ownership and perceived organizational support. *Personnel Review*, 48 (3), 748-773.
- Zhang, J.-Y., Nie, M., Yan, B.-S. and Wang, X.-D., 2014. Effect of network embeddedness on brand-related behavior intentions: Mediating effects of psychological ownership. *Social Behavior and Personality: An International Journal*, 42 (5), 721-730.
- Zhang, L., Zhang, Xu, Cheung, 2020. Psychological Ownership: A Meta-Analysis and Comparison of Multiple Forms of Attachment in the Workplace. *Journal of Management*.

## Appendices

## Appendix One: Ethics Checklist

BU Bournemouth University	Research Ethics Checklist
About Your Checklist	
Ethics ID	27778
Date Created	31/07/2019 14:39:39
Status	Approved
Date Approved	06/08/2019 09:25:42
Date Submitted	31/07/2019 14:51:57
Risk	Low
Researcher Details	
Name	Deborah Taylor
Faculty	BU Business School
Status	Postgraduate Research (MRes, MPhil, PhD, DProf, EngD, EdD)
Course	Postgraduate Research - BUBS
Have you received funding to support this research project?	No
Project Details	
Title	The Development of Feelings of Ownership in the Workplace
Start Date of Project	01/01/2019
End Date of Project	30/06/2020
Proposed Start Date of Data Collection	06/08/2019
Original Supervisor	Lois Farquharson
Approver	Michael Silk

## Appendix Two: Participant Information Form and Participant Agreement Form



#### **Participant Information Sheet**

#### The title of the research project

The Development of Feelings of Ownership in the Workplace

#### Invitation to take participate

You are being invited to take part in a research project. Before you decide it is important for you to understand why the research is being done and what it will involve. Please take time to read the following information carefully and discuss it with others if you wish. Ask us if there is anything that is not clear or if you would like more information.

#### What is the purpose of the project?

The purpose of this project is to ascertain how feelings of ownership for workplace targets such as a job role or the organisation may develop at the start of an individual's career. The study is looking in particular at individuals participating in a work placement during the third year of their degree.

#### Why have I been chosen?

You have been chosen because you are currently undertaking a work placement.

#### Do I have to take part?

It is up to you to decide whether or not to take part. If you do decide to take part, you will be given this information sheet to keep and be asked to sign a participant agreement form. You can withdraw from participation at any time and without giving a reason. If you decide to withdraw we will usually remove any data collected about you from the study. Once data collection activities have finished you can may still be able to withdraw your data up to the point where the data is analysed and incorporated into the research findings or outputs. At this point your data will usually become anonymous, so your identity cannot be determined, and it may not be possible to identify your data within the anonymous dataset. Withdrawing your data at this point may also adversely affect the validity and integrity of the research. Deciding to take part or not will not impact upon/adversely affect your treatment/care /education or studies at BU.

#### What would taking part involve?

Taking part would involve a semi structured interview during which you would be asked questions about your work placement.

#### What are the advantages and possible disadvantages or risks of taking part?

Whilst there are no immediate benefits for those people participating in the project, it is hoped that this work will give researchers a greater understanding regarding the development of ownership feelings in the workplace and the positive and negative implications of such feelings.

# What type of information will be sought from me and why is the collection of this information relevant for achieving the research project's objectives?

The type of questions that you will be asked will be concentrate on your work placement and may include questions about the skills you have developed and your feelings to workplace targets such as your job role or the organisation..

#### Will I be recorded, and how will the recorded media be used?

The audio and/or video recordings of your activities made during this research will be used only for analysis and the transcription of the recording(s) for illustration in conference presentations and lectures. No other use will be made of them without your written permission, and no one outside the project will be allowed access to the original recordings.

#### How will my information be kept?

All the information we collect about you during the course of the research will be kept strictly in accordance with current data protection legislation. Research is a task that we perform in the public interest, as part of our core function as a university. Bournemouth University (BU) is a Data Controller of your information which means that we are responsible for looking after your information and using it appropriately.

#### Publication

You will not be identifiable in any external reports or publications about the research without your specific consent\*. Otherwise your information will only be included in these materials in an anonymous form, i.e. you will not be identifiable.

#### Security and access controls

BU will hold the information we collect about you in hard copy in a secure location and on a BU password protected secure network where held electronically.

#### Retention of your data

All personal data collected for the purposes of this study will be held for one year after the award of the degree]. Although published research outputs are anonymised, we need to retain underlying data collected for the study in a non-anonymised form for a certain period to enable the research to be audited and/or to enable the research findings to be verified.

#### **Contact for further information**

If you have any questions or would like further information, please contact Deborah Taylor [dtaylor@bournemouth.ac.uk]

#### In case of complaints

Any concerns about the study should be directed to Lois Farquharson [Ifarquharson@bournemouth.ac.uk]. If you decide to take part, you will be given a copy of the information sheet and a signed participant agreement form to keep.

#### Thank you for considering taking part in this research project.



### **Participant Agreement Form**

Full title of project: The Development of Feelings of Ownership in the Workplace

Name, position and contact details of researcher: Deborah Taylor, Lecturer, dtaylor@bournemouth.ac.uk

Name, position and contact details of supervisor: Lois Farquharson, Deputy Dean, Ifarquharson@bournemouth.ac.uk

To be completed prior to data collection activity

Section A: Agreement to participate in the study

You should only agree to participate in the study if you agree with all of the statements in this table and accept that participating will involve the listed activities.

I have read and understood the Participant Information Sheet and have been given access to the BU Research Participant <u>Privacy Notice</u> which sets out how we collect and use personal information (<u>https://www1.bournemouth.ac.uk/about/governance/access-information/data-protection-privacy</u>).

I have had an opportunity to ask questions.

I understand that my participation is voluntary. I can stop participating in research activities at any time without giving a reason and I am free to decline to answer any particular question(s).

I understand that taking part in the research will include the following activity/activities as part of the research:

being audio recorded during the project

my words will be quoted in publications, reports, web pages and other research without using my real name.

I understand that, if I withdraw from the study, I will also be able to withdraw my data from further use in the study **except** where my data has been anonymised (as I cannot be identified) or it will be harmful to the project to have my data removed.

I understand that my data may be used in an anonymised form by the research team to support other research projects in the future, including future publications, reports or presentations.

	Initial box to
	agree
I consent to take part in the project on the basis set out above (Section A)	

#### Section B: The following parts of the study are optional

You can decide about each of these activities separately. Even if you do not agree to any of these activities you can still take part in the study. If you do not wish to give permission for an activity, do not initial the box next to it.

	Initial boxes to
	agree
I understand that my words may be quoted in publications, reports, web pages and other research outputs. Please choose one of the following two options: I agree that my real name can used in the above. I do <b>not</b> agree that my real name can be used in the above.	

I confirm my agreement to take part in the project on the basis set out above.

Name of participant

(BLOCK CAPITALS)

(dd/mm/yyyy)

Date

Signature

Signature

Name of researcher

(BLOCK CAPITALS)

(dd/mm/yyyy)

Date

#### **Appendix Three: Pilot Interview Questions**

#### **Pilot Interview**

- Tell me about your placement search and the process that took you to your placement at \_\_\_\_\_ company.
- Can you tell me about your first week did you have an induction/handover? Possible follow up questions:
- 3. What did they think of the job role/organisation?
- 4. Who were the key people that you initially met?
- 5. Tell me a little bit more about your job role and the organisation?
  - a. Are there any elements that you have particularly enjoyed or struggled with?
  - b. How has the job role changed over time?
  - c. Have you adapted the job role in any way?
  - d. How much autonomy do you believe you have over your work?
  - e. How do you feel about the job role/organisation
- 6. What skills/competencies do you think you have developed/improved on?
  - a. Have you gained confidence/self efficacy
  - b. Do you feel in control of their work load?
- 7. As you know my study is about psychological ownership. What does ownership in the workplace mean to you?
  - a. Do you see examples of colleagues who seem to demonstrate ownership?
    - i. Do you have examples of times that you have demonstrated ownership?
- 8. Have you experienced feelings of ownership previously perhaps in a part time job at uni or via your hobbies?
  - a. Can you provide any examples of ownership

- 9. Have you considered work options post-graduation?
  - a. Do you see yourself in a similar job role?
  - b. Would you work for this organisation again?

#### **Appendix Four: WPS Interview Questions**

#### Interview Schedule – Work Placement Students

1. Tell me about your placement search and the process that took you to your placement at company.

Aim: To ascertain how participants felt about the role/organisation before they started work e.g. was this one of their top choices re role/org or did they compromise.

- 2. Can you tell me about your first week on your work placement? Possible follow up questions:
  - a) Was there a structured induction process in place? How thorough was the handover?
  - b) What were their initial thoughts regarding job role/organisation?
  - c) Who were the key people that they initially met and what were their views of the organisation?
  - d) Did you hear any organisational stories which reflected the culture of the organisation?

Aim: To understand more about the transition into work and the crucial initial entry period. How the organisation helps new starters settle into the organisation.

- 3. Tell me a little bit more about your job role and the organisation now? Can you provide examples of
  - a) Any elements of the job role/organisation that you have particularly enjoyed or struggled with?
  - b) How has the job role changed over time?
  - c) Have you adapted the job role/Improved processes etc?
  - d) How much control do you have regarding the tasks you complete or your work load in general?
  - e) How would you describe the organisation?
  - f) What are your feeling towards the job role/organisation?

g) What are your colleague's feelings towards their work and the organisation? Aim: As per routes to PO – examples of investing self, intimate knowledge of job role/org, control over job role and motives of PO – examples of self efficacy, sense of belonging and self-identity. Examples of other individuals PO

- 4. Can you provide any examples of skills/competencies that you have developed/improved on during your time on placement?
  - a) To what extent has your self-confidence improved in this job role? If yes in what way/if no, why might this be so?

Aim: To ascertain levels of self efficacy

5. What opportunities have you had to work with senior managers or key decision makers?a) What are the attitudes of senior management to placement students?

b) Do you know how these individuals feel about the organisation?

Aim: To assess opportunities to invest self in company/sense of belonging via exposure to senior people and also to role modelling ("possible selves").

- 6. What contribution do you believe you have made to the organisation on your work placement?
  - a. Do you feel valued if so, by whom?
  - b. Does the environment around you encourage excellence?
  - c. Does the organisation have any mechanisms to show they value contributions? e.g. employee of the month

Aim: Assess organisational opportunities and its impact on PO development

- If you could change one thing about your placement what would that be? Aim: Link to PO motives and routes, impact of org decisions on those at the start of their career
- 8. As you know my study is about psychological ownership. What does ownership in the workplace mean to you?
  - a) Do you have any examples of colleagues who seem to demonstrate ownership in the workplace?
  - b) Do you have examples of times that you have demonstrated ownership in the workplace?

Aim: To assess other possible motives/routes to PO

9. At what point in your placement do you believe you developed feelings of ownership? (relevant to those students who believe they show feelings of ownership)

a. Was this the result of seeing others demonstrate ownership? Aim: To assess time period when developing PO (links back to questions 3,4,7)

- 10. Are there any negative implications associated with feeling ownership in the workplace (either for you, colleagues or the organisation)?
- 11. Can you provide any examples of individuals who are very territorial of their work, their ideas or their space around them?

Aim: Posited that there are negative implications of PO such as workplace stress, territoriality

12. Have you experienced feelings of ownership previously - perhaps at uni, in your hobbies or a part time job?

Aim: to assess if there are building blocks outside of the workplace for the development of PO

- 13. What sort of roles are you hoping to apply for after graduating?
  - a) What type of organisation do you see yourself working for?
  - b) Has this organisation provided you with information regarding their graduate roles?

Aim: Consider future plans to compare to their feelings of ownership (are they transferable, what are the conditions to become further linked with the organisation).

#### Appendix Five: Employer Interview Questions

**Employer Interview Questions** 

- 1. Could you start by telling me a little bit about the recruitment process that you use to hire placement students?
  - a. What skills and competencies do you look for in a student?
  - b. What are the key factors that you look for?
  - c. Do you ask students to undertake any tests?
  - d. Do you complete your own recruitment, is there are separate team, do you outsource some of the process?

Aim: Scene setting, but also to understand the opportunities for students to invest in the company during the recruitment process and start to gain a sense of belonging and see examples of "possible selves"

2. Could you tell me a little bit about the sort of work students undertake for you whilst on placement?

a. For how many years have you been employing placement students Aim: Ice breaker question to start the conversation, understand a bit more about the students work and provide level of experience in managing students. May also help us understand if these jobs fit the notion of complex roles.

- Going from education into the workplace can be a big transition for individuals and I wondered what sort of processes are in place to help students make that transition during their first few weeks.
  - a. What is the organisation role?
  - b. What is the departmental role?
  - c. What is the line manager role?
  - d. Role of the previous incumbent if a student?

Aim: To understand more about the transition into work and the crucial initial entry period. How the organisation helps new starters settle into the organisation and creating that sense of belonging. Examples of changing identity & developing "possible selves"

- 4. Do you have any examples of student's proactive behaviour prior to starting their work placement or in the first few weeks?
  - e) Do some students try and meet their team before they start their placement or email the team regarding how to prepare?
  - f) Do students immediately offer to be involved in any extra activities such as helping with student recruitment?
  - g) Do you think that if students actively try to facilitate a strong transition that this has positive outcomes further into the placement?

Aim: To understand if it is possible to show early signs of ownership via any of the established PO routes occurs examples of investing self, intimate knowledge of job role/org, control over job role and motives of PO – examples of self efficacy, sense of belonging and self-identity.

- 5. Going back to the job role, on average how long does it take for students to feel comfortable in their role and what factors do you think contribute to these feelings?
  - Which part of the job role do students become familiar with most quickly?
  - How do you support student development?

Aim: To understand if it is possible to show early signs of ownership, but also how that sense of ownership may develop. Is it via some of the established PO routes such as of investing self, intimate knowledge of job role/org, control over job role and motives of PO – examples of self efficacy, sense of belonging and self-identity.

- 6. What sort of contribution have students made to the dept /organisation whilst on placement?
  - a. Have the students adapted/improved processes

b. Have the students introduced new ideas to the team such as new technology Aim: to demonstrate if students are able to add value to a business

- 7. What do the students who perform well on their placement do differently from those students who perform less well?
  - a. At what point are the students considered to be a high performer?
  - b. How easy is it to maintain these accolades throughout a placement?
  - c. Do the students who perform well have lots of control over their workloads?
  - d. Is performance related to confidence in your experience?
  - e. Is it a matter of "fit" to the job role or organisation?
  - f. In your opinion what impact can you as a line manager have on the student's performance?

Aim: To ascertain if there are links between any of the roots to PO are mentioned e.g. investing self, intimate knowledge of job role/org, control over job role and motives of PO – examples of self efficacy, sense of belonging and self-identity as well as other elements that may impact on the students

- 8. What does the organisation or the department do to make staff feel valued are students included in this?
  - a. What are the attitudes of senior managers towards placement students?
  - b. Does the student take part in the normal company appraisal or do they have a separate review?
  - c. How regularly do you have one to ones with students?

Aim: Assess organisational opportunities and its impact on PO development

- As you know my study is about psychological ownership. How would you describe ownership in the workplace please?
   Aim: To ascertain an individual's personal view of what they believe ownership to be and compare to the current literature
- 10. Do you have any examples of students demonstrating ownership in the workplace?
  - c) Do they feel ownership for their job role?
  - d) Do they feel ownership for their team?
  - e) Do they feel ownership for the organisation?
  - f) Any other types of ownership such as ownership of their career?
  - g) What are the main characteristics that these students have?

Aim: To assess the type of ownership they may have observed in students including career ownership which has been mentioned by students

- 11. At what point might these feelings of ownership arise?
  - a. Do feelings of ownership for their job role, org and team appear at the same time or at different times?
  - b. Do you think it is easier to feel ownership for some targets more than others?

Aim: To ascertain time periods of PO development

- 12. Do you think the organisation encourages ownership and if so, do you have any examples of this? How might this message be communicated to staff? Aim: Assessing if they feel that ownership is something that can be learnt and developed
- 13. What are the benefits to the organisation when students demonstrate ownership in their role?
  - a. What are the benefits to the team?
  - b. What are the benefits to you as a line manager?
  - c. What are the benefits to the student?

Aim: Trying to assess who benefits most from ownership or is it mutually beneficial. To see if any of the current considered positives are mentioned such as organisational citizenship behaviour, organisational commitment, job satisfaction

- 14. Are there any negative to students demonstrating workplace ownership?
  - a. To the student
  - b. To their line manager
  - c. To the team
  - d. To the organisation

Aim: Trying to assess what negative impact PO may have and how it affects others and/or the organisation. Confirm if any of the posited negatives such as stress, territoriality are mentioned.

15. Is ownership in students constant or can it be more fluid and will come and go

Aim: Referring back to our relationship with possessions which some have suggested can wax and wane.

#### **Appendix Six: Placement Development Advisor Interview Questions**

#### **PDA Interview Questions**

- 16. Would you mind giving me a brief overview of the role of a Placement Development Advisor please?Aim: Ice breaker question to start the conversation and provide clarity regarding the PDA role
- 17. How long have you been working in this role please? Aim: Ascertain years of experience
- 18. Going from education into the workplace can be a big transition for individuals and I wondered what sort of processes do organisations have in place to help students make that transition such as an induction process?
  - a. Do you think there are any benefits to a strong induction process?
  - b. Do you have any examples of good practice?
  - c. Do you have any examples of poor practice?
  - d. Is there anything more that organisations could do to facilitate a successful transition?

Aim: To understand more about the transition into work and the crucial initial entry period. How the organisation helps new starters settle into the organisation and creating that sense of belonging. Examples of changing identity & developing "possible selves"

- 19. Have you seen any examples whereby students have been proactive in preparing for their placement and the transition into the workplace if so, do you have any examples please?
  - h) Do some students try and meet their team before they start their placement or email the team regarding how to prepare?
  - i) Do students immediately offer to be involved in any extra activities such as helping with student recruitment?
  - j) Do you think that if students actively try to facilitate a strong transition that this has positive outcomes further into the placement?

Aim: To understand if it is possible to show early signs of ownership via any of the established PO routes occurs examples of investing self, intimate knowledge of job role/org, control over job role and motives of PO – examples of self efficacy, sense of belonging and self-identity.

- 20. What do the students who perform well on placement do differently from those students who perform less well?
  - a. Do you find lots of high performers in one organisation why might this be so?
  - b. At what point are the students considered to be a high performer?
  - c. How easy is it to maintain these accolades throughout a placement?
  - d. Do the students who perform well have lots of control over their workloads?
  - e. Is it a matter of "fit" to the job role or organisation?
  - f. What impact does confidence regarding their job role have on their performance in your opinion?
  - g. In your opinion what impact does their line manager have on the student's performance?

Aim: To ascertain if there are links between any of the roots to PO are mentioned e.g. investing self, intimate knowledge of job role/org, control over job role and motives of PO – examples of self efficacy, sense of belonging and self-identity as well as other elements that may impact on the students

21. As you know my study is about psychological ownership. How would you describe ownership in the workplace please?
Aim: To assortain an individual's personal view of what they believe ownership to be a second view of what they believe ownership to believe ownership to second view of what they believ

Aim: To ascertain an individual's personal view of what they believe ownership to be

- 22. Do you have any examples of students demonstrating ownership in the workplace?
  - h) Do they feel ownership for their job role?
  - i) Do they feel ownership for their team?
  - j) Do they feel ownership for the organisation?
  - k) Any other types of ownership?
  - I) What are the main characteristics that these students have?

Aim: To assess the type of ownership they may have observed in students

- 23. At what point in their placement do these feelings of ownership arise?
  - a. Do feelings of ownership for their job role, org and team appear at the same time or at different times?
  - b. Do you think it is easier to feel ownership for some targets more than others?

Aim: To ascertain time periods of PO development

- 24. What characteristics/factors do you believe help the development of ownership feelings in these students?
  - a. From student perspective
  - b. From managers perspective
  - c. From Organisational perspective

Aim: Again to see if any of the posited routes and motives are mentioned – examples of investing self, intimate knowledge of job role/org, control over job role and motives of PO – examples of self efficacy, sense of belonging and self-identity.

25. In your opinion can you think of any positive and negative implications for students who develop feelings of ownership in the workplace for students?
a. What about positive or negative implications for the organisation?
Aim: To ascertain from their experience the positives/negatives attached to feelings of ownership e.g. purpose/workplace stress or territoriality

### Appendix Seven: Data Analysis Phase One (Data familiarisation)

## Extract One

Vite	What did you think of the company and the role initially? You	
	got in, your first couple of weeks, what were your thoughts?	
1 -	org culture	
souchset	The company, I think my opinion hasn't changed. It's extremely	
challer Breest	fast moving. I felt like at the beginning I was okay, I don't get	
1000	things yet, but I'm going to start getting them because you start	
complates	settling in and stuff, it doesn't change because everything moves	0
kst to	so, so fast within the company that it was always fast paced. And	a via
culture	we were always learning, and you can never get comfortable	Ele Settos
all learning	thinking that you know exactly what to do or where you are,	
not Jost H	because you don't. I think, as a company in terms of the culture, I	
feets prof	loved it because at no point I felt like I was any under or below the	
aviety more	level of obviously in terms of the manager, but because I was a	1011-000
(dettiby	placement student, I never felt like I was less part of the team, or	contractions feeling
out	not taken, or given as much responsibility. I was given the equal	equal
Social identit	amount of responsibility as other members in my team. And then	O

## Extract Two

× *	linusdity? - togethy coor
n A	Scared. It made me feel scared but I think also great. I felt very
dept of	I think I felt like I had, to a certain extent, and because I don't think Grote labore
moth het	my manager would have trusted me if she didn't think that I could
2 crops	have handled that, but it was so good to know that she was there.
not see	Even though I didn't want to bother her all the time, but I didn't feel Sopport 55
mer time	like I was just left to do this big job and didn't have no help. And I
	had always had like check-ins with her. So the whole process, she
emphasising	was there with me, but it wasn't handholding. She was very, very
Toroship	liberal, she allowed me to just do everything. I always kept her on
0	Cc's, so she was always informed of what's going on, but she very
	rarely came in and told me, you should do it like this, or you should
mention of	do it like that, she trusted me to do. And I think that was definitely
Gross -0	a step for me to then be able to do bigger things later because I felt
### Extract Three

abbonon

self

Yeah, that makes sense, thank you. Do you know at what point in your placement you think you started having feelings of ownership?

### arodif

Fairly early, but I think that was due to my manager actually. Full on giving me that ownership. Personally, I like to think that even if I wouldn't have a manager that full on gave me things on a silver platter, I would have eventually asked for some ownership because I am ... I like to be able to do things myself. So my regular day-today job; carry it however way I needed to, however there's always areas like, maybe we can look into this and then if the answer is yes, then can I then look into doing it myself, and exploring other areas. But even I feel like at the end of it I wasn't given the

# Appendix Eight: Data Analysis Phase Two (Generating Codes including Phase Two Codes)

DJs, and then the winner would actually go to Amsterdam to the biggest house event. That music.

#### Nice prize.

Yeah, so I helped ... it was during the summertime, so the member of the team that covers this area had left to go on holiday and my manager literally just said, you seem to know what you're doing, just run this. Obviously, she was always ... she was sat next to me, so she was always there to help me, but she gave me full control over that campaign and that took most of my first three months. And I ran it and it was great. One of the best things I did.

#### How did it make you feel, being given that control?

Scared. It made me feel scared but I think also great. I felt very ... I think I felt like I had, to a certain extent, and because I don't think my manager would have trusted me if she didn't think that I could have handled that, but it was so good to know that she was there. Even though I didn't want to bother her all the time, but I didn't feel like I was just left to do this big job and didn't have no help. And I had always had like check-ins with her. So the whole process, she was there with me, but it wasn't handholding. She was very, very liberal, she allowed me to just do everything. I always kept her on Cc's, so she was always informed of what's going on, but she very rarely came in and told me, you should do it like this, or you should do it like that, she trusted me to do. And I think that was definitely a



### Phase Two Codes

Factors which influenced Placement Search Choice	Feeling of being part of a group
Ideal Placements - Sectors WPSs were really interested in	Terminology - we
Organisation of Placement Search	Identifying with Org
Job Roles of Interest	Contract Staff
Recruitment Process - Interview Stage	Contamination
Time Period found Placement	Rituals
Induction at Organisation	Disinvestment-Decoupling
Organisation methods of socialisation	Contribution
Learning the Ropes-Sensemaking	High Quality Work
Thrown in the deep end	Feeling Valued
Feeling Scared-Nervous	Feeling of Happiness
General Overview of Job Role	Sense of Purpose
Job Role Variety	Control over workload - career
Volunteering	Ownership of self - opportunities - career
Mismatch between WPS expectations and role- org	Creating Something
Teamwork	Creativity on Placement
Work Team Experiences	Investing themselves into job role
Frustrations with individuals-teams	Opportunities to put own stamp on work
Communicating with others in organisation	Ethnicity
Challenging Environment	Feeling territorial about someone-something
Frustration at Org	Feeling of disappointing people
Working Environment	Feeling Worried

Placement influence on career aspirationsJob Role CharCommitment to the OrganisationJob Roles CharLiquidityOrganisationAppraisal's - formal or informationalJob-Org-IndiaWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	whelmed
Competition between participantsInternal PressLinks to Previous WPSs or CohortsFeeling TrussFuture GoalsTrust in otherThinking ahead to future workGiving OwnerWork ShadowingImpact of TrustPlacement influence on career aspirationsJob Role CharCommitment to the OrganisationJob Roles CharLiquidityOrganisationWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	
Links to Previous WPSs or CohortsFeeling TrustFuture GoalsTrust in otherThinking ahead to future workGiving OwnerWork ShadowingImpact of TrustPlacement influence on career aspirationsJob Role CharCommitment to the OrganisationJob Roles CharLiquidityOrganisationAppraisal's - formal or informationalJob-Org-IndiaWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	sure
Future GoalsTrust in otherThinking ahead to future workGiving OwnerWork ShadowingImpact of TrustPlacement influence on career aspirationsJob Role CharCommitment to the OrganisationJob Roles CharLiquidityOrganisationAppraisal's - formal or informationalJob-Org-IndiaWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	
Thinking ahead to future workGiving OwnerWork ShadowingImpact of TrPlacement influence on career aspirationsJob Role CharCommitment to the OrganisationJob Roles CharLiquidityOrganisationAppraisal's - formal or informationalJob-Org-IndiWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	ed
Work ShadowingImpact of TransPlacement influence on career aspirationsJob Role CharCommitment to the OrganisationJob Roles CharLiquidityOrganisationAppraisal's - formal or informationalJob-Org-IndiaWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	r people
Placement influence on career aspirationsJob Role CharCommitment to the OrganisationJob Roles CharLiquidityOrganisationAppraisal's - formal or informationalJob-Org-IndiaWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	rship
Commitment to the OrganisationJob Roles ChLiquidityOrganisationAppraisal's - formal or informationalJob-Org-IndiWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	avel on Placement
Liquidity       Organisation         Appraisal's - formal or informational       Job-Org-India         Work Metrics - targets to achieve       Justifying Out         Understanding organisation's expectations       Line Manage	nges due to Organisational Changes
Appraisal's - formal or informationalJob-Org-IndiWork Metrics - targets to achieveJustifying OutUnderstanding organisation's expectationsLine Manage	anges
Work Metrics - targets to achieveJustifying OuUnderstanding organisation's expectationsLine Manage	al Change
Understanding organisation's expectations Line Manage	vidual Fit
	itcomes
Access to sonior managers Wanting over	r
Access to senior managers Wanting-exp	ecting Praise
Influences on WPSs Loyalty to so	meone or something
Being Inspired Persuasive b	ehaviour
Mentors or similar staff members Poor Practice	2
Additional Learning Needs Power in an	ownership context
Additional Responsibility Previous WP	S Experience
WPSs wanting or taking Responsibility Pushing back	on People
Owning Mistakes Putting nam	e to something
Characteristics of Ownership of Job Role Recognition	by others
Ambassador for Organisation Pride in Wor	k
Ambition Self Doubt	
Independence Self Percepti	

Motivation	Shared Ownership
Area's WPSs would change	Skill Development whilst on Placement
Accountability	Stakeholder Management
Asking for help	Strategic behaviour
Management Control	Relationship Building
Feeling invested in	Networking Opportunities
Feeling listened too	WPS Beliefs
Work Life Balance	WPS Identity
Completing whole tasks	Hardworking
Control over job role	WPS Loneliness
Examples of Autonomy - not just WPSs	Taking on too much work
Examples of being Proactive	Struggling with workload
Freedom relating to job role	Time Management
Wanting too much control	Time period WPS felt ownership
Organisational Control	
WPSs taking Leadership	
WPS as Manger	
Confidence	
WPSs liking-disliking a challenge	
Out of Comfort Zone	
Connection to Org	

# Appendix Nine: Data Analysis Phase Two Continued (Data Coding Reduction)

Name	Reason for merging codes/Relevance	New Name	
Factors which influenced Placement Search Choice			
Ideal Placements - Sectors WPSs were really interested in	Similarity to other codes regarding to placement search		
Organisation of Placement Search		Placement Search	
Job Roles of Interest			
Recruitment Process - Interview Stage			
Time Period found Placement			
Induction at Organisation			
Organisation methods of socialisation		Socialisation Period	
Learning the Ropes-Sensemaking	Overlap of categories/all relate to the initial socialisation period		
Thrown in the deep end			
Feeling Scared-Nervous			
General Overview of Job Role			
Job Role Variety	Overlap of categories	Job Role Content	
Volunteering			
Mismatch between WPS expectations and role-org		??	
Teamwork			
Work Team Experiences	Overlap of categories	Team Dynamics	
Frustrations with individuals-teams			
Communicating with others in organisation	Overlap of categories	Organisational Culture	

Name	Reason for merging codes/Relevance	New Name	
Challenging Environment			
Frustration at Org			
Working Environment			
Organisation Culture			
Remote Working			
Participant's Peer Group			
Competition between participants	Overlap of categories	Participant's Peer Group	
Links to Previous WPSs or Cohorts			
Future Goals			
Thinking ahead to future work			
Work Shadowing	Catagorias all relate to future	Future Plans/Goals	
Placement influence on career aspirations	Categories all relate to future plans/goals		
Commitment to the Organisation			
Liquidity			
Appraisal's - formal or informational			
Work Metrics - targets to achieve	Overlap with both areas relating to organisational measurement of staff	Organisational Measurement of Staff	
Understanding organisation's expectations	organisational measurement of stan	Statt	
Access to senior managers			
Influences on WPSs		Role Models/Mentors/Influences	
Being Inspired	Overlap of categories		
Mentors or similar staff members			
Additional Learning Needs	Not Relevant to this study		
Additional Responsibility			
WPSs wanting or taking Responsibility	Overlap of categories	Responsibility	
248			

Name	Reason for merging codes/Relevance	New Name
Owning Mistakes		
Characteristics of Ownership of Job Role		
Ambassador for Organisation		Social Identity
Ambition		
Independence		WPS Traits
Motivation		
Area's WPSs would change	Overlapping with other categories – move to more relevant categories & withdraw code	
Accountability		Accountability
Asking for help		
Management Control		
Feeling invested in	Links to one construct	Psychological Safety
Feeling listened too		
Work Life Balance		
Completing whole tasks		
	Overlap of categories	Taking Control

Appendix Ten: Data Analysis Phase Three (Generating Initial Themes)



## Phase 3: Generating Initial Themes

## Phase 3: Generating Initial Themes (Codes)

Socialisation	Organisational Influenc e	What is Ownership	Ownership Outcomes	Future Careers
<ul> <li>Socialisation Period</li> <li>Participant Peer Group</li> <li>Social Identity</li> <li>Psychological Safety</li> <li>Sense of Belonging</li> <li>Rituals</li> <li>WPS Expectations</li> <li>Previous Work experience</li> <li>Time period WPS felt Ownership</li> </ul>	<ul> <li>Team Dynamics</li> <li>Organisational Culture</li> <li>Role Models/Influences</li> <li>Accountability</li> <li>Psychological Safety</li> <li>Sense of Belonging</li> <li>Rituals</li> <li>Trust</li> <li>Change</li> <li>Line Manager</li> <li>Power</li> <li>Shared Ownership</li> <li>Relationship Building</li> <li>Work Load</li> </ul>	<ul> <li>Job Role Content Responsibility WPS Traits</li> <li>Taking Control</li> <li>Self Efficacy</li> <li>Psychological Safety</li> <li>Sense of Belonging</li> <li>Investing self-Self Identity-Mastery</li> <li>Trust</li> <li>Self Congruity</li> <li>Managing Upwards</li> <li>Self Perceptions</li> <li>Skill Development</li> <li>Relationship Building</li> <li>Self Identity</li> </ul>	<ul> <li>WPS Leadership Role</li> <li>Positive Feelings</li> <li>Negative Feelings</li> <li>Territorality</li> <li>Pride &amp; Recognition</li> <li>Skill Development</li> </ul>	<ul> <li>Disinvestment- decoupling</li> <li>Future Plans/ Goals</li> <li>Career PO</li> <li>Liquidity</li> </ul>

# Appendix Eleven: Data Analysis Phase Four and Five (Reviewing and Defining Themes)

#### PO Motives Taking Giving Ownership Future Careers Strength of PO Disinvestment-decoupling •Future Plans/ Goals •Career PO Organisational Culture Liquidity PO Outcomes Manager Positive feelings Negative feelings •Pride & Recognition

## Phase 4: Developing and reviewing themes

## Phase 5:Refining, defining and naming themes





Appendix Twelve: A Mind Map of the Messy Reality

### Appendix Thirteen: Extracts from Reflective Journal

### Student One 3 May 2019

Interview today at 14.00. Frustratingly I have a bit of a headache. Just hoping it won't turn into a migraine. Looking forward to finally starting interviews. This is a student that I know a little. He came in and did a talk to other students whilst on placement, so we had some correspondence and a chat. I don't really know him from being part of the cohort however. Office is nice and quiet so I feel that I have been able to prep well. I've gone through my notes on interviews which has some reminders of what to do before, during and after.

The first interview went well. I think I set the scene and he felt relaxed. He talked a lot which was good, so I didn't always have to ask too many questions. I didn't have a watch with me and so wasn't sure of timing and my phone went dark. Think about a way of managing this. I think I probably could have probed a bit more on a couple of occasions, but generally the flow was good. I closed well, but annoyingly when I turned off the recorder, he started to talk about his volunteering role and gave loads of examples about identity & place of home – how annoying!!

### PDA Interviews 1, 2 and 3 24 June.

These are different types of participants and I am using different questions and so have that initial anxiety about starting again. I am completing all three interviews on the one day almost back to back and am quite excited about what they will have to say. I used to work with one of the participants, but do not know the others particularly well, so I am sure there will be a different dynamic. As they work with students on placement, they are on the outside looking in to organisations. They do however see all of our students whilst they are on placement almost in a mentor type role. This means they will have seen a good spread of students across the years.

They were all such different personalities, but even though they expressed themselves in different ways, they said such similar things which is really interesting.

### **Employer Two**

I was delighted that this person agreed to be interviewed as she is based at a global organisation and has been line managing placement students for many years. This organisation also state that ownership is one of their core values, so really excited when they said yes! She is Alumni of the university but graduated before I joined and so I have only met her when she has come in for the occasional guest lecture and I don't think we had been in contact for around 3 years. I wasn't sure how open she would be with me. I wasn't nervous at all – more intrigued. It was a fascinating interview as this is an organisation who has ownership embedded. Such a different set up in the employer compared to the first employer interview. Good to have that contrast, however.

End of Appendices